The roles of the chemokines CXCL12 and CXCL16 in breast cancer

Sharon Hampton-Smith, B.Sc. (Hons.)

Discipline of Microbiology & Immunology School of Molecular & Biomedical Science University of Adelaide

A thesis submitted to the University of Adelaide in fulfilment of the requirements for the degree of Doctor of Philosophy

September 2007



Declaration

This work contains no material that has been accepted for the award of any other degree or diploma in any university or other tertiary institution and, to the best of my knowledge, contains no material previously published or written by another person, except where due reference is made in the text.

I give consent to this copy of my thesis, once deposited in the University Library, being made available for loan and photocopying.

Sharon Hampton-Smith, B.Sc. (Hons.) September 2007

Acknowledgments

I could write a volume equivalent to this thesis expressing my gratitude to everyone who has helped me to navigate my way to the end of my PhD studies, but for the sake of a forest somewhere, I'll keep it to a couple of pages.

First, a big thank you to my supervisor, Shaun McColl. You conceived my project to begin with and provided me with the guidance I needed to get it up and running. But you were also brave enough to step back and let me try new things and develop new skills. You let me find the independence and confidence to conduct scientific research that can only come from getting in there and doing it. And of course, of late I have deeply appreciated your proofreading of innumerable drafts in amongst all the other responsibilities you have to take care of! Thank you also to my co-supervisor, Robin Anderson. You have provided invaluable cancer and metastasis expertise, as well as more specific advice regarding the 4T1.2 model that was central to this project. I appreciate all the practical tips and the conceptual assistance you have provided along the way and the time you took to meet with me when you made the trip from Melbourne to Adelaide.

Another big thank you to my colleagues; you are such a wonderful bunch of people! Your friendship and understanding have been invaluable as I have gone through the PhD process, as has your practical assistance in so many things. Adriana, Scott, and Brock, the stalwarts of the lab, thank you for always being so cheerful and fun, as well as for keeping the lab running. Marina and Manu, many thanks for all your good advice and laughs as well, and special thanks to Iain for your help with the ICCS, for bouncing ideas around with me and for much needed encouragement. Sarah, Julie, Meizhi and Wendel, my fellow PhD students, thanks for all the fun, laughter and sympathy – all crucial ingredients for surviving in science! Koi, thank you for constant encouragement, helpful tips and understanding as I have been writing, and Yuka, the ball is now in your very capable hands – good luck as you embark on your PhD studies, I know you'll do well. Matt, Mark and Leila, each of you has added your own special quality to the lab – it has been great getting to know you, and I wish you all the best for honours and beyond. Former members of the lab also have been great friends as well: Oli, thanks for your help with cytotox assays and for making me laugh, even when I thought I couldn't; Jane, beautiful lady, thank you for

bringing life to the lab; and KK, thank you so much for all the girly days, chick flicks and culinary adventures – your friendship has been such a blessing!

I wouldn't be where I am today if I did not have the love and support of my family to keep me going outside of the lab. Thanks, Mum, for making sure that I was properly fuelled and rested, and for girly chats in those few times when I had a break from uni. Dad, thank you for the rides home when I got held up at uni later than expected, and for sacrificing time on the computer while I have been writing up! And Rachel, thank you for understanding and sympathy, as well as good scientific advice. It has been great to be able to talk to someone who understands the whole science thing.

A collective thank you to all my friends 'on the outside': you must be sick of hearing about my study and my thesis, but you've kept encouraging me and commiserating with me throughout. Thank you to my extended family at TTGUC, and particularly to the Tuesday Night Girls – our girly get-togethers often turned a bad day into a good one. Thanks also to the Gang – there's nothing like chocolate fondue, a crazy board game and good friends to help a stressed PhD student to unwind!

My darling D! You have been there for me first as friend, and now as lover too. No human being has ever heard such griping and yet remained so steadfast in love and patience. On a purely practical level, I am so grateful for your help in setting up the formatting of the BBT – it has made the writing of it a whole lot easier. But way beyond that, you have provided the perspective I had such trouble maintaining and helped me to see that my glass was half full. Thank you for being there at all times, for silent sympathy, for gentle encouragement, for silliness and for big hugs! T...M...D...always.

Of course, to my Beautiful One: You alone have seen every high and low and cared more deeply than I can imagine about each. Yet long after the pages of this thesis have crumbled to dust, You will go on. Thank You that such love will never fade.

Table of Contents

Declarati	on	iii
Acknowl	edgments	iv
Table of	Contents	vi
Table of 1	Figures	X
List of ta	bles	xii
Abbrevia	itions	xiii
Publicati	ons arising from this work	xvi
Abstract		xvii
CHAPTE	R 1: Introduction	3
1.1 O [•]	verview	3
1.2 Bi	reast cancer	4
1.2.1	Breast cancer progression	4
1.2.2	The immune response to breast cancer	8
1.2.2	2.1 Immunosurveillance	8
1.2.2	2.2 Immune evasion	9
1.2.2	2.3 Active Immunosuppression	10
1.2.2	2.4 Overcoming tumour-induced immunosuppression	12
1.3 TI	he chemokine family	13
1.3.1	General properties of chemokines	13
1.3.2	Chemokine receptor signalling	14
1.3.3	Biological functions of chemokines	15
1.4 C	XCL12/Stromal cell-derived factor (SDF)-1, CXCR4 and CXCR7	17
1.4.1	General properties of CXCL12 and its receptors	17
1.4.2	Physiological functions of CXCL12	19
1.4.2	E.1 Functions of CXCL12 in the haematopoietic system	19
1.4.2	P.2 Functions of CXCL12 in non-haematopoietic tissues	21
1.4.3	CXCL12 and CXCR4 in cancer	22
1.5 C	XCL16 and CXCR6	25
1.5.1	General properties of CXCL16 and CXCR6	25
1.5.2	Physiological functions of CXCL16	26
1.5.3	CXCL16 and CXCR6 in cancer	

1.6	Th	e Research Project	
<u>CHAP</u>	TE	R 2: Materials & Methods	45
2.1	Su	bcloning of chemokine constructs	
2.1	.1	Chemokine DNA constructs	
2.1	.2	Cloning vector	
2.1	.3	Polymerase Chain Reaction	
2.1	.4	Subcloning	
2.1	.5	Identification of positive clones	
2.2	Ge	neration of transfected cell lines	
2.2	2.1	Cells	
2.2	2.2	Transfection	
2.3	Mo	ouse model	49
2.3	5.1	Animals	
2.3	5.2	Tumour model	
2.4	An	alytical and functional assays	50
2.4	.1	Flow cytometric staining	
4	2.4.1.	1 Cell preparation:	
4	2.4.1.	2 Staining procedure	
2.4	.2	Reverse-transcriptase PCR	
2.4	.3	Sandwich ELISA for detection of chemokines	
2.4	4.4	Calcium mobilisation	
2.4	.5	In vitro proliferation assay	
2.4	.6	Soft agar assay	
2.4	1.7	Haemoglobin assay	
2.4	.8	Immunohistochemical staining	
2.4	.9	In vitro cytotoxicity assay	
2.4	.10	IL-17 ELISPOT	
2.4	.11	Intracellular cytokine staining	
2.4	.12	Cytokine bead array	
2.5	Sta	atistical analysis	
2.6	So	lutions and buffers	
2.6	5.1	Media	
-	2.6.1.	1 Alpha Medium (serum-free)	

,	2.6.1.2	Alpha Medium (complete)	58
,	2.6.1.3	2x Iscove's Modified Dulbecco's Medium (complete)	59
,	2.6.1.4	RPMI (incomplete)	59
,	2.6.1.5	RPMI (minimal complete)	59
,	2.6.1.6	RPMI (complete)	59
2.6	5.2 C	ther solutions and buffers	59
,	2.6.2.1	Buffer P1	59
,	2.6.2.2	Buffer P2	60
,	2.6.2.3	Buffer P3	60
,	2.6.2.4	Calcium Buffer	60
,	2.6.2.5	Collagenase Solution	60
,	2.6.2.6	Gill's Haematoxylin	60
,	2.6.2.7	10x Hank's Balanced Salt Solution	60
,	2.6.2.8	Mouse Red Cell Removal Buffer (MRCRB)	61
,	2.6.2.9	PBS/azide (PA)	61
,	2.6.2.10	PBS/BSA/azide (PBA)	61
<u>CHAP</u>	TER 3	: Generation of Chemokine Constructs & Cell Lines	69
3.1	Overv	iew	69
3.2	Gener	ation of chemokine constructs	69
3.2	2.1 C	hemokine constructs used in this study	69
3.2	2.2 C	loning of chemokine constructs	70
3.3	The 4	Γ1.2 model system	72
3.4	In vitr	<i>o</i> generation and characterisation of chemokine construct-expres	sing
	cell l	ines	74
3.4	.1 G	eneration of chemokine construct-expressing cell lines	74
3.4	.2 Iı	vitro characterisation of chemokine construct-expressing cell lines.	75
3.5	Summ	ary of findings and preliminary discussion	76
СНАР	TER 4	• The Effect of Chemokine expression on breast cancer	in
	1 1/11 7	, the little of Chemokine capiession on breast cancer	105
<u>vivo</u>			105
4.1	Overv	iew	105

4.2	Effects of exogenous chemokine expression on primary tumour growth 105
4.3	Effects of CXCL12 construct expression on metastasis106

4.4	Effects of CXCL16 construct expression on metastasis
4.5	Summary and preliminary discussion108
<u>CHAP</u>	TER 5: The Effect of CXCL12 expression on the anti-tumour
<u>immur</u>	ne response 125
5.1	Overview125
5.2	Identification of leukocyte subsets important for the anti-tumour effect of
	CXCL12
5.2	.1 The role of T cells
5.2	.2 The role of dendritic cells
5.2	.3 The role of NKT cells
5.3	The role of cell-mediated immunity128
5.4	The role of cytokines
5.5	Summary of findings and preliminary discussion
CHAP	TER 6: Discussion163
6.1	Introduction
6.2	The effects of CXCL12 overexpression on breast tumour progression 164
6.2	.1 The effects of CXCL12 on tumour cells and the tumour vasculature 164
6.2	.2 The importance of T cells and dendritic cells
6.2	.3 The role of cell-mediated cytotoxicity
6.2	.4 The role of cytokines
6.2	.5 The potential effects of CXCL12 on suppressor cells
6.2	.6 Other potential mediators of the effects of CXCL12
6.2	.7 Summary and future directions
6.3	The effects of CXCL12 _(P2G) overexpression on breast tumour progression177
6.4	The effects of CXCL16 construct overexpression on breast tumour
	progression
6.5	Perspectives and concluding remarks183
<u>CHAP</u>	TER 7: References193

Amendments

- p. xv "phospholipases C" should read "phospholipase C"
- p. 35 In Table 1.4, the heading of the second column, "Role of CXCL16CXCR6" should read, "Role of CXCL16/CXCR6"
- p. 53 In Section 2.4.4, the concentration of CXCL12 and CXCL16 used for stimulation of the cells should read "300ng/ml" and not "300µg/ml"
- p. 107 In the first line, "angiogenesis with the tumour" should read "angiogenesis within the tumour"
- p. 129 In the second line of the second paragraph, "4TX12 1 tumour-bearing were" should read "4TX12 1 tumour-bearing mice were"
- p. 131 In the third-to-last line, "Type I IFN-γ-producing also" should read "Type I IFN-γproducing T cells also"

Table of Figures

Figure 1.1 The hallmarks of breast cancer.	37
Figure 1.2 The steps of breast cancer metastasis.	38
Figure 1.3 Immunosuppressive networks induced by breast tumours	40
Figure 1.4 Chemokine signalling.	41
Figure 2.1 The pEF-IRES-puro6 (pEF) mammalian expression vector	65
Figure 3.1 Amino acid sequences of chemokine constructs	80
Figure 3.2 Cloning of chemokine constructs	82
Figure 3.3 Generation of the CXCL16 ₍₉₋₂₂₀₎ -His construct by overlap extension PCR	84
Figure 3.4 Vector maps of chemokine constructs	86
Figure 3.5 Expression of CXCL12 by 4T1.2 cells	87
Figure 3.6 Expression of CXCL16 by 4T1.2 cells	89
Figure 3.7 Expression of CXCR4 and CXCR7 by 4T1.2 cells	90
Figure 3.8 Expression of CXCR6 by 4T1.2 cells.	92
Figure 3.9 Expression of chemokine constructs by transfected 4T1.2 cell populations	94
Figure 3.10 Level of CXCL12 construct production by individual clones from transfe	cted
4T1.2 populations	95
Figure 3.11 Level of CXCL16 construct production by individual clones from transfe	cted
4T1.2 populations	96
Figure 3.12 CXCL12 construct expression by cell lines derived from pooled clones	97
Figure 3.13 CXCL16 construct expression by cell lines derived from pooled clones	98
Figure 3.14 In vitro proliferation rates of CXCL12-expressing 4T1.2 cell lines.	.100
Figure 3.15 In vitro proliferation rates of CXCL16-expressing 4T1.2 cell lines	.101
Figure 3.16 In vitro tumorigenicity of chemokine construct-expressing cell lines	.102
Figure 4.1 Comparison of the growth of CXCL12 construct-expressing 4T1.2 mamm	nary
tumours and parental 4T1.2 tumours in vivo.	.110
Figure 4.2 Comparison of the growth of CXCL12 construct-expressing 4T1.2 mamm	nary
tumours and control 4T12Ala tumours in vivo	.111
Figure 4.3 Growth of CXCL12-expressing 4T1.2 mammary tumours in vivo.	.112
Figure 4.4 Growth of CXCL12 _(P2G) -expressing 4T1.2 mammary tumours in vivo	.113
Figure 4.5 Growth of CXCL16-expressing 4T1.2 mammary tumours in vivo	.114

Figure 4.6 Spontaneous metastasis to the lungs of 4TX12 1 and 4T12P2G tumour-bearing
mice
Figure 4.7 Haemoglobin content of CXCL12-construct expressing tumours
Figure 4.8 Vascularisation of CXCL12 construct-expressing tumours
Figure 4.9 Lymphangiogenesis within CXCL12 construct-expressing tumours
Figure 4.10 Experimental metastasis of CXCL12-expressing tumour cell lines to the lungs
of Balb/c mice
Figure 4.11 Metastases in the lungs of $4T16$ and $4T\Delta 16$ tumour-bearing mice 122
Figure 5.1 Growth of CXCL12-expressing tumours in SCID mice
Figure 5.2 Growth of CXCL12-expressing tumours cells in nude mice
Figure 5.3 Correlation of the number and proportion of splenic T cells subsets with tumour
weight
Figure 5.4 Characterisation of T cell subsets from early stage tumours and tumour draining
lymph nodes
Figure 5.5 T cell infiltration of CXCL12-expressing 4T1.2 tumours at early time points.
Figure 5.6 CD4 ⁺ T cell infiltration of late stage CXCL12-expressing 4T1.2 tumours 145
Figure 5.7 CD8 ⁺ T cell infiltration of late stage CXCL12-expressing 4T1.2 tumours 147
Figure 5.8 Accumulation of CD11c ⁺ cells in the tumour-draining lymph nodes
Figure 5.9 CD11c ⁺ cells within CXCL12-expressing tumours
Figure 5.10 Growth of CXCL12-expressing tumours in invariant NKT cell-deficient mice.
Figure 5.11 Cytotoxic activity of lymphocytes against 4T1.2 tumour cells in vitro 154
Figure 5.12 Growth of CXCL12-expressing tumours in perforin- and TRAIL-deficient
mice
Figure 5.13 Growth of CXCL12-expressing tumours in IFN-γ knockout mice
Figure 5.14 Quantitation of IL-17-producing cells in tumour-draining lymph nodes by
ELISPOT assay
Figure 5.15 Intracellular cytokine staining of lymphocytes isolated from tumour-draining
lymph nodes
Figure 5.16 Cytokine production by cells isolated from tumour-draining lymph nodes. 159
Figure 6.1 Working model of CXCL12 action in primary mammary tumours
Figure 6.2 Working model of CXCL12 _(P2G) action in mammary tumour metastasis 188

List of tables

Table 1.1 Chemokine nomenclature	31
Table 1.2 The chemokine receptors and their biological functions.	33
Table 1.3 The biological functions of CXCL12 and CXCR4	34
Table 1.4 The biological functions of CXCL16 and CXCR6	35
Table 2.1 Primers used in this study	62
Table 2.2 Antibodies used in this study.	63
Table 2.3 Other reagents used in this study.	64
Table 3.1 Chemokine constructs and their functions.	79
Table 5.1 Comparison of tumour growth inhibition by CXCL12 in wild-type an	d nude
mice.	137
Table 5.2 Correlation of splenic T cell populations with final tumour weight	137

Abbreviations

300.19	a pre-B lymphocyte cell line
4T12Ala	4T1.2 cells transfected with the CXCL12(Ala)::pEF DNA construct
4T12P2G	4T1.2 cells transfected with the CXCL12(P2G)::pEF DNA construct
4TX12	4T1.2 cells transfected with the CXCL12::pEF DNA construct
4T16	4T1.2 cells transfected with the CXCL16::pEF DNA construct
4TΔ16	4T1.2 cells transfected with the CXCL16 ₍₉₋₂₂₀₎ ::pEF DNA construct
ADAM10	a disintegrin and metalloproteinase 10
ADCC	antibody-dependent cell-mediated cytotoxicity
APC	antigen presenting cell
APC-Cy7	allophycocyanin-cytochrome 7
ASMC	aortic smooth muscle cell
bFGF	basic fibroblast growth factor
BCIP/NBT	5-bromo-4-chloro-3' indolylphosphate p-toluidine salt/Nitro Blue tetrazolium
BCX6	300.19 cells transfected with a CXCR6::pEF DNA construct
BSA	bovine serum albumin
CAM	cell adhesion molecule
CCL	CC chemokine
CCR	CC chemokine receptor
CTL	cytotoxic T lymphocyte
CXCL	CXC chemokine
CXCR	CXC chemokine receptor
DAB	3,3'-diaminobenzidine
DC	dendritic cell
DLN	draining lymph node
EC	endothelial cell
ECM	extracellular matrix
EF-1α	elongation factor-1a
EGF	epidermal growth factor
ELISA	enzyme-linked immunosorbent assay
ELISPOT	enzyme-linked immunosorbent spot assay
EMMPRIN	extracellular matrix protease inducer
EPC	endothelial progenitor cell
ER	oestrogen receptor
ERK	extracellular signal-regulated kinase

FCS	foetal calf serum
FITC	fluorescein isothiocyanate
GAPDH	glutaraldehyde 3-phosphate dehydrogenase
GM-CSF	granulocyte-macrophage colony stimulating factor
GRK	G protein-coupled receptor kinase
HBSS	Hank's balanced salt solution
HD	Hodgkin's disease
HER2	human epidermal growth factor receptor 2
HIF-1a	hypoxia-inducible factor-1a
HIV	human immunodeficiency virus
HRP	horseradish peroxidase
HSC	haematopoietic stem cell
HUVEC	human umbilical vascular endothelial cell
IEL	intraepithelial lymphocyte
IFN-γ	interferon-γ
IKDC	interferon-producing killer dendritic cell
IL	interleukin
iNOS	inducible nitric oxide synthase
IRES	internal ribosome entry site
JAK	janus kinase
LDL	low density lipoprotein
LN	lymph node
MAPK	mitogen activated protein kinase
МНС	major histocompatibility complex
MMP	matrix metalloproteinase
MQ H ₂ O	MilliQ H ₂ O
MRCRB	mouse red cell removal buffer
MSC	myeloid suppressor cell
NK cell	natural killer cell
NO	nitric oxide
NPC	nasopharyngeal carcinoma
oxLDL	oxidised low density lipoprotein
PA	PBS/Azide
PBA	PBS/BSA/azide
PBL	peripheral blood leukocyte
PBS	phosphate buffered saline

PCR	polymerase chain reaction
pDC	plasmacytoid dendritic cell
PDGF	platelet-derived growth factor
PE	phycoerythrin
PE-Cy7	phycoerythrin-cytochrome 7
pEF	pEF-IRES-puro6
pfp	perforin
PGC	primordial germ cell
PI3K	phosphoinositol 3-kinase
PIP3	phosphatidylinositol 3,4,5-phospate
РКС	protein kinase C
PLC	phospholipase C
PMA	phorbol 12-myristate 13-acetate
PMS	N-methyl dibenzopyrazine methyl sulfate
PTEN	phosphatase and tensin homologue deleted on chromosome 10
RNAi	ribonucleic acid interference
RT-PCR	reverse transcription polymerase chain reaction
SCID	severe combined immunodeficient
SDF-1	stromal cell-derived factor-1
siRNA	small interfering ribonucleic acid
STAT	signal transducer and activator of transcription
TAA	tumour-associated antigen
TAM	tumour-associated macrophage
TCR	T cell receptor
TGF-β	transforming growth factor-β
Th	T helper
TLR	Toll-like receptor
TNF-α	tumour necrosis factor-α
Tr, T _{reg}	T regulatory
TRAIL	tumour necrosis factor-related apoptosis-inducing ligand
uPA	urokinase-type plasminogen activator
VCAM-1	vascular cell adhesion molecule-1
VEGF	vascular endothelial growth factor
VLA-4	very late antigen-4
XTT	sodium 3'-[1-(phenylaminocarbonyl)-3.4-tetrazolium]-bis (4-methoxy-6-nitro)
	benzene sulfonic acid hydrate

Publications arising from this work

Manuscripts

Sharon A. Hampton-Smith & Shaun R. McColl. 'Overexpression of CXCL12 in an orthotopic model of breast cancer improves the efficacy of the anti-tumour immune response.' *Manuscript in preparation*.

Conference Proceedings

Adelaide Breast Development and Cancer Meeting, 2006 (Adelaide, Australia).

Title: Chemokines in breast cancer: whose side are they on anyway?

Oral Presentation

Adelaide Immunology Retreat II, 2006 (Adelaide, Australia).

Title: The role of the chemokine CXCL12 in breast cancer.

Oral presentation

<u>Abstract</u>

A growing body of work implicates chemokines and their receptors in the progression of various types of cancer, including breast cancer. However, as potent chemotactic factors for leukocytes, chemokines also have the potential to enhance anti-cancer immunity. Evidence suggests that the chemokine CXCL12 and its receptors may be important in a number of aspects of breast cancer progression and site-specific metastasis. Another chemokine, CXCL16, has been identified as a specific chemotactic factor for Type Ipolarised T lymphocytes, which are major effectors of cell-mediated immunity and hence efficacious anti-tumour immune responses. The aim of this study, therefore, was to further elucidate the roles of CXCL12 and CXCL16 in breast cancer development and metastasis. To achieve this, wild-type CXCL12 and CXCL16 and antagonists of CXCL12 and CXCL16 activity, CXCL12_(P2G) and CXCL16₍₉₋₂₂₀₎ respectively, were overexpressed in the 4T1.2 mouse model of breast carcinoma. Overexpression of wild-type CXCL12 potently inhibited both primary tumour growth and metastasis in this model. This was attributed to the induction of an anti-tumour response dependent, in part, on T cells, interferon-y and the cytotoxic mediators perforin and TRAIL. This response was characterised by increased numbers of CD11c⁺ cells in the tumour-draining lymph nodes and enhanced cytolytic activity of lymph node-derived effector cells against tumour cells. Unexpectedly, CXCL12_(P2G) inhibited metastasis of tumour cells to the lungs of tumour-bearing mice, without affecting primary tumour growth. Intravenous injection of tumour cells revealed that CXCL12(P2G) expression could block metastatic steps occurring post tumour cell escape from the primary tumour, though a role for CXCL12_(P2G) at earlier metastatic steps could not be ruled out. Further work is needed to clarify the precise stages of metastasis at which CXCL12_(P2G) exerts its effects. No obvious effects on primary breast tumour growth were observed when CXCL16 or CXCL16₍₉₋₂₂₀₎ were overexpressed in tumour cells. Interestingly, CXCL16₍₉₋₂₂₀₎ expression inhibited experimental metastasis but not spontaneous metastasis. The findings of this study begin to shed light on the roles of CXCL12 and CXCL16 in breast cancer progression and also highlight the potential therapeutic applications of CXCL12, CXCL16 and/or their antagonists in the treatment of breast cancer and breast cancer metastasis.



CHAPTER 1

Introduction



CHAPTER 1: INTRODUCTION

1.1 Overview

Breast cancer is the most common malignancy occurring in women worldwide and represents approximately 1 million of the estimated 10 million cancers diagnosed in both sexes each year. It is the most common cause of cancer-related mortality in women throughout the world, claiming 375 000 lives in the year 2000³. Though it is primarily a disease found in women, breast cancer can also occur in men. The incidence in men is considerably lower than that in women and accounts for only 0.7% of breast cancer diagnoses⁴. However, the mortality rate of male breast cancer has not declined in recent years as has happened for female breast cancer, due to the lack of awareness and understanding of the disease⁵.

Compared with patients with other malignancies, those with breast cancer have a relatively good overall survival rate, particularly when the cancer is diagnosed early. However, prognosis is significantly poorer if the disease is more advanced upon presentation: in the USA in 1994, the 5 year survival rate for localised disease was reported to be 97%, while that for metastatic disease was just $25\%^3$. Thus, despite improvements in screening programs and early detection, more research is required to improve the outcome of patients presenting with advanced disease.

To this end, intensive efforts have been directed towards attaining a better understanding of the pathobiology of breast cancer: how it arises, evolves and progresses to metastatic disease. An improved knowledge of the mechanisms underlying metastasis in particular is essential, as this is the most dangerous aspect of breast cancer and also the most difficult to treat. Much research has focused on improving current therapies and developing new treatment modalities, and better understanding of the disease will provide insights into new therapeutic strategies that can be employed.

Immunotherapy holds great promise as a means to combat metastatic breast disease. The immune system is a powerful weapon ideally suited to targeting malignant cells: it has the ability to patrol the entire body for cancerous cells as well as the arsenal with which to destroy them. Harnessing the inherent power of the immune system of a patient is thus an

attractive prospect for treatment of breast cancer, and considerable research has focused on identifying ways to mobilise the anti-tumour response and overcome tumour immune evasion mechanisms.

In this vein, the chemokine family of chemotactic cytokines has been investigated as a potential source of immunotherapeutic mediators. The chemokines were originally identified based on their ability to guide the migration of leukocytes around the body, a property that makes them attractive as a means of targeting immune effector cells to the site of a tumour. Over the last six years, however, the chemokines have come under intense scrutiny for their role in promoting breast cancer progression. This followed the publication of a study by Muller and co-workers demonstrating that inhibiting the chemokine receptor CXCR4 could block breast cancer cell metastasis⁶.

In this thesis, the pro- and anti-tumour effects of two chemokines, CXCL12 and CXCL16, are investigated. This research is conducted with a view to better understanding the influence of these chemokines on breast cancer progression and to provide insight into their potential as therapeutic agents to treat this very prevalent disease.

1.2 Breast cancer

1.2.1 Breast cancer progression

The mechanisms of breast tumour development and metastasis are extremely complex. For many years, the study of breast cancer was limited largely to the properties of the cancerous cells within a tumour. More recently, the focus has shifted to encompass the other components within a tumour, namely stromal cells, endothelial cells (EC) and infiltrating leukocytes, as the significance of their contribution to the growth of cancerous cells has been realised. Not surprisingly then, a huge array of genes, signalling molecules, growth factors and cells have been implicated in the progression of malignant breast disease.

The entity that we recognise as breast cancer is likely to include a range of distinct diseases, each with their own individual aetiology. Despite the differences in molecular mechanisms that give rise to individual malignant breast lesions, there are a few

characteristics common to most, if not all breast cancers, which are required for tumour development and metastasis. These 'hallmarks' of breast cancer, as outlined by Hanahan and Weinberg¹, include the following: self-sufficiency in growth signals, insensitivity to anti-growth signals, evasion of apoptosis, limitless replicative potential (immortality), sustained angiogenesis and tissue invasion and metastasis (Figure 1.1). An additional hallmark, not specifically highlighted by Hanahan and Weinberg, is the requirement for 'normal' accessory cells in tumour development; cancerous cells must acquire the ability to co-opt and activate local stromal cells to promote malignant progression (Figure 1.1). Finally, the acquisition of genomic instability is crucial for the accumulation of the large number of genetic mutations required for the aforementioned characteristics to become manifest¹.

Of the eight hallmarks of cancer required for malignant progression, four of these (selfsufficient production of growth signals, insensitivity to anti-growth signals, evasion of apoptosis and immortality) directly pertain to the proliferative behaviour of tumour cells themselves. A well-studied example of a tumour-promoting growth signal in breast cancer is oestrogen. This has been demonstrated through success of the anti-oestrogen tamoxifen as a treatment modality for oestrogen receptor (ER)-positive breast cancers. Resistance to anti-growth signals can manifest as a result of mutations in any of a number of cell cycle regulators that allow malignant cells to progress through mitosis unchecked. For instance, deregulated expression of the cell cycle regulator Cyclin E is correlated with poor prognosis in human breast cancer⁷ and mammary carcinogenesis in transgenic mice⁸. Mutations in the tumour suppressor p53 are associated with enhanced resistance to apoptosis in a number of cancers, including breast cancer⁹. The p53 protein is responsible for inducing cell death in response to irreparable DNA damage and/or mutations¹⁰. Thus eliminating p53 function is a considerable advantage for breast tumour cells, which frequently harbour multiple genetic mutations. In terms of immortality, activation of telomerase is a significant event in the majority of breast cancers¹¹. Telomerase maintains the length of telomeres at the ends of chromosomes over many cell divisions, thus preventing end-to-end chromosome fusions, loss of karyotypic integrity and ultimately cell death. Telomerase is inactive in most normal adult cells, thus ensuring deletion of cells that have undergone multiple cell divisions and that potentially, have accumulated genetic mutations over time. Activation of telomerase allows breast cancer cells to continue dividing and accumulating genetic mutations, thus perpetuating the cancerous phenotype.

The importance of the enzyme was shown in a study where inhibition of telomerase increased the susceptibility of breast cancer cells to anti-cancer drugs in a manner that was dependent on the shortening of telomeres¹².

As well as altering their own behaviour, breast cancer cells must co-opt other normal cells to promote tumour progression. Tumour cell secretion of the cytokine transforming growth factor (TGF)- β is thought to promote activation and conversion of fibroblasts to a promalignant phenotype¹³. Although TGF- β has been shown to inhibit proliferation and transformation of normal breast epithelial cells, the activation of other oncogenic pathways can block the response of transformed cells to TGF-B inhibition and enhance the protumorigenic properties of this cytokine¹⁴. At later stages of tumour growth, TGF- β can promote epithelial-to-mesenchymal transition of malignant cells, enhancing their invasive capabilites¹⁴, and as mentioned above, TGF- β can enhance the pro-malignant activity of tumour stromal cells^{13, 14}. These stromal cells may provide necessary growth factors as well as depositing ECM components that promote tumour growth and invasion¹³. Another stromal cell type important in breast tumour development is the tumour-associated macrophage (TAM), which can supply extracellular matrix (ECM)-degrading enzymes needed for tumour invasion¹⁵. In addition to promoting invasion, TAM-derived ECMdegrading enzymes can release growth factors sequestered in the surrounding ECM. TAM also promote breast tumour growth directly, through production of mitogenic cytokines such as epidermal growth factor $(EGF)^{15}$.

Perhaps the most striking contribution of normal cells to the growth of cancerous cells is that of EC. As a solid tumour, breast cancer requires a vascular network to supply oxygen and nutrients to the growing tumour mass. Tumour angiogenesis can be triggered by a number of factors including low oxygen tension, low pH, hypoglycaemia, increased pressure due to cell proliferation, inflammatory cell infiltration and genetic mutations¹⁶. Tumour blood vessels do not resemble the normal vasculature: they are highly disorganised and excessively permeable. They lack functional smooth muscle cells, which are important for vessel contractility, and perivascular cells, which protect vessels from changes in the environment and provide cues to control their metabolic needs¹⁶. Vascular endothelial growth factor (VEGF) is a major angiogenic cytokine. It promotes proliferation of EC and permeability of capillaries that are required for expansion and remodelling of

the vasculature. VEGF expression is induced by hypoxia-inducible factor (HIF)-1 α under hypoxic conditions, as are often found in the centre of the growing tumour¹⁷. However, the dependence of breast tumour cells on VEGF is not absolute. Tumour cells can co-opt other angiogenic factors, such as basic fibroblast growth factor (bFGF) and platelet-derived growth factor (PDGF) for neovascularisation. Alternatively, some tumours can develop the ability to survive even in the absence of an ample blood supply¹⁷.

The real danger of breast cancer lies in its ability to metastasise: distant metastases are the ultimate cause of 90% of cancer deaths¹. Breast cancer commonly metastasises to the regional lymph nodes (LN), bone marrow, lung and liver⁶. Metastasis is a complex process of multiple steps, all of which are required for malignant cells to form secondary tumours. Malignant cells first need to escape from the primary tumour and invade surrounding tissue. This requires the expression of degradative enzymes, such as matrix metalloproteinases (MMP)¹⁸ and urokinase type plasminogen activator (uPA)¹⁹, that can break down ECM components and basement membranes. In addition, detached cancer cells must survive anoikis, a form of apoptosis triggered by the loss of anchoragedependent survival signals. One way in which this can be achieved by breast cancer cells is through inactivation of E-cadherin and $p53^{20}$. Interestingly, high level expression of extracellular matrix protease inducer (EMMPRIN), which can stimulate the production of MMP by both tumour and stromal $cells^{21}$, enhances the resistance of tumour cells to anoikis²². Thus expression of EMMPRIN might promote the invasive capacity of breast tumour cells by providing a means of moving through tissues, as well as the ability to survive whilst doing so.

Invasive capacity is considered to be the precursor of distant metastasis. Invasion of local tissue is required for malignant cells to reach a conduit for their passage to secondary sites. Two types of conduit are available for tumour cell metastasis (Figure 1.2): the blood vasculature (haematogenous metastasis) and the lymphatic vasculature (lymphatic metastasis). Angiogenesis, then, can serve a dual role in the progression of breast cancer by supplying oxygen and nutrients to the primary tumour, and by providing easy access to the circulation for metastatic cells. Upon arrival at a secondary tissue, a tumour cell needs to express appropriate adhesion molecules to permit it to adhere to the tissue endothelium and extravasate. Degradative enzymes are also required for the cell to cross the endothelial

basement membrane and migrate within the new tissue. Finally, to become established in its new niche, the tumour cell must be able to adhere to the local matrix components and to proliferate in its new environment to form a secondary tumour¹. Lymphatic metastasis is the most direct route for breast tumour cells to reach the regional LN and induction of lymphangiogenesis by breast tumours has been shown to enhance metastasis to the LN^{23} . Once in the LN, cancer cells can proliferate to form a secondary tumour from which metastatic cells can be shed. These cells eventually reach the bloodstream, thus embarking on the haematogenous route of metastasis to more distant anatomical locations. Indeed, enhanced lung metastasis in the absence of increased tumour angiogenesis has been attributed to the increase in lymphatic metastasis²³.

1.2.2 The immune response to breast cancer

One important aspect of breast tumour progression not included in the list of hallmarks outlined by Hanahan and Weinberg¹ is the ability to avoid destruction by the immune system. Tumour growth is an inherently inflammatory process because it inevitably involves tissue disruption. This leads to the release of pro-inflammatory signals in the form of cytokines and ECM fragments²⁴. To continue growing to a clinically detectable size, a tumour must somehow prevent the mobilisation of a detrimental anti-tumour immune response, and a variety of mechanisms that achieve this end have been reported. Some tumours can even subvert the immune response to their advantage, using growth factors and degradative enzymes released by inflammatory cells to further their growth and invasion. However, successful evasion of immune-mediated eradication is likely to be the exception rather than the norm. It is thought that many tumours arise and are eliminated before they become evident due to constant surveillance of the tissues for signs of abnormality by the immune system²⁴.

1.2.2.1 Immunosurveillance

The theory of tumour immunosurveillance has enjoyed a renaissance in recent years. According to this hypothesis, tumours arise in tissues at a similar frequency to pathogenic infections, but are recognised and eliminated by the immune system before they become clinically evident. It is thought that intraepithelial lymphocytes (IEL), which are dispersed throughout epithelial tissues, may be responsible for surveying these tissues for transformed cells²⁴. NKT cells²⁵ and innate immune cells, including eosinophils²⁶ natural killer (NK) cells and possibly macrophages²⁵, also play a part in tumour immunosurveillance. IFN- γ and intact cytolytic functions are required for this process, as has been determined from studies in IFN- $\gamma^{-/-}$, perforin^{-/-} and tumour necrosis factor-related apoptosis-inducing ligand (TRAIL)^{-/-} mice²⁵. Recognition of nascent tumours is though to occur in two ways: through recognition of tumour-associated antigens (TAA) presented on major histocompatibility complex (MHC) molecules, and through recognition of tumour-expressed ligands of the NKG2D activating receptor expressed on NK cells and T cells²⁵. Clearly, however, immunosurveillance of tumours is not a completely efficient process, otherwise cancer would be a disease unknown to medicine. It is thought that the immune response against a tumour selects for cells that are inherently less immunogenic; such cells overcome the immune response and grow into a clinically detectable tumour using both passive and active immune evasion mechanisms.

1.2.2.2 Immune evasion

Passive resistance mechanisms utilised by tumours can include down-regulation of MHC molecules or particular TAA^{24, 25}. In human breast cancer, total loss or reduced expression of MHC class I antigens was frequently observed in primary tumours and LN metastases^{27, 28}, whereas these antigens were present in all normal breast tissue and benign lesions examined²⁷. Moreover, mice receiving immunotherapy targeting the human Mucin 1 (MUC1) TAA escaped immune control through down-regulation of MHC class I molecules²⁹. Mammary tumours from neu-transgenic mice transplanted into non-transgenic mice induced a strong anti-tumour response; however, escape variants emerged that had lost expression of the neu TAA³⁰. This was attributed to the action of T cell-derived IFN- γ , because IFN-y treatment of mammary tumour cells induced methylation of the neu promoter³¹. These antigen-loss variants also exhibited reduced expression of proinflammatory cytokines, chemokines and immunostimulatory chaperones and increased expression of tumour survival factors³², suggesting that immune escape requires more than simple down-regulation of TAA. Indeed, down-regulation of MHC molecules and TAA would be expected to protect the tumour from T cell responses, but not from NK cellmediated lysis, which is triggered by the loss of MHC molecules, and other innate immune

responses that are antigen-independent. Additional mechanisms are required, therefore, for a tumour to successfully evade the immune response.

1.2.2.3 Active Immunosuppression

As an alternative to 'hiding' from the immune system, tumours may actively blunt or disarm the immune response (Figure 1.3). Cytokines are important factors responsible for dampening immune responses. These molecules play a significant role in determining the polarisation and effector function of tumour-infiltrating leukocytes. Cytokines such as IL-12 and IFN- γ are responsible for the differentiation of T helper 1 (Th1) cells, the development of immunogenic dendritic cells (DC) and the activation of cytotoxic macrophages³³. IFN- γ is also a major effector cytokine of CD8⁺ effector T cells and NK cells³⁴. It is therefore in the interests of the tumour to steer the cytokine balance away from this pro-immunogenic profile towards a more immunosuppressive profile. Such a cytokine profile induces the accumulation of regulatory T cells and other suppressor cells, and generates tolerogenic antigen presenting cells (APC) and pro-tumorigenic macrophages, the net effect of which is to enhance tumour growth while dampening deleterious immune responses.

Interleukin (IL)-10 and TGF- β are the two major cytokines associated with immunosuppression. IL-10 drives the differentiation of tolerogenic DC, which in turn produce more IL-10. This enables tolerogenic DC to induce T regulatory 1 (Tr1) cells, which also produce IL-10³⁵. IL-10 promotes production of TGF- β by T_{reg} cells and enhances the responsiveness of T cells to the inhibitory effects of this cytokine³⁵. However, it should be noted that the immunosuppressive role of IL-10 in anti-tumour responses is controversial³⁵. In breast cancer, depletion of IL-10-producing $\gamma\delta$ T cells was found to enhance anti-tumour cytotoxic T lymphocyte (CTL) and NK cell cytotoxic activity³⁶, while in another model, IL-10-transfected tumours were rejected by the immune system³⁷. It is thought that the location of its expression determines how IL-10 influences the immune responses in the secondary lymphoid organs, but localised IL-10 expression within the tumour enhances anti-tumour immunity³⁵.

TGF- β is an important differentiation factor for T regulatory (T_{reg}) cells, and is also a major product of these cells. As such, it is an important mediator of immune suppression. TGF- β inhibits T cell proliferation and differentiation of naïve T cells into Th1, Th2 or CTL effector cells, as well as B cell proliferation and activation³⁸. TGF- β also inhibits macrophage effector functions and maturation of DC, which results in reduced antigen presentation by these cells³⁸. The significance of TGF- β in breast cancer immunosuppression has been shown by studies where neutralisation of TGF- β restored T cell activity and cytotoxic function³⁹⁻⁴¹ and enhanced the efficacy of DC-based vaccines⁴².

A critical factor determining the outcome of an anti-tumour immune response (i.e. immunogenic versus tolerogenic) is the type of immunostimulatory signals provided by APC, particularly DC. The nature of these signals depends on both the development and activation state of DC^{43} . In breast cancer patients, the number of mature DC infiltrating tumours was correlated with improved relapse-free and overall survival⁴⁴. However, maturation of DC does not always generate effective immunostimulatory APC, particularly when induced by tumour-cell derived factors⁴⁵. Whether a DC induces activated or suppressive T cells seems to depend more upon the microenvironment in which the DC finds itself (e.g. surrounded by 'danger' signals in the form of TLR ligands versus an IL-10-rich milieu) and the kind of cytokines the DC itself secretes (e.g. IL-12 versus IL-10)⁴³.

Much attention has focused on the role of regulatory T cells (T_{reg}) in tumour-induced immunosuppression. T_{reg} cells are increased in breast cancer patients⁴⁶, and depletion of these cells improves anti-tumour responses in mouse models of breast cancer^{47, 48}. T_{reg} can be classified broadly into 'natural' and 'induced' T_{reg} on account of their developmental pathway. Natural T_{reg} are generally CD4⁺CD25⁺FoxP3⁺ and develop normally in the thymus as a distinct T cell population. Induced T_{reg} differentiate from naïve T cells in response to antigen stimulation in the absence of co-stimulatory molecules and/or the presence of cytokines such as IL-10 and TGF- β^{49} . These cells constitute a number of regulatory T cell types, including Tr1 cells, T helper 3 (Th3), some $\gamma\delta$ T cells and CD8⁺CD28⁻ T cells. Also, natural T_{reg} -like cells, which express FoxP3, can be induced to differentiate from naïve T cells⁵⁰. T_{reg} are able to inhibit the functions of a variety of

leukocytes from both the innate and adaptive arms of immunity. They can suppress the activation and/or proliferation of naïve $CD4^+$ and $CD8^+$ T cells, and to a lesser extent, the function of effector T cells. They also inhibit the proliferation and effector functions of B cells, the cytolytic functions of NK and NKT cells, and the function and maturation of DC. This suppression can be mediated in various ways: through direct cell-cell contact and accessory molecules such as CTLA-4 and lymphocyte activation gene 3 (LAG3), through secretion of immunosuppressive cytokines such as IL-10 and TGF- β , and also through cytotoxic activity against effector T cells⁵⁰.

Recently, a new cell type has come to the fore as a mediator of tumour immunosuppression. Myeloid suppressor cells (MSC), defined as CD11b⁺Gr1⁺ or Gr1⁺CD115⁺F4/80⁺ in mice and CD34⁺CD33⁺CD15⁻CD13⁺ in humans, are a heterogeneous population of immature myeloid cells and myeloid precursors that accumulate in tumour-bearing patients and animals⁵¹⁻⁵³. MSC can suppress anti-tumour immune responses either at the tumour site, or by influencing T cell priming in secondary lymphoid organs⁵¹. They impair T cell function through expression of arginase and inducible nitric oxide synthase (iNOS), the products of which can induce down-regulation of the T cell receptor (TCR) ζ -chain^{54, 55}, nitration of the TCR and inhibition of its peptide-MHC recognition capacity⁵⁶, and blockade of signalling intermediates required for the response of T cells to IL-2⁵⁷. In a separate mechanism, MSC can suppress immune responses through production of TGF- β , which directly suppresses CD8⁺ CTL⁵⁸. In addition, MSC-derived IL-10 induces the development of antigen-specific T_{reg} cells, indicating that MSC can suppress immune responses indirectly⁵¹. MSC can inhibit NK cells as well, impairing their cytotoxicity by blocking their production of perforin and by inhibiting their ability to respond to $IL-2^{59}$.

1.2.2.4 Overcoming tumour-induced immunosuppression

Overcoming the immunosuppressive mechanisms of tumour cells is a key goal of immunotherapy. A variety of tactics have been employed to accomplish this and elicit effective tumour-specific immune responses. One strategy is to passively immunise breast cancer patients with monoclonal antibodies specific for breast TAA (e.g. trastuzumab (Herceptin), which targets the human epidermal growth factor receptor 2 (HER2) TAA⁶⁰). In addition, a number of trials investigating the potential of cell-based vaccines, in

particular DC-based vaccines, have been conducted⁶¹. As yet, these therapies have achieved only modest success^{60, 61}, and further work is underway to optimise the induction of anti-tumour immunity.

One potential way of enhancing the efficacy of vaccination is to promote the colocalisation, and hence activation, of immune effectors such as DC and T cells, either in tumours or in secondary lymphoid organs. As chemotactic factors for leukocytes, chemokines are attractive candidates to achieve this end, and a few studies examining their potential in breast cancer therapy have been published⁶²⁻⁶⁷. The following Section provides an overview of the chemokines and describes the general properties and functions that have brought them to the attention of tumour immunologists.

1.3 The chemokine family

1.3.1 General properties of chemokines

The chemokines are a family of low molecular weight polypeptide chemotactic cytokines. This family is divided into four subgroups, CXC, CC, C and CX₃C chemokines, based on the number and position of the conserved cysteine residues at the amino-terminal end of the peptide⁶⁸ (Table 1.1). Chemokines may also be classified according to their function and expression pattern. Homeostatic chemokines are constitutively expressed, while expression of inflammatory chemokines is up-regulated in response to inflammatory stimuli such as lipopolysaccharide, interleukin (IL)-1 and tumour necrosis factor (TNF)- α^{69} .

With over 50 members identified to date, the chemokine family has been found to play a multitude of roles in biology, including cell recruitment, leukocyte activation, lymphocyte development and trafficking, lymphoid organ development, development of cell-mediated versus humoral immune bias, inflammation, wound healing, tumour metastasis, angiogenesis and angiostasis⁶⁸. They exert their effects through their interactions with seven transmembrane domain G protein-coupled receptors (Table 1.2). Chemokine-chemokine receptor interactions can be quite promiscuous, with individual chemokines often binding multiple receptors, and receptors binding multiple chemokines. However, each receptor binds a unique panel of ligands^{69, 70}. Although in recent years the sphere of

influence of the chemokine family has been found to encompass a wide variety of cell types, much of what is known about chemokine biology comes from studies of their effects on leukocytes. The activity of chemokines on leukocytic and non-leukocytic cell types is generally similar, but it should be kept in mind that extrapolation of the findings in leukocytes to non-leukocytic cell types may not be always valid.

1.3.2 Chemokine receptor signalling

Chemokines generally act as monomers with their N-termini governing the characteristics of receptor binding. The amino-terminus is typically the critical mediator of receptor triggering. Its length determines the ability of a chemokine to bind its receptor, and whether it will act as an agonist or an antagonist⁶⁹. Binding of a chemokine to its receptor induces a conformational change that leads to dissociation of the receptor-associated heterotrimeric G proteins into α and $\beta\gamma$ subunits (Figure 1.4). These G protein subunits can then activate various effector enzymes, including phospholipases, which leads to inositol phosphate production and an increase in intracellular calcium, and protein kinases⁷¹. Phosphoinositol 3-kinases (PI3K) are also important downstream effectors of chemokine signalling⁷², and both the class IA⁷³ and class IB⁷⁴ PI3K have been implicated in chemokine signalling. Activation of PI3K leads to induction of the downstream effectors Rho, Rac and Cdc42⁷⁵. This results in modulation of actin-dependent cellular processes and increased expression and clustering of adhesion molecules⁷⁶, ultimately leading to cell chemotaxis.

Mitogen activated protein kinases (MAPK), also known as extracellular-signal regulated kinases (ERK), are activated in response to chemokine binding in a manner dependent on Gβγ signalling through Ras² (Figure 1.4). MAPK/ERK signalling is required for a variety of chemokine-induced responses, including degranulation⁷⁷, cell adhesion⁷⁸, chemotaxis⁷⁹, anti-apoptotic responses⁸⁰ and gene expression⁸¹. The requirement for chemokine-induced MAPK/ERK signalling does not seem to be absolute², however, and may depend on the particular cell type, the receptor activated and the functional response being examined.

Chemokine binding also induces activation of the JAK/STAT pathway (Figure 1.4). JAK1, 2 and 3 and STAT1, 3 and 5 have been implicated in chemokine receptor signalling, though the particular JAK and STAT activated may depend on the receptor transducing the

signal and/or the cell type examined⁸²⁻⁹⁰. JAK/STAT activation seems to be independent of $G\alpha_i$ activation^{84, 88, 90}; however, whether or not JAK activation is required for $G\alpha_i$ activation seems to depend on the particular chemokine receptor studied^{84, 91}. The biological significance of chemokine-induced JAK/STAT signalling has not been well characterised, but it appears to be important for cell adhesion and chemotaxis^{83, 87}, angiogenesis⁸³ and for induction of gene transcription⁸⁹.

Another outcome of chemokine signalling is receptor internalisation (Figure 1.4). This can occur via one of two pathways: clathrin-mediated endocytosis or lipid raft/caveolae-dependent internalisation⁹². Ligand binding to the receptor induces phosphorylation of serine and threonine residues in the C-terminal tail and intracellular loops by G protein-coupled receptor kinases (GRK). Adaptor proteins such as β -arrestin and adaptin-2 bind to these phosphorylated residues and link the receptors to a clathrin lattice. This enables receptor endocytosis, a process that requires the GTPase activity of dynamin. Once within the endosome, the receptor is dephosphorylated and either degraded, or recycled to the cell surface⁹². Little is known about the mechanism of lipid raft/caveolae-dependent internalisation, but CCR4, CCR5 and CXCR4 have been shown to use this internalisation route, at least in some cell types⁹². The functional role of receptor internalisation remains controversial, but a recent study suggests that it may be important for the fine-tuning of cell migration⁹³.

1.3.3 Biological functions of chemokines

Chemokines were initially identified as chemotactic factors for leukocytes, and accordingly, the way in which they influence this process has been the subject of intensive study. A critical requirement for chemotaxis is cell polarisation. Chemokine signalling induces actin polymerisation, which is manifested in the extension and retraction of lamellipodia, the driving force of cell movement, at the leading edge of the cell⁹⁴. Polymerisation of actin is the result of the accumulation of phosphatidylinositol 3,4,5-phospate (PIP3), a product of PI3K, at the leading edge. There, PIP3 activates Rac and Cdc42, which in turn polymerise actin and contribute to cell polarisation and chemotaxis⁹⁴. The negative regulator of PIP3 accumulation, phosphatase and tensin homologue deleted on chromosome 10 (PTEN), is restricted to the rear of the cell, thus ensuring a strong differential in PIP3 concentration within the cell and amplifying cell

polarity⁷². Chemokine signalling also induces expression of integrins on the cell surface and their clustering at the leading edge, again contributing to cell polarity⁹⁴. This facilitates adhesion of the cell to surfaces and provides the traction necessary for migration along a chemokine gradient.

The mechanism by which chemokines facilitate leukocyte adhesion to endothelium and transendothelial migration has been well studied⁹⁵. The egress of a leukocyte from the circulation is a process consisting of precise stages, namely tethering, rolling, activation, adhesion, margination, diapedesis (or extravasation) and migration within the tissue. Tethering and rolling are mediated by the selectin family of adhesion molecules expressed either by leukocytes (L-selectin) or the endothelium (E- and P-selectins). Chemokines are immobilised on the endothelial surface, and may be expressed constitutively by the tissue (e.g. CCL19 and CCL21 in the LN) or in response to inflammation (e.g. during tissue damage or infection). Chemokines signal a rolling leukocyte to up-regulate and activate integrins expressed on its surface, increasing their affinity for cell adhesion molecules (CAM) expressed by EC. This causes the cell to stop rolling and enables it to adhere firmly to the endothelium. The cell flattens itself against the endothelial surface ('margination') as it engages an increasing number of integrin-CAM interactions, before beginning diapedesis through the endothelial barrier into the tissue. Chemokines immobilised on ECM components guide the migration of the leukocyte within the tissue⁹⁵.

Chemokine signalling not only promotes cell migration, but also activates the effector function of various leukocytes. Chemokines can induce the production of microbicidal oxygen free radicals and enhance phagocytosis by macrophages^{71,96}. They can also promote synthesis of bioactive lipid mediators^{71,96} and degranulation of neutrophils^{97,98}, eosinophils⁹⁹ and mast cells⁷⁵. Chemokine-induced degranulation results in the release of MMP¹⁰⁰, which facilitate degradation of ECM components and enable leukocytes to migrate within the tissues to sites of inflammation.

Another major biological role of chemokines is the modulation of angiogenesis. Originally, chemokines of the CXC family were classified as angiogenic or angiostatic based on the presence or absence, respectively, of a Glu-Leu-Arg (ELR) motif N-terminal to the first conserved cysteine residue⁶⁸. Two exceptions to this are CXCL12 and CXCL16, which are ELR-negative chemokines with pro-angiogenic properties. Moreover, it has been shown

that CCL21 is able to bind CXCR3 and exert angiostatic effects¹⁰¹. Thus it may be more meaningful to classify chemokines as angiogenic or angiostatic depending on which receptors they bind: chemokines that bind CXCR1, CXCR2, CXCR4 and CXCR6 promote angiogenesis, while those binding CXCR3 exert angiostatic effects. Apart from CCL21, other CC chemokines that influence angiogenesis include CCL1¹⁰², CCL2¹⁰³, CCL11^{104, 105}, CCL15¹⁰⁶, CCL16¹⁰⁷ and CCL23¹⁰⁸. All of these chemokines influence angiogenesis directly through interaction with their receptors expressed on EC. The CC chemokines are largely pro-angiogenic, with the exception of CCL11^{104, 105} and CCL21 (discussed above), which are angiostatic. The pro-angiogenic effects of chemokines can be attributed to their ability to promote migration, invasion and differentiation of EC expressing their cognate receptors^{102, 103, 106-111}. In some cases, chemokines can enhance EC production of other pro-angiogenic chemokines¹⁰⁷ and cytokines (e.g. VEGF¹¹¹), as well as enhance sensitivity of EC to VEGF signalling¹⁰⁷. Angiostatic chemokines generally regulate these same processes, but in a negative instead of a positive way^{105, 110}.

As mentioned previously, the chemokines CXCL12 and CXCL16 were selected for study in this thesis. Like the other members of the chemokine family, they mediate the basic cellular functions described above. Both chemokines are involved in the orchestration of immune responses, and both have been reported to facilitate cancer progression. The characteristics and biological functions of CXCL12 and CXCL16 are described more fully in the following sections, and the data published so far pertaining to their roles in cancer are summarised.

1.4 CXCL12/Stromal cell-derived factor (SDF)-1, CXCR4 and CXCR7

1.4.1 General properties of CXCL12 and its receptors

CXCL12, or stromal cell-derived factor (SDF)-1, is the most widely expressed chemokine. To date, six CXCL12 isoforms, which arise from a single gene through alternative splicing, have been identified in humans: CXCL12 α , CXCL12 β , CXCL12 γ , CXCL12 δ , CXCL12 ϵ and CXCL12 ϕ^{112} . All exhibit chemotactic activity *in vitro*, though their tissue distribution patterns differ. CXCL12 α and CXCL12 β are the most widely expressed isoforms, found in primary and secondary lymphoid organs, heart, lung, lung, liver, pancreas, placenta, skeletal muscle and kidney, but not peripheral leukocytes¹¹². Despite the variation in expression patterns, the functional differences, if any, between the CXCL12 isoforms are not clear. A comparison of the α and β isoforms revealed differences in proteolytic processing by dipeptidyl-peptidase/CD26¹¹³, which may differentially influence the halflife of the two isoforms and facilitate fine-tuning of the biological activity of CXCL12 *in vivo*. To date, only the CXCL12 α and β isoforms have been identified in the mouse.

CXCL12 is the ligand for the chemokine receptor CXCR4, which has attracted much attention for its role as a co-receptor for entry of X4 HIV strains into host cells¹¹⁴. Like its ligand, CXCR4 has a broad expression pattern, being found in brain, primary and secondary lymphoid organs, stomach, intestine and kidney. It is expressed on the majority of leukocytes, as well as on EC^{111, 115-118}, various epithelial cells¹¹⁹⁻¹²¹, cells of the central nervous system¹²² and primordial germ cells^{123, 124}. The expression of both CXCR4 and CXCL12 is regulated by hypoxia^{125, 126}. This is important for recruitment of haematopoietic progenitor cells to hypoxic niches in the bone marrow and for recruitment of circulating endothelial progenitor cells for neovascularisation and repair of ischaemic tissues¹²⁶. Hypoxic regulation of CXCR4 and CXCL12 also has significance for the growth and metastasis of tumours, which frequently become hypoxic. As discussed in Section 1.4.3, CXCR4 and CXCL12 both have pro-tumorigenic properties, and the up-regulation of this receptor/ligand pair by hypoxia may confer a growth advantage on malignant cells^{125, 127}.

For many years it was thought that the interaction between CXCL12 and CXCR4 was monogamous. However, recent work by two groups has identified the orphan G proteincoupled receptor RDC1 as a receptor for CXCL12^{128,129}. To acknowledge this, it has been proposed that RDC1 be renamed CXCR7 in accordance with the chemokine receptor nomenclature. Though there is agreement that CXCL12 does indeed bind CXCR7, controversy remains regarding the expression pattern of CXCR7 and the functional consequences of the CXCL12-CXCR7 interaction. One group reported that CXCR7 was expressed on various human leukocyte subpopulations and that CXCL12 could induce CXCR7-dependent chemotaxis of peripheral blood leukocytes^{128,130}. The second group, in contrast, identified surface CXCR7 expression only on tumour cell lines, human EC, and mouse foetal liver cells¹²⁹. They also reported that CXCL12 ligation of CXCR7 did not lead to the typical chemokine-induced responses of calcium mobilization and chemotaxis¹²⁹. The reason for the inconsistencies in the findings of the two groups is not apparent, and the precise function and expression pattern of CXCR7 await further clarification. What is clear is that work on CXCL12 and CXCR4, both previous and current, must be interpreted carefully in light of the identification of CXCR7 as a receptor for CXCL12.

A single amino acid substitution in the N-terminus of the CXCL12 peptide sequence converts CXCL12 into a potent antagonist of its wild-type activity. The CXCL12_(P2G) analogue of CXCL12 has an amino acid sequence identical to that of CXCL12 with the exception of a glycine substituted for the proline residue at position 2. This analogue has only 3-fold lower binding affinity for the CXCL12 receptor CXCR4 than wild-type CXCL12. However, the amino acid substitution results in a complete loss of activity, as shown by the inability of the analogue to induce calcium mobilization and receptor internalisation in the human CEM T cell line. Moreover, CXCL12_(P2G) is able to inhibit chemotaxis of CEM cells towards CXCL12¹³¹.

1.4.2 Physiological functions of CXCL12

1.4.2.1 Functions of CXCL12 in the haematopoietic system

The significance of the role of CXCL12 and CXCR4 in physiology is suggested by the high degree of evolutionary conservation across species: they share 98 and 94% homology between human and mouse, respectively⁶⁹. A CXCL12 homologue has also been identified in the lower vertebrate *Xenopus laevis*, which is capable of inducing the migration of human lymphoblastic cells expressing CXCR4¹³². Further support for the crucial biological role of this chemokine/receptor pair has come from studies of CXCL12 and CXCR4 knockout mice. For both, the genetic deletion is lethal, and the mice display similar severe disruptions to haematopoiesis, cardiac development¹³³ and vascularisation of the gastrointestinal tract^{134, 135}. In addition, CXCR4 knockout mice display abnormal cerebellar development¹³⁵.

Not surprisingly, then, multiple functions for CXCL12 and CXCR4 have been identified. Though it was originally characterised as a growth stimulatory factor for pre-B cells¹³⁶, CXCL12 has since been found to influence a variety of cells of both haematopoietic and non-haematopoietic origin (Table 1.3). CXCL12 is highly expressed in the bone marrow where it is critical for the engraftment¹³⁷ and retention¹³⁸ of haematopoietic stem cells (HSC). Blockade of CXCR4 by the small molecule inhibitor AMD3100 has been shown to be an effective strategy for mobilising HSC for transplantation¹³⁹⁻¹⁴¹. As well as stimulating growth of pre-B cells¹³⁶, CXCL12 and CXCR4 serve to retain B cell and myeloid progenitors within the bone marrow microenvironment¹⁴². Despite expressing CXCR4 throughout their maturation, B cells are only responsive to CXCL12 at certain stages of their development¹⁴³. This property is thought to enable B cells to migrate from the primary to the secondary lymphoid organs and to position themselves optimally within secondary lymphoid organs during activation¹⁴⁴.

CXCL12 plays a role in the survival, expansion and differentiation of early thymocytes within the thymus. Thymic precursors in foetal blood are responsive to CXCL12, which is thought to induce their migration to and population of the thymic anlage¹⁴⁵. Immature, but not mature, thymocytes respond to CXCL12¹⁴⁶, which may be important for their recruitment to the cortex for positive selection¹⁴⁷. There is also evidence to suggest that CXCL12 and CXCR4 are required for thymic egress. Blockade of CXCR4 or elimination of the CXCL12 gradient leads to retention of mature T cells within the thymus, indicating that CXCL12 may provide a chemorepellent signal to induce emigration of such cells to the periphery¹⁴⁸.

In addition to its effects on haematopoietic progenitors, B cells and thymocytes, CXCL12 is a potent chemotactic factor for mature lymphocytes, monocytes¹⁴⁹, mature DC¹⁵⁰, NK cells^{151,152}, NKT cells¹⁵³ and megakaryocyte progenitors¹⁵⁴. In mature T cells, the action of CXCL12 is not limited to induction of chemotaxis. This chemokine has been implicated in co-stimulation of T cells upon encounter with antigen, though the signalling mechanisms underlying this process are only just beginning to be elucidated. Recent work from Karen Hedin's lab has shown that CXCL12 stimulates physical association of CXCR4 and the TCR⁸¹. CXCL12 up-regulates activation markers on T cells in a CXCR4-dependent manner and promotes their proliferation and production of IFN- γ , IL-2 IL-4 and IL-10^{81,155}. In contrast to the findings outlined above, Peacock et al. found that anti-CD3 stimulation of CD4⁺ T cells reduced surface CXCR4 expression and inhibited CXCL12-induced chemotaxis¹⁵⁶. The reason for this discrepancy is not clear, however differences in
T cell activation protocols could be one explanation. Others have reported that CXCR4 is down-regulated upon TCR activation^{157, 158}, but in those experiments, CXCR4 expression was examined days after TCR stimulation, rather than within hours as in the studies described above. One study demonstrated that CXCL12 induced apoptosis of Jurkat T cells by induction of surface Fas/CD95 and intracellular FasL/CD95L expression¹⁵⁹. Since activation-induced cell death is important for the resolution of immune responses¹⁶⁰, it may be that one role of CXCL12 in T cell activation is to sensitise activated cells to apoptosis at late stages of the immune response. IL-2 is an important mediator of this phenomenon, in addition to its role in inducing clonal expansion of T cells at the initiation of an immune response¹⁶¹. Thus down-regulation of CXCR4 may be an important step to allow antigen-experienced T cells to avoid activation-induced cell death and instead enter the memory pool.

1.4.2.2 Functions of CXCL12 in non-haematopoietic tissues

The action of CXCL12 and CXCR4 is not restricted to the haematopoietic system (Table 1.3), as evidenced by the severe defects in multiple organs arising in CXCL12^{-/-} and CXCR4^{-/-} mice. Indeed, disruptions to the vascularisation of the gastrointestinal tract in these mice provided the first indication of a role for CXCL12 and CXCR4 in angiogenesis¹³⁴. Expression of both CXCL12^{117, 162} and CXCR4^{111, 115, 118} is detectable in EC; CXCL12 can also reciprocally up-regulate expression of VEGF^{116, 163}. CXCL12 is a potent chemotactic factor for EC¹⁶² and promotes endothelial tube formation *in vitro*^{116, 117, 162}, as well as neovascularisation in various *in vivo* models of angiogenesis^{111, 116, 117}. CXCL12 not only acts on differentiated EC but also recruits endothelial progenitor cells (EPC) to ischaemic tissues: CXCL12 gene transfer into ischaemic muscle tissues was found to mobilize EPC into the peripheral blood, increase capillary density in, and augment recovery of blood perfusion to, these tissues¹⁶⁴.

CXCL12 and CXCR4 are important for migration of a variety of non-haematopoietic progenitor cells. They are required for primordial germ cell (PGC) migration and colonisation of the gonads in both zebrafish^{165, 166} and mice^{123, 124}. Muscle progenitors express CXCR4 and CXCL12 is expressed along the migratory routes and at the target sites of these cells¹⁶⁷. Development of the retina is dependent on the interaction of CXCL12 and CXCR4, where they provide important guidance signals for axons. Unlike its

effect on leukocytes and other cells, CXCL12 does not seem to have any direct chemoattractive or chemorepulsive effects on axons, rather it acts by antagonising the repellent effects of Slit-2 in a CXCR4-dependent manner^{168, 169}.

Within the central nervous system, CXCL12 and CXCR4 are also key players, as is indicated by the developmental defects in the cerebellum of CXCR4 knockout mice. CXCR4 is expressed on a variety of neural cells, and its ligation by CXCL12 leads to enhanced migration and proliferation, stimulation of cytokine production and even the production of pain through excitation of nociceptive neurons (reviewed in ref. 122). Epithelial cells of the alveoli, bladder and mammary tissue are responsive to CXCL12 and express the CXCR4 receptor¹²⁰. Human colon epithelial cells also express CXCR4 and produce CXCL1 and CXCL8 in response to CXCL12 stimulation^{170, 171}. Moreover, down-regulation of CXCR4 induces differentiation of these cells, and may play a role in maintenance and renewal of the colonic epithelium¹⁷². CXCL12 may play a role in trophoblast survival during pregnancy as cultured human cytotrophoblasts treated with anti-CXCR4 antibody displayed reduced cellularity¹⁷³. CXCL12 is also important for recruitment and differentiation of osteoclast precursors^{174, 175}, and it has been associated with the osteolytic bone disease characteristic of multiple myeloma¹⁷⁶.

1.4.3 CXCL12 and CXCR4 in cancer

In a recent landmark study, CXCR4 was shown to be expressed by various human breast cancer cells and cell lines⁶. When injected either intravenously or into the mammary fat pad of severe combined immunodeficient (SCID) mice, these CXCR4-expressing cells metastasized to organs where CXCL12 was highly expressed, namely the lungs, liver, LN and bone marrow. This migration pattern mimics the metastatic pattern typical of human breast cancer. A subsequent study of human breast carcinomas found that increased expression of CXCR4 correlated with higher incidence of LN metastases¹⁷⁷. CXCR4 has since been found to be expressed on a variety of cancer cells and cell lines including cervical cancer¹⁷⁸, colorectal cancer^{179, 180}, head & neck cancer¹⁸¹, nasopharyngeal carcinoma¹⁸², non-small cell lung carcinoma¹⁸³, melanoma^{184, 185}, neuroblastoma^{186, 187}, oral squamous cell carcinoma^{188, 189}, pancreatic cancer^{190, 191}, renal carcinoma^{127, 192}, rhabdomyosarcoma¹⁹³ and prostate cancer^{194, 195}. Such widespread expression of CXCR4 axis

in malignant progression. The mechanisms by which this receptor-ligand pair influence tumour growth and metastasis are only beginning to be understood, but it is apparent that CXCL12 and CXCR4 have a variety of tumour-promoting functions. For the purposes of this thesis, these functions will be examined in the context of breast cancer, though they have been reported in other forms of cancer as well.

CXCL12 and its receptors are not only involved in processes specific to metastasis. Indeed, many of their functions promote primary tumour growth and progression in addition to facilitating the metastasis of malignant cells. Various studies have demonstrated a role for CXCL12 in promoting breast tumour cell proliferation¹⁹⁶⁻¹⁹⁹. Oestrogen can enhance tumour cell expression of CXCL12^{198, 200}, or CXCL12 can be supplied by stromal components to act in a paracrine fashion to enhance tumour cell proliferation^{196, 199}.

Another pro-tumour property of CXCL12 is its ability to promote cell survival and prevent apoptosis. For example, treatment of mice bearing glioblastoma or medulloblastoma with the CXCR4 inhibitor AMD3100 inhibited tumour growth *in vivo* by increasing apoptosis of tumour cells²⁰¹. Very little has been reported regarding the ability of CXCL12 to directly promote survival of breast cancer cells, although CXCL12 was found to prevent apoptosis of cultured MDA-MB-231 cells upon withdrawal of serum from the culture medium²⁰². Interestingly, introducing the novel atypical CXCL12 receptor, CXCR7, into the recently redefined human melanoma cell line MDA-MB-435²⁰³ enhanced cell survival and reduced apoptosis under low serum conditions *in vitro*¹²⁹. This cell line produces endogenous CXCL12²⁰⁴, indicating that the CXCL12-CXCR7 interaction could be a relevant means by which cancer cells gain a survival advantage. Specifically antagonizing CXCR7 inhibited tumour growth in mouse models of lung carcinoma and lymphoma¹²⁹, though this effect is yet to be reported in breast cancer.

A defining feature of malignant tumours is their ability to invade surrounding tissues. The ability of CXCL12 and CXCR4 to enhance breast tumour growth and metastasis is frequently associated with their ability to induce invasion of breast cancer cells *in vitro*^{197, 205}. CXCL12 can promote adhesion of breast cancer cells to the ECM components fibronectin and collagen²⁰⁶, and knockdown of either CXCL12²⁰⁴ or CXCR4^{197, 205} is sufficient to block invasion of breast cancer cells through Transwells coated with ECM. Invasion also may be facilitated through MMP induction: both MMP2 and MMP9 are

produced by breast cancer cells in response to CXCL12 stimulation²⁰⁷. In addition, CXCL12 can induce breast cancer cell invasion though endothelial monolayers in a CXCR4-dependent manner^{208, 209}. This highlights the potential role of CXCL12, acting through CXCR4, in the entry of metastatic cells into the circulation and their extravasation at secondary sites.

Although there is considerable evidence for the angiogenic function of CXCL12 and CXCR4, and indeed, that this function is a significant factor in the progression of various cancers, there is little direct evidence of a role for CXCL12/CXCR4 in vascularisation-dependent progression of breast cancer. It has been shown that VEGF up-regulates CXCR4 expression in a breast cancer cell line, and that these cells migrate towards CXCL12 in a VEGF-dependent manner²¹⁰. While these data point to an interplay between CXCL12/CXCR4 and VEGF as important for migration of breast tumour cells, no link to the angiogenic functions of these molecules was demonstrated in this system. In another study, fibroblasts isolated from invasive human breast carcinomas were implanted along with breast carcinoma cells into mice, and these fibroblasts enhanced tumour growth more effectively than fibroblasts isolated from normal mammary tissue¹⁹⁶. This effect was in part dependent on CXCL12 produced by host stromal cells to contribute to tumour vascularisation.

Somewhat paradoxically, one other way in which CXCL12 may serve to promote tumour progression is by enhancing immune evasion, although very little has been published in this area, particularly with respect to breast cancer. A recent report showed that CXCR4^{hi} MDA-MB-231 cells expressed lower basal and IFN-γ-inducible levels of MHC class II molecules than did CXCR4^{lo} cells²¹¹. CXCL12 was able to down-regulate MHC class II mRNA expression in both cell types, an effect that was reduced in CXCR4 siRNA-expressing cells. While by no means conclusive, these data implicate CXCL12 in assisting tumour cells to 'hide' from T cells by down-regulating their expression of antigen-presenting molecules. A handful of studies have implicated CXCL12 in immune evasion in other forms of cancer. Antagonism of CXCR4 with the small molecule inhibitor T22 enhanced the susceptibility of melanoma cells to CTL killing²¹². Moreover, CXCL12

induced expression of CCL5 by melanoma cells, which in turn induced apoptosis of tumour-infiltrating lymphocytes²¹³. These data suggest that CXCL12 may not only promote immune evasion by tumour cells, but also actively induce immune suppression. Further support for this idea came from a study of ovarian cancer. Ovarian tumour cells were found to express high levels of CXCL12, which resulted in the recruitment of plasmacytoid dendritic cell (pDC) precursors to the tumour mass where they were protected from apoptosis in a CXCR4-dependent manner. These pDC precursors induced IL-10 production by T cells, which inhibited T cell activation and tumour destruction²¹⁴.

A small number of studies have investigated the potential anti-tumour properties of CXCL12 by inducing its overexpression in various models of cancer, including fibrosarcoma²¹⁵, ovarian carcinoma²¹⁵, leukaemia²¹⁶, myeloma²¹⁷, lung carcinoma²¹⁸, colon carcinoma²¹⁸ and melanoma^{216, 218, 219}. In the majority of cases, CXCL12 expression induced anti-tumour immune responses that were dependent on T cell activity²¹⁵⁻²¹⁸. In one study of melanoma, however, engineered expression of high levels of CXCL12 inhibited the anti-tumour immune response²¹⁹. This effect was attributed to the ability of high levels of CXCL12 to induce chemorepulsion of tumour-specific CTL away from the tumour mass, and it was proposed that such a phenomenon might represent a tumour immune evasion strategy²¹⁹. Accumulation of such high levels of CXCL12 in a physiological setting has not been demonstrated and seems unlikely. However, the data suggest that in an immunotherapeutic setting, care should be exercised in selecting the dose of CXCL12 to be used to generate an optimal anti-tumour response. To date, no studies have investigated the effect of CXCL12 overexpression in breast cancer cells.

1.5 CXCL16 and CXCR6

1.5.1 General properties of CXCL16 and CXCR6

CXCL16 is one of the most recently identified chemokines. Whilst it is a member of the CXC subfamily, CXCL16 shows highest amino acid sequence homology to the CC chemokine CCL4 ²²⁰, although it has significant structural homology to the transmembrane chemokine CX₃CL1. Like CX₃CL1, CXCL16 is comprised of a globular extracellular chemokine domain attached via a mucin-like stalk to a transmembrane domain and a short intracellular cytoplasmic tail²²¹. CXCL16 is expressed *in vivo* in transmembrane form, but

it can also be cleaved by proteases such as ADAM10, releasing the chemokine domain as a soluble factor^{222, 223}.

CXCL16 is expressed by DC²²¹, macrophages, B cells²²⁰, and bone marrow stromal cells²²⁴, and at lower levels on aortic smooth muscle cells^{222, 225}, EC^{222, 226} and T cells²²⁷. It binds to the receptor CXCR6, which can act as a co-receptor for certain strains of HIV^{228, 229}. Expression of CXCR6 so far seems to be limited to activated CD8⁺ and CD4⁺ T cells²³⁰, particularly Type I polarised T cells²³¹, natural killer²²⁹ and NKT cells²²¹, B cells at low levels²²⁹, plasma cells²²⁴, human skin-derived DC and both immature and mature blood monocyte-derived DC²³².

1.5.2 Physiological functions of CXCL16

Various biological roles for CXCL16 have been reported (Table 1.4). The expression pattern of CXCR6 on Type I polarised T cells and the induction of CXCL16 by the Type I cytokine IFN- $\gamma^{222, 233, 234}$, suggests that this chemokine/receptor pair has a significant role in cell-mediated immunity. Further support for this hypothesis comes from evidence that CXCL16 induces IFN- γ production by CD8⁺ T cells, at least *in vitro*²³⁵, suggesting that CXCL16 and IFN- γ form a positive feedback loop to perpetuate Type I immune responses. In addition, the expression pattern of CXCL16 by various APC and its cognate receptor on T cells raises the possibility that the CXCL16/CXCR6 interaction may contribute to the activation of T cells, though this has yet to be verified. Certainly, CXCR6⁺ T cells adhere firmly to cells expressing surface CXCL16, in a manner that does not require CXCR6 signalling, suggesting direct involvement of the chemokine and its receptor in T cell activation²³⁶. In addition, CXCL16 is capable of binding to bacteria, facilitating their uptake and destruction by phagocytes²³⁷.

Undoubtedly, CXCL16 and CXCR6 are required for the recruitment of various leukocytes as part of the immune response. Several studies have demonstrated that blockade of the CXCL16-CXCR6 interaction inhibits recruitment of lymphocytes to sites of inflammation²³⁸⁻²⁴⁰. CXCR6 is frequently associated with leukocytic infiltrates in inflammatory hepatitis²⁴¹⁻²⁴³, and indeed, CXCR6^{-/-} mice are less susceptible to the disease²⁴⁴. Elevated CXCL16 expression has been documented in the synovia of

rheumatoid arthritis patients, and CXCR6⁺ cells are frequently observed in inflamed joints²⁴⁵⁻²⁴⁷. Moreover, an anti-CXCL16 antibody reduces the clinical arthritis score of mice with collagen-induced arthritis, with associated reduction of inflammatory cell infiltration and bone destruction in the synovium²⁴⁶. Of course, not all CXCL16/CXCR6-mediated immune responses are deleterious. CXCL16 and CXCR6 are important in the development of protective immune responses to acute bacterial infections. This was demonstrated in a model of acute *Salmonella* infection where depletion of CXCL16 resulted in increased bacterial colonisation of the spleen and liver and reduced IFN- γ production by splenic lymphocytes²³⁸.

Interestingly, under homeostatic conditions, the majority of CXCR6⁺ cells in the liver are NKT cells, which appear to 'patrol' hepatic sinusoids. CXCR6 deficiency leads to reduced survival of these cells, though not their patrolling speed or pattern ²⁴⁴. The exact function of these cells is not clear, however CXCL16 expression has been associated with increased NKT cell infiltration of cardiac allografts and graft tolerance²⁴⁸. Thus, it seems that under certain conditions, CXCL16 recruits inflammatory cells, while under other conditions, CXCL16 recruits cells that have an immunoregulatory role.

Within the vasculature, CXCL16 has been found in both aortic smooth muscle cells (ASMC)^{222, 225} and EC^{222, 226}. Recent work revealed that CXCL16 could enhance the proliferation, migration and tube formation *in vitro* of HUVEC²⁴⁹, suggesting that this chemokine may have angiogenic properties. The ability of CXCL16 to induce vascularisation *in vivo* is yet to be demonstrated, however.

The expression of CXCL16 on macrophages has significance for the pathogenesis of atherosclerosis. Minami et al. cloned CXCL16 as a surface receptor for oxidised low density lipoproteins (oxLDL) on macrophages, and demonstrated that it is localised to lipid-laden macrophages present in atherosclerotic plaques²⁵⁰. Smooth muscle cells from human atherosclerotic lesions also have been found to express CXCL16, and cultured ASMC up-regulate CXCL16 expression and oxLDL uptake in response to IFN- γ^{233} . The authors of these studies concluded that CXCL16 has a pro-atherosclerotic effect, but subsequent findings have led others to dispute these conclusions. In one study, reduced plasma CXCL16 was found to correlate with coronary artery disease²⁵¹, suggesting a

possible atheroprotective effect for CXCL16. Indeed, CXCL16^{-/-} LDL receptor^{-/-} double knockout mice display accelerated atherosclerosis²⁵². Macrophages from CXCL16^{-/-} mice have a reduced capacity to bind and internalise oxLDL *in vivo*, indicating that CXCL16-scavenging of oxLDL is a relevant process in atherosclerosis, and may actually have a protective effect²⁵².

1.5.3 CXCL16 and CXCR6 in cancer

To date, very little is known about the role of this chemokine receptor/ligand pair in tumorigenesis. CXCL16 and CXCR6 expression have been reported in cell lines derived from Hodgkin and Reed-Sternberg cells from Hodgkin's disease (HD). It was suggested that CXCL16 production by HD tissues may lead to the recruitment of CXCR6-expressing plasma cells that are commonly found in the inflammatory cell infiltrates characteristic of HD²⁵³.

Two groups have looked at the CXCL16/CXCR6 axis in human colorectal cancer. One study demonstrated that CXCL16 protein expression in TAM is suppressed compared with that in macrophages in normal adjacent tissue²⁵⁴. Further work revealed a similar trend in CXCR6 expression, with suppression in tumour tissue compared to normal tissue observed in 41% of cases²⁵⁵. The second group found that CXCL16 expression by colorectal tumour cells correlated with increased infiltration of the tumours by CD4⁺ and CD8⁺ lymphocytes and an improved prognosis²⁵⁶. This highlights the potential role of CXCL16 and CXCR6 in the anti-tumour immune response. It is possible that reduced expression of CXCL16 by TAM and tumour cells could inhibit recruitment and activation of CXCR6-expressing T cells, thereby providing a mechanism by which the tumour can evade their destructive cytolytic activity.

Interestingly, CXCR6 may have an anti-tumour function independent of the immune response. Sasai and colleagues reported that 3 to 4 per cent of mice lacking one or both functional copies of CXCR6 develop medulloblastoma²⁵⁷. Tumours from these mice displayed elevated activity of the sonic hedgehog signalling pathway, and were inhibited by a small molecule inhibitor of Smoothened, a signalling mediator within this pathway. These data point to a potential anti-tumour function of CXCR6 through blockade of the

sonic hedgehog mitogenic pathway, although a direct effect of CXCR6 signalling on this pathway is yet to be demonstrated.

It should be noted that the anti-tumour effect of CXCL16 and CXCR6 may not be universal for all tumours. In contrast to the observations of TAM in human colorectal cancer, immunohistochemical analysis of mouse fibrosarcomas revealed increased CXCL16 expression by resting TAM compared to normal macrophages²⁵⁸, suggesting that in this context, TAM CXCL16 expression provides a survival advantage to the tumours. Ludwig et al. found that CXCL16 was highly expressed and continuously shed by human glioma cells, in contrast to normal brain tissue. Since CXCL16 induced proliferation of mouse glial cells and CXCR6 mRNA could be detected in glioma cells, it was hypothesised that CXCL16 could act in an autocrine fashion to enhance the growth and/or survival of these cells²⁵⁹. CXCL16 could also promote tumour growth indirectly, as activated T cells recruited to the tumour site may produce factors that promote the survival and growth of tumour cells. This was postulated to be the case in Kaposi's sarcoma²⁶⁰.

There is some indication that the CXCL16-CXCR6 axis could play a role in metastasis, at least in some forms of cancer. Functional CXCR6 was detected in several human nasopharyngeal carcinoma (NPC) cell lines. CXCR6 was also detected in biopsies of regional and metastatic NPC from human patients, though expression in the primary tumour was only observed in patients with liver metastasis²⁶¹. A subsequent study reported that CXCL16 was required for migration of mammary epithelial tumour cells induced by Akt1, and that Akt1 deficiency led to reduced lung metastasis of these cells²⁶². Furthermore, Seidl et al. detected *de novo* expression of CXCR6 protein in primary melanomas and melanoma metastases, though it was not expressed in melanoma cell lines²⁶³. This highlights the need for care in extrapolating results from cancer cell lines to primary tissues. It also raises the possibility that CXCL16 and CXCR6 may play a more significant role in tumorigenesis than studies of tumour cell lines have revealed thus far.

1.6 The Research Project

As described in the preceding sections, CXCL12 and CXCL16 and their receptors are implicated in tumour progression and metastasis, or in processes that may be co-opted by tumours to enhance their progression, such as cell survival, proliferation, migration and

angiogenesis. In addition, these chemokines can participate in the orchestration of immune responses. An intriguing dichotomy of function for CXCL12 and CXCL16 in malignancy thus emerges: do they act to promote tumour development, or to enhance anti-tumour immune responses? Either way, the literature indicates that these chemokines have the potential to affect breast cancer progression, and this led to the following hypotheses:

- that expression of wild-type CXCL12/CXCL16 by breast tumour cells will influence the progression of breast cancer *in vivo*.
- (2) that expression of the respective antagonists of these chemokines by breast tumour cells will exert an opposing influence to wild-type chemokine expression on the progression of breast cancer *in vivo*.

To test these hypotheses, the experimental aims were as follows:

- (1) To develop DNA constructs expressing wild-type chemokines and their respective antagonists.
- (2) To characterize the 4T1.2 mouse model of breast cancer in terms of chemokine receptor/ligand expression and function.
- (3) To generate and characterize chemokine- and antagonist-expressing cell lines.
- (4) To investigate the effects of chemokine and antagonist expression on tumour progression *in vivo*.

The results of the experimental investigation of the aforementioned hypotheses and the implications of these findings will be addressed in the subsequent chapters of this thesis.

Systematic Name	Human Ligand	Mouse Ligand	Receptor(s)
CXC family			
CXCL1	GROα/MGSAα	Gro/KC?	CXCR2>CXCR1
CXCL2	GROB/MGSAB	Gro/KC?	CXCR2
CXCL3	GROv/MGSAv	Gro/KC?	CXCR2
CXCL4	PF4	PF4	CXCR3b
CXCL5	ENA-78	LIX?	CXCR2
CXCL6	GCP-2	СКа-3	CXCR1, CXCR2
CXCL7	NAP-2	Unknown	CXCR2
CXCL8	IL-8	Unknown	CXCR1. CXCR2
CXCL9	MIG	MIG	CXCR3a
CXCL10	IP-10	IP-10/CRG-2	CXCR3a
CXCL11	I-TAC	I-TAC	CXCR3a
CXCL12	$SDF-1\alpha/\beta$	SDF-1 α/β	CXCR4. CXCR7
CXCL13	BLC/BCA-1	BLC/BCA-1	CXCR5
CXCL14	BRAK	BRAK	Unknown
(CXCL15)	Unknown	Lungkine	Unknown
CXCL16	CXCL16	CXCL16	CXCR6
C family	Chello	CACETO	enent
XCI 1	Lymphotostin/SCM1 or/ATAC	Izemphotostin	VCP1
XCL 2	CM10	Lymphotaethi	XCR1
ACL2	SCMIP	Ulikhown	ACKI
CX ₃ C family			
CX ₃ CL1	Fracktalkine	Neurotactin	CX ₃ CR1
CC family			
CCL1	I-309	TCA-3/P500	CCR8
CCL2	MCP-1/MCAF/TDCF	IE?	CCR2
CCL3	$MIP-1\alpha/LD78\alpha$	MIP-1a	CCR1 CCR5
CCL3L1		Unknown	CCR1 CCR5
CCL4	MID 18	MID 18	CCR5
CCLT	MIII-Ip	MIT-IP	CCR1 CCR3
CCL5	RANTES	RANTES	CCR5
(CCL6)	Unknown	C10/MRP-1	Unknown
CCI 7	MCD 2	MADC9	CCR1, CCR2,
CCL/	MCP-3	MARC ?	CCR3
CCL8	MCP-2	MCP-2?	CCR3
(CCL9/10)	Unknown	Unknown	Unknown
CCL11	Eotaxin	Eotaxin	CCR3
(CCL12)	Unknown	MCP-5	CCR2
CCL13	MCP-4	Unknown	CCR2, CCR3
CCL14	HCC-1	Unknown	CCR1
CCL15	HCC-2/Lkn-1/MIP-1δ	Unknown	CCR1, CCR3
CCL16	HCC-4/LEC/LCC-1	LCC-1	CCR1
CCL17	TARC	TARC	CCR4
CCL18	DC-CK1/PARC/AMAC-1	Unknown	Unknown
CCL19	MIP-3β/ELC/exodus-3	MIP-3β/ELC/exodus-3	CCR7
CCL20	MIP-3a/LARC/exodus-1	MIP-3a/LARC/exodus-1	CCR6
CCL21	6Ckine/SLC/exodus-2	6Ckine/SLC/exodus-2	CCR7
CCL22	MDC/STCP-1	ABCD-1	CCR4

 Table 1.1 Chemokine nomenclature^{a, b}.

Systematic Name	Human Ligand	Mouse Ligand	Receptor(s)	
CC family (continued)				
CCL23	MPIF-1/CKβ8/CKβ8-1	Unknown	CCR1, CCR12	
CCL24	Eotaxin-2/MPIF-2	Unknown	CCR3	
CCL25	TECK	TECK	CCR9	
CCL26	Eotaxin-3	Unknown	CCR3	
CCL27	CTACK/ILC	ALP/CTACK/ILC/ESkine	CCR10	
CCL28	MEC	Unknown	CCR3/CCR10	

a, Adapted from refs 264 & 75.

b, Abbreviations: AMAC, alternative macrophage activation-associated CC chemokine; ATAC, activationinduced chemokine-related molecule; BCA, B cell attracting chemokine; BLC, B lymphocyte chemoattractant; BRAK, breast and kidney expressed chemokine; CK, chemokine; CTACK, cutaneous T cellactivating chemokine; DC-CK, dendritic cell-derived CC chemokine; ELC, EBL-1-ligand chemokine; ENA, epithelial neutrophil activating; GCP, granulocyte chemotactic protein; GRO, growth-related oncogene; HCC, human CC chemokine; ILC, IL-11 receptor alpha-locus chemokine; IP, IFN-y inducible protein; I-TAC, IFN-inducible T cell alpha chemoattractant; KC, keratinocyte chemoattractant; LARC, liver and activation-regulated chemokine; LCC, liver-specific CC chemokine; LEC, liver-expressed chemokine; LIX, LPS-induced chemokine; Lkn, leukotactin; MCAF, monocyte chemotactic and activating factor; MCP, monocyte chemoattractant protein; MDC, macrophage-derived chemokine; MEC, mucosae-associated epithelial chemokine; MGSA, melanoma growth stimulatory activity; MIG, monokine induced by IFN- γ ; MIP, macrophage inflammatory protein; MPIF, myeloid progenitor inhibitory factor; MRP, MIP-related protein; NAP, neutrophil-activating peptide; PARC, pulmonary and activation-regulated chemokine; PF, platelet factor; RANTES, regulated on activation normal T cell expressed and secreted; SCM, single C motif; SDF, stromal-derived factor; SLC, secondary lymphoid tissue chemokine; STCP, stimulated T cell chemoattractant protein; TARC, thymus and activation-regulated chemokine; TCA, T cell activation gene; TDCF, tumour-derived chemotactic factor; TECK, thymus-expressed chemokine.

Table 1.2 The chemokine receptors and their biological functions^{*a*}.

NOTE: This table is included on page 33 of the print copy of the thesis held in the University of Adelaide Library.

a, Adapted from ref 70.

Cell Type/Tissue	Role of CXCL12	References
HSC	Promotes engraftment and retention in the bone marrow	137-141
B cells	Growth factor for early B cells Optimal localisation of B cells in secondary lymphoid organs	136 144
Thymocytes	Population of thymic anlage Thymocyte development Thymic egress	145 147 148
T cells	Co-stimulation of CD4 ⁺ T cell activation HIV co-receptor	81, 155, 265 114
Mature leukocytes	Chemotaxis	149-154
Endothelial cells	Induction of VEGF expression Increased response to VEGF Chemotaxis Vascularisation	115, 116, 162, 163, 266
Endothelial precursors	Mobilisation Vascularisation in response to ischaemia	267 164
PGC	Migration Colonisation of gonads	123, 124, 165, 166
Tissue progenitors	Migration during development	167
Retinal axons	Survival & migration	168, 169, 268
Nervous system	Migration & proliferation of neural cells Induction of pain through triggering of nociceptive neurons	122
Epithelial cells	Renewal of colonic epithelium? Ca ²⁺ flux (alveolar epithelial cells)	172 120
Trophoblasts	Cytotrophoblast survival	173
Osteoclasts	Recruitment & differentiation	174, 175

Table 1.3 The biological functions of CXCL12 and CXCR4^{*a*}.

a, Abbreviations: *HIV*, human immunodeficiency virus; *HSC*, haematopoietic stem cell; *PGC*, primordial germ cell; *VEGF*, vascular endothelial growth factor.

Table 1.4 The biological functions of CXCL16 and CXCR6^{*a*}.

Cell Type/Tissue	Role of CXCL16/CXCR6	References
Type 1 polarised T cells	Recruitment to sites of inflammation	238-240, 246
NKT cells	Immunosurveillance of liver? Graft tolerance	244 248
Macrophages	Atherosclerosis	234, 250, 252
Endothelial cells	Proliferation Chemotaxis Tube formation	249
ASMC	Atherosclerosis	225, 233, 269
PBL	HIV/SIV co-receptor	228, 229, 270

a, Abbreviations: *ASMC*, aortic smooth muscle cells; *HIV*, human immunodeficiency virus; *PBL*, peripheral blood leukocytes; *SIV*, simian immunodeficiency virus.

This page left blank intentionally.



Figure 1.2 The steps of breast cancer metastasis.

(A) Breast tumour cells proliferate to form a primary tumour by developing the ability to evade apoptosis and stimulate their own unlimited growth. (B) Tumour cells either directly or indirectly (via stromal cells) induce angiogenesis to supply oxygen and nutrients to support the growing tumour mass. (C) To invade through the tissue, malignant cells must acquire the ability to survive without anchorage-dependent signals. Also, they must induce the production of ECM degradative enzymes that allow them to move through the tissue. *i*, Tumour cells can invade through basement membranes directly into blood vessels. *ii & iii*, Alternatively, tumour cells can enter the draining lymphatics and migrate to the local DLN, where they can become established and form secondary tumours (lymphatic metastasis). (D) Tumour cells can arrive in the circulation (haematogenous metastasis) either through direct intravasation at the tumour site, or through shedding from secondary LN tumours into the efferent lymphatics. These cells are carried through the body to distant metastatic sites. (E) Malignant cells arrest in the vasculature of a distant organ, adhere to the vessel wall and extravasate into the tissue. This requires the expression of degradative enzymes that facilitate tumour cell passage through the endothelial basement membrane and into the underlying tissue. (F) The final stage of distant metastasis is tumour cell survival and proliferation within the new tissue. This may involve autocrine mechanisms and recruitment of growth-supporting accessory cells. DLN, draining lymph node; ECM, extracellular matrix; LN, lymph node. Adapted from ref 271.







productive anti-tumour immune responses. Although IL-10 is generally regarded as an immunosuppressive cytokine, there is some evidence to suggest that IL-10 localised to the tumour site can enhance anti-tumour immune responses (depicted by a dashed arrow). Green arrowsdenote Breast cancers need to evade the immune response if they areto become clinically evident and life -threatening tumours. Immunosuppressive cytokines such as TGF- β and other tumour-derived factors can promote the generation of tolerogenic DC and T_{reg}. Tolerogenic DC in turn positive effects, red barsdenote inhibitory effects. DC, dendritic cell; IL-10, interleukin-10; iNOS, inducible nitric oxide synthase; MSC, promote the differentiation of Treg, which can suppress active anti-tumour immune responses. Tumour-derived factors also stimulate the accumulation of MSC. These cells produce immunosuppressive cytokines, as well as iNOS and arginase, the products of which inhibit myeloid suppressor cell; T_{reg} , regulatory T cell; TDF, tumour-derived factors; TGF- β , transforming growth factor- β . NOTE: This figure is included on page 41 of the print copy of the thesis held in the University of Adelaide Library.

Figure 1.4 Chemokine signalling.

Binding of a chemokine to its receptor induces a conformational change that leads to dissociation of the receptor-associated heterotrimeric G proteins into α and β γ subunits. These G -protein subunits can then activate various downstream signalling molecules including phosphoinositol 3-kinase (PI3K), protein kinase C (PKC) and mitogen-activated protein kinases (MAPK), as well as phospholipase C (PLC), which leads to inositol phosphate production and an increase in intracellular calcium. The overall effect of this signalling is to modulate actindependent cellular processes and increase expression and clustering of adhesion molecules, ultimately leading to cell chemotaxis and/or degranulation. Chemokine binding also stimulates tyrosine phosphorylation of the receptor and activation of the JAK/STAT pathway. This, together with MAPK activation, leads to modulation of gene expression. Finally, ligand binding induces internalisation of chemokine receptors through either clathrinmediated endocytosis or, in some cases, a lipid raft/caveolae-dependent mechanism (not shown). Internalised receptors are then degraded or recycled to the cell surface.

Adapted from ref 2.



CHAPTER 2 Materials &

Methods

CHAPTER 2: MATERIALS & METHODS

2.1 Subcloning of chemokine constructs

2.1.1 Chemokine DNA constructs

The three CXCL12 constructs used in this study were variants of the murine CXCL12 protein. CXCL12 was identical to wild-type CXCL12 protein. CXCL12_(P2G), an antagonist CXCL12 of activity, was a mutant form of the CXCL12 protein in which the second proline residue of the mature peptide was mutated to a glycine residue¹³¹. CXCL12_(Ala) was a putative biologically inactive form of CXCL12 in which the first two conserved cysteines (Cys9 and Cys11) of the mature peptide were replaced with alanine residues, effectively destroying the tertiary structure of the molecule. The two CXCL16 constructs used in this study were variants of the murine CXCL16 protein. CXCL16 was identical to wild-type membrane-bound CXCL16. CXCL16₍₉₋₂₂₀₎, an antagonist of CXCL16 activity, lacked the coding sequence for the first eight amino acids of the mature CXCL16 protein (but retained the signal sequence) and was expressed in membrane-bound form. All five chemokine proteins were tagged at the carboxyl-terminus with six histidine residues.

2.1.2 Cloning vector

The five histidine-tagged chemokine constructs used in this study were expressed in the pEF-IRES-puro6 (pEF) vector (Figure 2.1). This vector, a variation of the pEFIRES-P vector²⁷², was a kind gift from Dr. Dan Peet in the School of Molecular and Biomedical Science at the University of Adelaide. In pEF-IRES-puro6, the constitutive human elongation factor (EF)-1 α promoter controls expression of the chemokine construct and a puromycin selection marker, which are transcribed as a bicistronic mRNA separated by an internal ribosome entry site (IRES).

2.1.3 Polymerase Chain Reaction

Polymerase chain reaction was used to amplify the $CXCL12_{(Ala)}$ coding sequence without the KDEL peptide motif and to amplify the CXCL16 coding sequence with the nucleotides encoding a C-terminal 6x histidine residue tag appended. The mCXCL12(His) and the mCXCL16FLH primer sets (Table 2.1) were used to amplify the CXCL12_(Ala) and

CXCL16 sequences, respectively. The 5' primers contained an EcoRI restriction site while the 3' mCXCL12(His) primer contained a SpeI restriction site and the 3' mCXCL16FLH-R primer contained an EcoRV restriction site. These sites were incorporated to facilitate subcloning of the DNA cassettes into the multiple cloning site of pEF. The polymerase chain reaction was carried out in a final volume of 25µl and contained 5µl 5' primer (5µM), 5µl 3' primer (5µM), 2.5µl 10x Mg²⁺-free EXT buffer and 1.25µl MgCl₂ (50mM) (Finnzymes, Espoo, Finland), 1µl dNTPs (10mM) (Amersham Biosciences, Buckinghamshire, England, UK), 0.25µl DyNAzyme EXT polymerase (Finnzymes) and 125ng of template DNA. The reaction was carried out on a Programmable Thermal Controller (MJ Research Inc., Waltham, MA, USA) under the following conditions: 94°C for 2 minutes, then 15 cycles of 94°C for 30 seconds, 60°C for 45 seconds and 72°C for 1 minute, followed by a final extension step of 72°C for 10 minutes.

Overlap extension PCR was used to generate the CXCL16(9-220) construct. The mCXCL16FLH-F and mCXCL16SP-R PCR primers (Table 2.1) were used to amplify the signal sequence of CXCL16, while the mCXCL16A-F and mCXCL16FLH-R primers (Table 2.1) were used to amplify the nucleotide sequence encoding amino acid residues 9 to 220. Sequence analysis software (SignalP v3.0; http://www.cbs.dtu.dk/services/ SignalP/^{273, 274}) was employed to verify that deletion of the eight N-terminal amino acids of the CXCL16 protein would not alter the cleavage site of the signal peptide. The mCXCL16SP-R primer was designed such that it included a region complementary to the 5' end of the antagonist portion of CXCL16. The reaction mix for the first round of amplification of the signal peptide and antagonist portions of CXCL16 was the same as described above, with the following cycling conditions: 94°C for 2 minutes, then 15 cycles of 94°C for 30 seconds, 57°C for 45 seconds and 72°C for 1 minute, followed by a final extension step of 72°C for 10 minutes. For the second round of amplification, reaction mixes were as described above, with 1.25µl each of the two first round reactions used as template and mCXCL16FLH-F and mCXCL16FLH-R as primers. Cycling conditions were as described for the first round of amplification.

2.1.4 Subcloning

The CXCL12_(Ala) PCR product, the CXCL12- and CXCL12_(P2G)-containing pCR2.1-TOPO constructs and the pEF cloning vector were subjected to restriction digest with EcoRI and SpeI restriction enzymes (New England Biolabs, Beverly, MA, USA) in a 1:10 dilution of Eco Buffer (New England Biolabs) for 2 hours at 37°C. The CXCL16 and CXCL16(9-220) PCR products and pEF were digested under similar conditions with EcoRI and EcoRV (New England Biolabs). Fragments were then separated by agarose gel electrophoresis through a 1.2% agarose gel and the desired bands excised. DNA was purified from the gel using the Ultra Clean DNA purification kit (MoBio Laboratories Inc, Solana Beach, CA, USA), according to the manufacturer's instructions. The purified chemokine constructs were ligated into the digested pEF vector in a final reaction volume of 10µl: 1µl 10x TL-1 ligation buffer and 1µl T4 DNA ligase (Geneworks, Adelaide, SA, Australia), 3µl purified chemokine DNA, 1µl pEF DNA and 4µl H₂O. Reactions were incubated overnight at 4°C. The ligation mixtures were then used to transform chemically competent TOP10 Escherichia coli cells (Invitrogen, Carlsbad, CA, USA). Cells were incubated with the ligation mixture on ice for up to 2 hours. Cells were then subjected to heat shock at 37°C for 3 minutes followed by 10 minutes on ice. Cells were then allowed to recover in Luria Broth at 37°C for 45 minutes before being plated on Luria agar plates supplemented with 25µg/ml ampicillin. Plates were incubated at 37°C overnight.

2.1.5 Identification of positive clones

Overnight cultures of potential positive clones were set up in 10ml Luria broth with $25\mu g/ml$ ampicillin and plasmid mini-preps were performed the following day using the alkaline lysis method. Briefly, 1.5ml of culture were spun down and supernatant removed. Cells were resuspended in 100µl buffer P1 (Section 2.6.2.1), lysed in 100µl buffer P2 (Section 2.6.2.2), and cell debris was precipitated from solution with 100µl buffer P3 (Section 2.6.2.3). Debris was spun down for 8 minutes at 4°C, and plasmid DNA-containing supernatant was transferred to a fresh eppendorf tube. DNA was washed with 100% ethanol and resuspended in 20µl milliQ H₂O. DNA concentration was determined on a BioRad SmartSpec 3000 spectrophotometer (BioRad Laboratories, Hercules, CA, USA). Positive clones were identified by subjecting plasmid DNA to restriction digest with EcoRI and SpeI (CXCL12 constructs) or EcoRI and EcoRV (CXCL16 constructs) and separating

the fragments generated by agarose gel electrophoresis. The presence of the CXCL12, $CXCL12_{(P2G)}$, $CXCL12_{(Ala)}$, CXCL16 or $CXCL16_{(9-220)}$ coding sequence was confirmed by sequencing plasmid DNA from positive clones. Sequencing reactions contained 6µl Big Dye Terminator Ready Reaction Mix (Version 3; Perkin-Elmer, Foster City, CA, USA), 2µl primer (pEF-F or pEF-R at 5µM; Table 2.1), and 250ng plasmid DNA made up to a final volume of 20µl with H₂O. Cycling conditions were as follows: 94°C for 2 min, followed by 25 cycles of 94°C for 30 seconds, 50°C for 30 seconds, 65°C for 10 minutes, then a final extension step of 65°C for 10 minutes. DNA was precipitated with 60% isopropanol, pelleted by centrifuging at 14000rpm for 25 minutes at -9°C, and then washed in 70% ethanol, before being vacuum dried and sent to the Institute of Medical and Veterinary Science (IMVS; Adelaide, SA, Australia) for analysis on an ABI 373A DNA sequencer (Perkin-Elmer).

2.2 Generation of transfected cell lines

2.2.1 Cells

The murine breast epithelial cancer cell line 4T1.2 was derived from a spontaneously arising mammary carcinoma in a Balb/cfC3H mouse, and was a kind donation from Dr. Robin Anderson (Peter MacCallum Cancer Institute, Melbourne, Vic, Australia). Cells were cultured in complete Alpha MEM (Section 2.6.1.2) and incubated at 37°C in 5% CO_2 . Cells were passaged every 2-3 days by rinsing flasks with sterile phosphate buffered saline (PBS; endotoxin-free) and then incubating flasks with 10mM EDTA in PBS (endotoxin-free) at 37°C in 5% CO_2 for 10 minutes to remove adherent cells. Cells were passaged for no more than 6 weeks.

The murine pre-B lymphocyte cell line 300.19 was cultured at 37° C in 5% CO₂ in complete RPMI (Section 2.6.1.6). Cells were passaged every 2-3 days, and cells were maintained in culture for no more than 6 weeks.

2.2.2 Transfection

4T1.2 cells were removed from tissue culture flasks with 10mM EDTA as described above and washed once in iced PBS. Aliquots of 370μ l containing 5 x 10⁶ cells were placed into

chilled electroporation cuvettes (BioRad) and incubated on ice with 20-40 μ g of plasmid DNA for 10 minutes. Cells were electroporated with a BioRad Gene Pulser apparatus at 300V and 960 μ F. Cells were then returned to ice for 10 minutes before being resuspended in 15ml of pre-warmed complete Alpha MEM and placed at 37°C in 5% CO₂. Two to three days later, selection with 6 μ g/ml puromycin (Calbiochem, La Jolla, CA, USA) was commenced. Individual clones of stably transfected cells were obtained by limiting dilution. Cells were assayed by ELISA for chemokine production as described in Section 2.4.3. For each of the 5 clones secreting the greatest amounts of chemokine protein, equal numbers of cells were taken, pooled and cultured for further experiments.

2.3 Mouse model

2.3.1 Animals

Female 6-8 week old Balb/c mice were bred and maintained at the University of Adelaide Animal Facility (Adelaide, SA, Australia). Severe combined immunodeficient (SCID; C.B-17-Igh-1^b-Prkdc^{scid}) mice and athymic (Nude) mice on the Balb/c background (BALB/c-Foxn1^{nu}) were bred at the Animal Resources Centre (ARC; Canning Vale, WA, Australia) and maintained at the University of Adelaide Animal Facility. Female 6-8 week old J α 18^{-/-} mice and 9-16 week old pfp^{-/-}, TRAIL^{-/-} and IFN- $\gamma^{-/-}$ mice (all on the Balb/c background) were obtained from the Peter MacCallum Cancer Centre (Melbourne, Vic, Australia) and maintained at the University of Adelaide Animal Facility.

2.3.2 Tumour model

Tumour cells were removed from flasks with PBS containing 10mM EDTA and resuspended in PBS at a final concentration of 1×10^7 cells/ml. Mice were anaesthetized by halothane inhalation and 1×10^5 cells in a 10µl volume were injected into the fourth mammary fat pad using a 50µl Hamilton syringe (Hamilton, Reno, NE, USA). Tumour growth was assessed every 2 days and tumour diameter was measured using digital callipers (Mitutoyo, Japan). Tumour volume was calculated using the equation width² x length/2 to approximate the volume of a spheroid. Mice were euthanised once tumour diameter reached 15mm or tumours became ulcerated. To quantify metastasis, lungs were fixed for 5 days in 4% formaldehyde in PBS. Lungs were passed through two 30% sucrose

baths over 2 days before being immersed in 70% ethanol. Lungs were dissected into five lobes and surface metastases were counted under a dissecting microscope. For correlation of tumour weight with numbers and proportions of total, CD4⁺ and CD8⁺ splenocytes, spleens were removed, minced and macerated through 70µm nylon sieves (BD, Franklin Lakes, NJ, USA) to generate single cell suspensions in complete RPMI. Splenocytes were centrifuged and treated with mouse red cell removal buffer (MRCRB; Section 2.6.2.8) for 5 minutes at 37°C before being washed and resuspended in PBS/BSA/Azide (PBA; Section 2.6.2.10). Splenocytes were then stained with fluorochrome-conjugated anti-mouse CD4 or anti-mouse CD8 antibodies for flow cytometric analysis (see Section 2.4.1 and Table 2.2).

For the experimental metastasis experiments, tumour cells were harvested as described and resuspended to 2.5×10^6 or 3.75×10^6 cells/ml in PBS. Cells in a 200µl volume were injected into the tail vein using a 0.5ml insulin syringe (BD). After 14 days, mice were sacrificed and the lungs were harvested for enumeration of surface metastases as described above.

For the cytotoxicity experiments, mice received 10^6 cells in a 25µl volume of PBS in each hind foot pad. Draining (popliteal) LN were removed 7 days later, and lymphocytes extracted for use as effector cells in *in vitro* cytotoxicity assays.

For analysis of early stage tumours and draining lymph node (DLN) by immunohistochemistry, flow cytometry, intracellular cytokine staining, ELISPOT or cytokine bead array, mice received 1×10^5 cells by mammary fat pad injection or 1×10^6 cells by foot pad injection as described above, as indicated. Tissues were harvested at the indicated time points for analysis.

2.4 Analytical and functional assays

2.4.1 Flow cytometric staining

2.4.1.1 Cell preparation:

Cell lines: Cells were washed once in PBS and resuspended at 4×10^6 cells/ml in PBS/BSA/azide.

LN and tumour cells: Female 6-8 week old Balb/c mice were injected in both hind foot pads with 1×10^6 cells per foot pad. Seven days later, tumours and tumour-draining popliteal LN were harvested. Tumours were minced in 5mg/ml collagenase solution (Section 2.6.2.5) and incubated 1 hour at 37°C. Tumours were then macerated through a 70µm nylon sieve to generate single cell suspensions and washed in incomplete RPMI. LN cells were macerated through a 70µm nylon sieve to generate single cell suspensions. All cells were resuspended at 4×10^6 cells/ml in PBA.

2.4.1.2 Staining procedure

Cells were incubated at room temperature with 20µg/ml mouse gamma globulin (Rockland, Gilbertsville, PA, USA) for at least 20 minutes. Cells were added in 50µl aliquots to a 96-well V-bottom tray (Costar, Corning, NY, USA). Primary antibodies and isotype-matched controls were prepared at the appropriate concentrations in PBA (Table 2.2) and added to the appropriate wells. Cells were incubated with primary antibody for 30 minutes at room temperature, before being washed in 200µl PBA. If required, cells were then incubated for 30 minutes at 4°C with 50µl streptavidin-conjugated phycoerythrin (Rockland) diluted 1:150 in PBA and washed in 200µl PBA. Cells were washed in 200µl of PBS containing 0.04% sodium azide (PA; Section 2.6.2.9), before being resuspended in 150µl PA containing 2.5µM hydroxystilbamidine (Invitrogen Molecular Probes, Eugene, OR, USA) to distinguish non-viable cells. Cells were analysed on the same day on an LSRII flow cytometer (BD Biosciences, San Jose, CA, USA).

2.4.2 Reverse-transcriptase PCR

Total cell RNA was extracted using TRIzol® reagent (Invitrogen) according to the manufacturer's instructions and treated with RNase-free DNase (1U/µl; Promega, Madison, WI, USA). A total of 5µg DNase-treated RNA was reverse transcribed using random hexamers (Promega) and the reverse-transcriptase Superscript III (200U/µl; Invitrogen), as directed by the manufacturer. The cDNA obtained was treated with DNase-free RNase A (10mg/ml; Sigma, St Louis, MO, USA) for 1 hour at 37°C, and 1.25µl was used as template in subsequent PCR reactions. The sequences of the primers (synthesised

at Geneworks) used for PCR amplification are shown in Table 2.1. The reaction was carried out in a final volume of 25μ l and contained 5μ l 5' primer (5μ M), 5μ l 3' primer (5μ M), 2.5μ l 10x Mg²⁺-free EXT buffer and 1.25μ l MgCl₂ (50mM), 1μ l dNTPs (10mM), 0.25\mul DyNAzyme EXT polymerase and 1.25μ l of template cDNA. The reaction was carried out on a Programmable Thermal Controller under the following conditions: 94°C for 2 minutes, then 35 cycles of 94°C for 30 seconds, 52° C for 45 seconds and 72° C for 1 minute, followed by a final extension step of 72° C for 10 minutes. Fragments were visualized by agarose gel electrophoresis on a 2% agarose gel.

2.4.3 Sandwich ELISA for detection of chemokines

Capture and detection antibodies were purchased from R & D Systems (Minneapolis, MN, USA; Table 2.2). Anti-CXCL12 and anti-CXCL16 capture antibodies were used at 2µg/ml and lµg/ml, respectively. Biotinylated anti-CXCL12 and biotinylated anti-CXCL16 detection antibodies were used at 200ng/ml and 25ng/ml, respectively. Cell culture supernatants were collected from 2.5×10^6 cells grown in 500µl complete Alpha MEM. Protease inhibitor cocktail (Sigma) was added to the supernatants at a 1:200 dilution to prevent protein degradation. The wells of a 96-well ELISA plate (Costar) were coated overnight at room temperature with 100µl/well of capture antibody prepared in NaHCO₃ (0.1M) coating buffer. Wells were rinsed 3 times with PBS containing 0.05% Tween20 (Sigma), then blocked with 300µl/well of PBS containing 1% BSA and 5% sucrose or PBS containing 3% BSA for 1-2 hours at room temperature. Wells were rinsed again 3 times with PBS/Tween. One hundred µl per well of standard dilutions of recombinant mouse chemokine/cytokine prepared in diluent (0.1% in PBS) and culture supernatants were added to the tray, which was then incubated, shaking, for 2 hours at room temperature. Wells were rinsed 3 times with PBS/Tween, then 100µl/well of detection antibody prepared in diluent was added. The tray was incubated, shaking, for 2 hours at room temperature, then wells were washed 3 times with PBS/Tween. Streptavidin-conjugated horseradish peroxidase (HRP; Table 2.3) was diluted 1:20000 in diluent and 100µl was added to each well. The tray was incubated for 30 minutes, shaking, at room temperature. OPD solution was prepared from a kit (Sigma) as per the manufacturer's instructions, and 200µl was added to each well. The tray was incubated in the dark at room temperature for 20 minutes while colour developed, then 50µl 3M HCl was added to each well to stop the

reaction. The tray was read at 490nm on a Biotrak II Plate Reader (GE Healthcare, NSW, Australia).

2.4.4 Calcium mobilisation

4T1.2 cells were grown to confluence in 10cm tissue culture dishes (approximately 2 x 10^7 cells) and harvested using a cell scraper. Approximately 2 x 10^7 300.19 cells were collected from tissue culture flasks. Cells were resuspended in 2ml pre-warmed calcium buffer (Section 2.6.2.4) and incubated at 37° C in 5% CO₂ with 1µg/ml FURA-2AM (Invitrogen Molecular Probes) for 30 minutes. Cells were washed once and resuspended in 4.3ml in warmed calcium buffer. Two ml of cell suspension was added to a 10mm x 10mm x 48mm cuvette (Sarstedt, Germany) with a magnetic stirrer and analysed in an SLM AMINCO Bowman Series 2 Luminescence Spectrophotometer. Cells were stimulated with control CCL2_{(9.76)(4Ala)} (CCL2_(Ala)) peptide, 300ng/ml synthetic human CXCL12 (both kindly provided by Professor Ian Clark-Lewis, Biomedical Research Centre, University of British Columbia, Vancouver, Canada) or 300ng/ml recombinant mouse CXCL16 (R & D Systems) at the indicated times over a monitored period of 100 seconds. Maximum and minimum calcium mobilisation values were obtained by lysing the cells with 10% Triton X-100 (Sigma), then sequestering Ca²⁺ ions with 100mM EGTA and 2M NaOH, respectively.

2.4.5 In vitro proliferation assay

Cells were washed twice in serum-free Alpha MEM (Section 2.6.1.1) and resuspended at 1×10^5 cells/ml. Alpha MEM was prepared containing either 0.1%, 1% or 10% FCS. One millilitre of the appropriate medium and 25µl of cell suspension, or 1ml medium alone to serve as a blank, were added to each well of a 12-well tray. Cells were allowed to proliferate at 37°C in 5% CO₂ for up to 7 days. At days 0, 2, 5 and 7, sodium 3'-[1-(phenylaminocarbonyl)-3,4-tetrazolium]-bis (4-methoxy-6-nitro) benzene sulfonic acid hydrate (XTT; 1mg/ml; Sigma) was combined with N-methyl dibenzopyrazine methyl sulfate (PMS; 1.25mM; Sigma) in a 50:1 ratio. Two hundred microlitres were added per well and colour was allowed to develop at 37°C in 5% CO₂ for 2 hours. One hundred microlitres of culture supernatant from each well was transferred to a 96-well tray, which was then read at 490 nm (ref 650nm) on a Biotrak II Plate Reader.

2.4.6 Soft agar assay

Complete 2x Iscove's Modified Dulbecco's Medium (IMDM; Section 2.6.1.3) was prepared and warmed in a 40°C water bath. A 1.4% Bacto Agar (BD) solution was prepared in milliQ H₂O and boiled until dissolved. Complete 2x IMDM and 1.4% Bacto Agar were combined in a 1:1 ratio, then 1ml aliquots were used to coat the wells of a 6well tissue culture plate (BD) and allowed to set. A 0.6% Bacto Agar solution was prepared and combined with complete 2x IMDM as described above, and then placed in a 40°C water bath to equilibrate. Cells were prepared at 1 x 10⁴ cells/ml in complete Alpha MEM, and 100µl aliquots were mixed with 1ml of the 0.3% Bacto Agar in IMDM. The agar-cell mix was overlaid onto the 0.7% agar in IMDM in the 6-well trays and allowed to solidify at room temperature for half an hour. Trays were incubated at 37°C in 5% CO₂ for 10 days and colonies formed were counted under a dissecting microscope.

2.4.7 Haemoglobin assay

Tumours were harvested and snap frozen in liquid nitrogen. Tumours were then homogenized to a fine powder using a liquid nitrogen cooled stainless steel homogenizer (custom-made by the Peter MacCallum Cancer Centre). The powdered tumour was weighed and resuspended in 500µl deionised water. The homogenate was sonicated for approximately 30 seconds and debris was removed by centrifugation at 14000rpm for 30 minutes at 4°C. The resulting supernatant was then diluted 1:3 and used for the haemoglobin assay as follows. Drabkin's reagent was prepared containing Brij 35 Solution (both from Sigma) as per the manufacturer's instructions. Fifty microlitres per well of tumour supernatants were transferred to a 96-well tray and 200µl/well of Drabkin's solution was added. The plate was incubated for 15 minutes at room temperature to allow colour to develop and then read at 562nm on a Biotrak II Plate Reader. A standard curve was generated using haemoglobin from bovine blood (Sigma) dissolved in deionised water.

2.4.8 Immunohistochemical staining

Tissues were embedded in Tissue-Tek O.C.T. embedding medium (Miles, Eckhart, IN, USA), snap-frozen in liquid nitrogen and stored at -80°C. Six-micron cryostat sections on

gelatin-coated slides were fixed in 60%acetone/ 40%methanol for 10 minutes. Slides were rehydrated in PBS, and endogenous peroxidase activity was blocked with 0.3% hydrogen peroxide for 10 minutes. Slides were washed through 3 changes of PBS, then incubated with 100µg/ml mouse gamma globulin for 30 minutes. After 3 washes in PBS, slides were incubated with primary antibody for 2 hours in a humidified chamber at room temperature. After 3 washes in PBS, slides were incubated with HRP-conjugated secondary antibody (for unconjugated primary antibodies) or HRP-conjugated streptavidin (for biotinylated primary antibodies) for 1 hour. Slides were washed 3 times in PBS and binding of antibodies was revealed by incubation with 3,3'-diaminobenzidine (DAB) substrate (DAKO, NSW, Australia). Sections were counterstained with haematoxylin (Section 2.6.2.6), mounted and examined by routine light microscopy.

2.4.9 In vitro cytotoxicity assay

Cytotoxicity assay with effector cells from late stage tumours: Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 tumour cells. After approximately 5 weeks, spleens and tumour-draining (brachial) LN for each mouse were pooled and macerated through a 70µm nylon sieve to generate single cell suspensions of effector cells. Effector cells were resuspended in 7ml MRCRB and incubated for 3 minutes in a 37°C water bath to lyse red blood cells. Effector cells were then prepared in minimal complete RPMI (Section 2.6.1.5) at the appropriate concentrations to give the indicated effector:target ratios. Target 4T1.2 cells were prepared in the same medium containing 0.2µM calcein.AM (Invitrogen Molecular Probes) and incubated for 30 minutes at 37°C in 5% CO₂. Target cells were washed twice with PBS, then prepared in minimal complete RPMI and plated at 2.5×10^4 cells/well (100µl/well) in a 96-well flat-bottomed tray. Targets were allowed to adhere approximately 3 hours at 37°C in 5% CO₂. Effector cell suspensions were plated at 100µl/well and the plates were incubated at 37°C in 5% CO₂. Four days later, wells were washed 3 times with 200µl/well PBS to remove non-adherent cells, then 100µl/well PBS was added for reading of the plate. Fluorescence of the remaining adherent cells was measured on a Molecular Imager FX using Quantity One software version 4.6 (BioRad) and percentage cytotoxicity was calculated as 100-(fluorescence of target cells/fluorescence of target cells only x 100).

Cytotoxicity assay with effector cells from early stage tumours: Female 6-8 week old Balb/c mice were injected in both hind foot pads with 1×10^6 cells per foot pad. Seven days later, tumour-draining popliteal LN were obtained and macerated through a 70µm nylon sieve to generate single cell suspensions of effector cells. Effector cells were prepared in minimal complete RPMI at the appropriate concentrations to give the indicated effector:target ratios. Target 4T1.2 cells were prepared in the same medium and plated at 1×10^4 cells/well (100µl/well) in a 96-well flat-bottomed tray. Targets were allowed to adhere approximately 3 hours at 37°C in 5% CO₂. Effector cell suspensions were plated at 100µl/well and the plates were incubated at 37°C in 5% CO₂. After 3 days, non-adherent cells were removed and the wells were washed 3 times with 200µl/well of incomplete RPMI (Section 2.6.1.4). Remaining adherent target cells were labelled with 1µM calcein.AM in incomplete RPMI for 25 minutes at 37°C in 5% CO₂. Wells were washed 3 times with 200µl/well PBS, then 100µl/well PBS was added for reading of the plate. Fluorescence was measured on a Molecular Imager FX using Quantity One software version 4.6 and percentage cytotoxicity was calculated as 100-(fluorescence of target cells/fluorescence of target cells only x 100).

2.4.10 IL-17 ELISPOT assay

Female 6-8 week old Balb/c mice received 1×10^{6} tumour cells in 25µl PBS in each hind foot pad. Seven days later, tumour-draining popliteal LN were obtained and macerated through a 70µm nylon sieve to generate single cell suspensions. A mouse IL-17 ELISPOT kit (R & D Systems) was used to assess IL-17 production by the LN cells as per the manufacturer's instructions. Briefly, the wells of the supplied 96-well filter plate precoated with anti-mouse IL-17 antibody were pre-wet with 200µl/well complete RPMI for 20 minutes at room temperature. Cells were prepared in complete RPMI containing 20ng/ml phorbol 12-myristate 13-acetate (PMA; Sigma) and 1µM ionomycin (Invitrogen Molecular Probes) and aliquoted at 1.2 x 10⁵ cells/well (100µl/well) into the wells of the filter plate. Recombinant mouse IL-17 at 0.25ng/ml was used as a positive control and medium alone was used as a negative control. Cells were incubated overnight at 37°C. The following day, cells were flicked off and the plate washed four times with the wash buffer supplied. Bound IL-17 was detected by incubation of plate with 100µl/well biotinylated anti-mouse IL-17 overnight at 4°C. The plate was washed four times with wash buffer, then incubated for 2 hours at room temperature with 100μ l/well HRP-conjugated streptavidin. The plate was washed again and 100μ l/well 5-bromo-4-chloro-3' indolylphosphate p-toluidine salt/Nitro Blue tetrazolium (BCIP/NBT) chromogenic substrate was added to visualise spots. The plate was incubated 1h at room temperature in the dark before being washed and allowed to dry. Spots representing individual IL-17-producing cells were counted under a dissecting microscope.

2.4.11 Intracellular cytokine staining

Female 6-8 week old Balb/c mice received 1 x 10⁶ tumour cells in 25µl PBS in each hind foot pad. Seven days later, tumour-draining popliteal LN or pooled popliteal, inguinal, brachial and axillary LN from two naïve mice were obtained and macerated through a 70µm nylon sieve to generate single cell suspensions. Cells were prepared in complete RPMI and aliquoted at 200µl/well into the wells of a 96-well U-bottom plate. Cells were incubated overnight at 37°C and stimulated with 20ng/ml PMA and 1µM ionomycin in the presence of GolgiStopTM (BD Biosciences) at a 1:100 dilution. Cells were fixed, permeabilised (using the BD Biosciences Cytofix/Cytoperm kit) and stained with fluorochrome-conjugated antibodies to mouse CD4, CD8, IFN- γ , IL-4 and IL-17 (Table 2.2) and analysed on an LSRII flow cytometer.

2.4.12 Cytokine bead array

Female 6-8 week old Balb/c mice received 1 x 10^5 tumour cells in 10µl PBS in the fourth mammary fat pad on each flank. Six days later, tumour-draining inguinal LN were obtained and macerated through a 70µm nylon sieve to generate single cell suspensions. Cells were prepared at 5 x 10^6 cells/ml in complete RPMI containing 20ng/ml PMA and 1µM ionomycin and aliquoted at 500µl/well into the wells of a 24-well plate. The plate was incubated overnight at 37°C. The following day, culture supernatants were taken and filtered through a 0.2µm filter in preparation for cytokine bead array analysis of IL-1 α , IL-2, IL-4, IL-5, IL-6, IL-10, IL-17, IFN- γ , TNF- α , and GM-CSF production. The mouse Th1/Th2 10plex FlowCytomix kit (Bender MedSystems, Vienna, Austria) was used according to the manufacturer's instructions with minor modifications. Briefly, standards were prepared in the assay buffer supplied. The wells of the 96-well filter plate supplied

were pre-wet with 50µl/well of assay buffer. Standards and samples were added at 25µl/well to the filter plate. Bead mixture and biotinylated antibody mixture were prepared at a 1:10 dilution of the suggested concentration of each bead set and antibody, except for the IL-10 bead set and antibody, which were prepared at a 1:4 dilution. The bead mixture and the antibody mixture were added at 25µl/well and 50µl/well, respectively, to the plate, which was covered and incubated for 2 hours in the dark at room temperature on a microplate shaker. Wells were emptied using a vacuum filtration manifold (Millipore, Billerica, MA, USA) and washed twice with 100µl/well of assay buffer. Assay buffer was added at 100µl/well and streptavidin-PE conjugate, prepared at a 1:31.25 dilution as directed, was added at 50µl/well. The plate was covered and incubated for 1 hour in the dark at room temperature on a microplate shaker. Wells were washed as before, then 200ml/well assay buffer was added to the plate. The contents of each well were resuspended and transferred to FACS tubes (BD) for analysis on a BD FACSCanto flow cytometer (BD Biosciences). Data was analysed using the Bender MedSystems FlowCytomix Pro v2.1 software.

2.5 Statistical analysis

Statistical analysis was performed using GraphPad Instat version 3.05 and GraphPad Prism version 5 for Windows (GraphPad Software, San Diego California USA, www.graphpad.com).

2.6 Solutions and buffers

- 2.6.1 Media
- 2.6.1.1 Alpha Medium (serum-free)

Alpha MEM (Invitrogen Life Technologies, Grand Island, NY, USA) was supplemented with 1% penicillin/gentamycin stock solution (IMVS).

2.6.1.2 Alpha Medium (complete)

Alpha MEM was supplemented with 10% foetal calf serum (FCS; JRH Biosciences, Lenexa, KS, USA) and 1% penicillin/gentamycin stock solution.
2.6.1.3 2x Iscove's Modified Dulbecco's Medium (complete)

Powdered incomplete IMDM (Invitrogen Life Technologies) was made up to 2x concentration according to the manufacturer's instructions. Twenty ml of complete 2x IMDM was prepared by adding 4ml FCS, 400µl penicillin/gentamycin stock and 400µl Fungizone (Invitrogen Life Technologies) to 15ml incomplete 2x IMDM.

2.6.1.4 RPMI (incomplete)

RPMI 1640 medium (IMVS) was supplemented with 0.1% FCS, 1% penicillin/gentamycin stock solution and 0.1M HEPES (IMVS).

2.6.1.5 RPMI (minimal complete)

RPMI 1640 medium was supplemented with 10% FCS and 1% penicillin/gentamycin stock solution.

2.6.1.6 RPMI (complete)

RPMI 1640 medium was supplemented with 10% FCS, 1% penicillin/gentamycin stock solution, 100mM sodium pyruvate (Sigma), 10mM non-essential amino acids (Invitrogen Life Technologies) and 30mM β2-mercaptoethanol (Sigma).

2.6.2 Other solutions and buffers

2.6.2.1 Buffer P1

0.61g Tris base and 0.37g EDTA.H₂O were dissolved in 80ml milliQ H₂O and the pH was adjusted to 8.0 with HCl. The volume was made up to 100ml with H₂O and the solution autoclaved. DNase-free RNase A (Sigma) was added to a final concentration of 1μ g/ml.

2.6.2.2 Buffer P2

0.8g NaOH and 1g sodium dodecyl sulfate (SDS) were dissolved in 100ml milliQ H₂O and the solution autoclaved.

2.6.2.3 Buffer P3

29.5g potassium acetate was dissolved in 50ml milliQ H_2O and the pH was adjusted to 5.5 with glacial acetic acid. The volume was made up to 100ml with milliQ H_2O and the solution autoclaved.

2.6.2.4 Calcium Buffer

10ml 10x Hank's Balanced Salt Solution (Section 2.6.2.7), 1ml HEPES (1M) and 1ml $CaCl_2$ (0.16M) were added to 88ml milliQ H₂O. The pH was adjusted to 7.4 with 2M NaOH.

2.6.2.5 Collagenase Solution

Collagenase Type I-A (Sigma) was prepared at 5mg/ml in incomplete RPMI (Section 2.6.1.4) containing 1M CaCl₂ and 1U/ml DNAse (Promega).

2.6.2.6 Gill's Haematoxylin

250ml ethylene glycol, 2g Gurr Haematoxylin 75290, 0.2g sodium periodate (NaIO₄), 17.6g aluminium sulphate (Al₂(SO₄)₃.18H₂O), and 20ml glacial acetic acid were added in that order to 730ml distilled H₂O wile stirring. The solution was stirred 1 hour at room temperature until dissolved.

2.6.2.7 10x Hank's Balanced Salt Solution

80g NaCl, 4g KCl, 0.32g Na₂HPO₄, 0.6g KH₂PO₄ and 10g D-glucose were dissolved in 1L milliQ H₂O and the solution filter sterilized.

2.6.2.8 Mouse Red Cell Removal Buffer (MRCRB)

To make Solution A, 8.3g NH₄Cl was dissolved in 1L MQ H₂O. To make Solution B, 20.594g Tris base was dissolved in 1L MQ H₂O and the pH adjusted to 7.65 with HCl. Solution A and Solution B were combined in a 9:1 ratio to make 1L. The pH was adjusted to 7.2 with HCl and then the solution was filter sterilized and stored at 4° C.

2.6.2.9 PBS/azide (PA)

2ml 20% sodium azide (final concentration 0.04%) was added to 1L PBS.

2.6.2.10 PBS/BSA/azide (PBA)

10g bovine serum albumin (BSA; Sigma; final concentration 1%) was dissolved in 1L PBS and 2ml 20% sodium azide (final concentration 0.04%) was added.

Table 2.1Primers used	in this study.	
Name	Sequence (5'-3')	Purpose
mCXCL12(His)-F	GGAATTCGCCACCATGGACGCCAAGGTCGTCG	Cloning; RT-PCR
mCXCL12(His)-R	GACTAGTTCAGTGATGGTGGTGGTGGTGCATCTTGAGCCTC	Cloning; RT-PCR
mCXCL16FLH-F	CGAGAATTCATGAGGCGGGGCTTTGCACCC	Cloning
mCXCL16FLH-R	CGAGATATCCTAGTGATGGTGGTGGTGGGGGGTCTTGGTTC	Cloning
mCXCL16SP-R	CGATCACAAGAACAGCCATCGCCTGGCAGGG	Overlap extension PCR
mCXCL16A-F	TGTTCTTGTGATCGTACCATTTCTTCTGG	Overlap extension PCR
pEF-F	CCACTCCCAGTTCAATTACAGCTCTTAAGGCTAG	Sequencing
pEF-R	GGGAGAGGGGGGGAATTGGGC	Sequencing
mGAPDH-F	TCCTTGGAGGCCATGTAGGCCAT	RT-PCR
mGAPDH-R	TGATGACATCAAGAAGGTGGTGAAG	RT-PCR
mCXCL16RT-F	TCTTCTGGCACCCAGATACC	RT-PCR
mCXCL16RT-R	CGAGATATCTTAAGGCAAATGTTTTTGGTG	RT-PCR
mCXCR4-F	CCGGTACCTCGCCATTGTC	RT-PCR
mCXCR4-R	ACCATTATATGCTGGAATTGAAACAC	RT-PCR
mCXCR7-F	GTCAGCTGGAGAATGTGCTCTTTA	RT-PCR
mCXCR7-R	CTCTGTCTCTGACACTCTGGAGGC	RT-PCR
mCXCR6-F	GCTAGCATGGATGATGGGCATCAAGAGTCA	RT-PCR
mCXCR6-R	CAGACAAACACCAGGTCAGCCAGGGGGC	RT-PCR
All primers were synthe	sised at Geneworks (Adelaide, SA, Australia).	

AntigenCloneIsotypecontrol $A95-1$ rat IgG2bcontrol 54447 rat IgG2acontrol 54447 rat IgG2acontrol 54447 rat IgG2acontrol $R35-95$ rat IgG2acontrol $R4-22$ rat IgG1control $R1-22$ rat IgG1control $R1-22$ rat IgG1muCCL12 n/a goat polyclonalhu/muCXCL12 79018 mouse IgG1muCD4 $H129.19$ rat IgG2amuCD4 $H129.19$ rat IgG2amuCD3muCD4 $GK1.5$ rat IgG2amuCD4 $H123.3$ rat IgG2amuCD11c $HL3$ rat IgG2amuCD13 $MEC13.3$ rat IgG2amuCCL16 142417 rat IgG2amuCCL6 n/a goat polyclonalmuCCCL6 n/a rat IgG2amuCCCL6 n/a rat IgG2amuCCCCL6 n/a rat IgG2amuCCCCL6 n/a rat IgG2bmuCCCCL6 n/a rat IgG2bmuCCCR6 n/a <td< th=""><th>Conjugationabiotinbiotinnoneamster IgG1biotinabbit IgGnoneAPC-Cy7, FITC, PEFITCFITCPEonalbiotin</th><th>VendorBD BiosciencesBD Biosciences</th><th>Application^b FC IHC-Fr IHC-Fr FC FC FC FC FC FC ELISA ELISA ELISA IHC-Fr; FC</th></td<>	Conjugationabiotinbiotinnoneamster IgG1biotinabbit IgGnoneAPC-Cy7, FITC, PEFITCFITCPEonalbiotin	VendorBD BiosciencesBD Biosciences	Application ^b FC IHC-Fr IHC-Fr FC FC FC FC FC FC ELISA ELISA ELISA IHC-Fr; FC
controlA95-1rat $\lg G2b$ control54447rat $\lg G2a$ control54447rat $\lg G2a$ control $835-95$ rat $\lg G2a$ control $R4-22$ rat $\lg G1a$ control $R35-95$ rat $\lg G1a$ control $R4-22$ rat $\lg G1a$ control $R1-22$ rat $\lg G1a$ muCXCL12 n/a goat polyclonalhu/muCXCL12 n/a goat polyclonalhu/muCXCL12 n/a goat polyclonalmuCD4 $H129.19$ rat $\lg G2a$ muCD4 $H129.19$ rat $\lg G2a$ muCD11c $HL3$ rat $\lg G2a$ muCD11c $HL3$ rat $\lg G2a$ muCD11c $HL3$ rat $\lg G2a$ muCD11c $hU-2a$ rat $\lg G2a$ muCCL16 n/a goat polyclonalmuCXCL16 n/a rat $\lg G2a$ muCXCL16 n/a rat $\lg G2a$ muCXCR4 $2B1$	biotin none amster IgG1 biotin abbit IgG none FITC FITC PE pirtC pE biotin; PE-Cy7 FITC PE biotin; PE-Cy7 FITC	BD Biosciences R & D Systems BD Biosciences lab preparation BD Biosciences lab preparation BD Biosciences R & D Systems R & D Systems BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences	FC IHC-Fr IHC-Fr FC FC FC ELISA ELISA ELISA IHC-Fr; FC
control 5447 rat IgG2acontrolA19-3Armenian hamster Igcontrol n/a polyclonal rabbit IgGcontrol $R35-95$ rat IgG2acontrol $R3-22$ rat IgG1control $R4-22$ rat IgG1control $R4-22$ rat IgG1control $R4-22$ rat IgG1control $R4-22$ rat IgG1control $R1-22$ rat IgG1nuCCL12 n/a goat polyclonalhu/muCXCL12 79018 mouse IgG1hu/muCXCL12 79018 mouse IgG1nuCD4 $H129.19$ rat IgG2anuCD4 $GK1.5$ rat IgG2anuCD3 $muCD3$ rat IgG2anuCD11c $HL3$ Armenian hamster IgnuCD31 $muCXCL16$ n/a nuCCL16 n/a rat IgG2anuCCL16 n/a rat IgG2anuCCCL16 n/a rat IgG2anuCCCCL16 n/a rat IgG2anuCCCCL16 n/a rat IgG2anuCCCCL16 n/a rat IgG2anuCCCCL16 n/a rat IgG2bnuCCCCL16 n/a rat IgG2bnuCCCCCL16 n/a rat IgG2bnulF-17 $muH-12$ rat IgG1nulL-17 $TC11-18H10.1$ rat IgG1	amster IgG1 biotin abbit IgG none APC-Cy7, FITC, PE FITC FITC PE pe biotin biotin FITC PE biotin PE-Cy7 FITC	R & D Systems BD Biosciences lab preparation BD Biosciences BD Biosciences lab preparation BD Biosciences R & D Systems BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences	IHC-Fr IHC-Fr FC FC FC FC FC ELISA ELISA IHC-Fr, FC
control $A19-3$ Armenian hamster Igcontrol n/a polyclonal rabbit IgGcontrol $R35-95$ rat IgG2acontrol $R4-22$ rat IgG1control $R4-22$ rat IgG1control $R3-34$ rat IgG1control $R3-34$ rat IgG1control $R3-34$ rat IgG1hu/muCXCL12 n/a goat polyclonalhu/muCXCL12 79018 mouse IgG1hu/muCXCL12 79018 mouse IgG1hu/muCXCL13 n/a rat IgG2amuCD3muCD4 $GK1.5$ rat IgG2amuCD3muCC14 $H129.19$ rat IgG2amuCD3muCC16 n/a rat IgG2amuCC16 n/a rat IgG2bmuCXCL16 n/a rat IgG2bmuCXCL16 n/a rat IgG2bmuCXCL6 n/a rat IgG2bmuCXCR4 $2B11$ rat IgG2bmuL-4 $BVD4-1D11$ rat IgG1mulL-17TC11-18H10.1rat IgG1	amster IgG1 biotin abbit IgG none APC-Cy7, FITC, PE FITC FITC PE pe biotin biotin; PE-Cy7 FITC biotin; PE-Cy7	Cy7 BD Biosciences lab preparation BD Biosciences BD Biosciences lab preparation BD Biosciences R & D Systems BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences	IHC-Fr FC FC FC FC FC ELISA ELISA IHC-Fr; FC
controln/apolyclonal rabbit lgGcontrolR35-95rat lgG2acontrolR4-22rat lgG1controlGL113rat lgG1controlGL113rat lgG1controlGL113rat lgG1controlR3-34goat polyclonalhu/muCXCL12n/agoat polyclonalhu/muCXCL1279018mouse lgG1hu/muCXCL1279018mouse lgG1hu/muCXCL1279018mouse lgG1hu/muCXCL1279018mouse lgG1muCD4H129.19rat lgG2amuCD3muCD4GK1.5rat lgG2amuCD3muCD11cHL3rat lgG2amuCD3muCD11cHL3rat lgG2amuCD11chL3rat lgG2amuCCL16n/agoat polyclonalmuCXCL16n/agoat polyclonalmuCXCL16n/arat lgG2bmuCXCL16n/arabbit polyclonalmuCXCR42B11rat lgG2bmulFN-yXMG1.2rat lgG1mulFN-yWM1.101rat lgG1mulL-17TC11-18H10.1rat lgG1	abbit IgG none APC-Cy7, FITC, PE FITC FITC PE pe biotin FITC biotin; PE-Cy7 FITC biotin; ADC-Cy7 ET	Cy7lab preparationDSBD BiosciencesBD Bioscienceslab preparationBD BiosciencesR & D SystemsR & D SystemsBD BiosciencesBD BiosciencesBD BiosciencesBD BiosciencesBD Biosciences	FC FC FC FC ELISA ELISA IHC-Fr, FC
control $R35-95$ rat IgG2acontrol $R4-22$ rat IgG1control $GL113$ rat IgG1control $GL113$ rat IgG1huCXCL12 n/a goat polyclonalhu/muCXCL12 n/a goat polyclonalhu/muCXCL12 79018 mouse IgG1hu/muCXCL12 79018 mouse IgG1hu/muCXCL12 79018 mouse IgG1hu/muCXCL12 79018 mouse IgG1muCD4 $H129.19$ rat IgG2amuCD4 $GK1.5$ rat IgG2amuCD31 $MEC13.3$ rat IgG2amuCD31 $MEC13.3$ rat IgG2amuCD31 $MEC13.3$ rat IgG2amuCD11c $HL3$ rat IgG2amuCCL16 n/a goat polyclonalmuCXCL16 n/a goat polyclonalmuCXCL16 n/a rat IgG2bmuCXCL16 n/a rat IgG2bmuCXCR4 $2B11$ rat IgG2bmuLA4 $BVD4.1D11$ rat IgG1muIF-4 $BVD4.1D11$ rat IgG1muIL-4 $BVD4.1D11$ rat IgG1	APC-Cy7, FITC, PE FITC FITC PE piotin biotin FITC biotin; PE-Cy7 FITC	Cy7 BD Biosciences BD Biosciences lab preparation BD Biosciences R & D Systems BD Biosciences BD Biosciences BD Biosciences BD Biosciences BD Biosciences	FC FC FC FC ELISA ELISA IHC-FT, FC
controlR4-22rat IgMcontrolGL113rat IgG1controlGL113rat IgG1hu/muCXCL12n/agoat polyclonalhu/muCXCL1279018mouse IgG1hu/muCXCL1279018mouse IgG1muCD4H129.19rat IgG2amuCD4H129.19rat IgG2amuCD853-6.7rat IgG2amuCD853-6.7rat IgG2amuCD11cHL3Armenian hamster IgmuCD11cHL3rat IgG2amuCD11cHL3rat IgG2amuCD11cHL3rat IgG2amuCD11cHL3rat IgG2amuCD11cHL3rat IgG2amuCD11cHL3rat IgG2amuCCL16n./agoat polyclonalmuCXCL16n./agoat polyclonalmuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuCXCR42B11rat IgG2bmuL-17rat IgG1muIL-17rat IgG1muIL-17rat IgG1muIL-17rat IgG1	FITC FITC PE biotin biotin FITC FITC biotin: ADC.Cv7 ET	BD Biosciences lab preparation BD Biosciences R & D Systems R & D Systems BD Biosciences BD Biosciences BD Biosciences	FC FC FL ELISA ELISA IHC-Fr; FC
controlGL113rat IgG1control $R3-34$ rat IgG1huCXCL12n/agoat polyclonalhu/muCXCL12 79018 mouse IgG1muCD4H129.19rat IgG2amuCD4GK1.5rat IgG2amuCD8 $53-6.7$ rat IgG2amuCD31HL3Armenian hamster IgmuCD31MEC13.3rat IgG2amuCD31muCD31rat IgG2amuCD31muCD31rat IgG2amuCD31muCC116n/amuCCL16n/arat IgG2amuCXCL16n/arat IgG2amuCXCL16n/arat IgG2amuCXCL16n/arat IgG2amuCXCL16n/arat IgG2amuCXCL16n/arat IgG2amuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuLXCR42B11rat IgG2bmuIFN-yXMG1.2rat IgG1muIF-17TC11-18H10.1rat IgG1muIL-17TC11-18H10.1rat IgG1	FITC PE biotin biotin; PE-Cy7 FITC biotin: ADC.Cy7 ET	Iab preparationBD BiosciencesR & D SystemsR & D SystemsBD BiosciencesBD BiosciencesBD Biosciences	FC FC BLISA BLISA IHC-Fr, FC
controlR3-34rat IgG1huCXCL12n/agoat polyclonalhu/muCXCL1279018mouse IgG1muCD4H129.19rat IgG2amuCD4GK1.5rat IgG2amuCD853-6.7rat IgG2amuCD11cHL3Armenian harnster IgfmuCD31MEC13.3rat IgG2amuCD31MEC13.3rat IgG2amuCD11cHL3Armenian harnster IgfmuCD11cHL3Armenian harnster IgfmuCD11cn/arat IgG2amuCD11cn/agoat polyclonalmuCXCL16n/agoat polyclonalmuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuCXCL16n/arat IgG2bmuCXCR42B11rat IgG2bmuLABVD4+1D11rat IgG1mulL-4BVD4+1D11rat IgG1mulL-17TC11-18H10.1rat IgG1	PE biotin none biotin; PE-Cy7 FITC biotin: ADC.Cy7 ET	BD Biosciences R & D Systems R & D Systems BD Biosciences BD Biosciences C PF-Cv7 BD Biosciences	FC ELISA ELISA IHC-FT, FC
huCXCL12 n/a goat polyclonalhu/muCXCL1279018mouse IgG1muCD4H129.19rat IgG2amuCD4GK1.5rat IgG2amuCD853-6.7rat IgG2amuCD11cHL3Armenian hamster IgmuCD31MEC13.3rat IgG2amuCD31MEC13.3rat IgG2amuCD11c142417rat IgG2amuCD11b142417rat IgG2amuCCL16 n/a goat polyclonalmuCXCL16 n/a rabbit polyclonalmuCXCL16 n/a rabbit polyclonalmuCXCL16 n/a rabbit polyclonalmuCXCL16 n/a rabbit polyclonalmuCXCR42B11rat IgG2bmuLN- γ XMG1.2rat IgG1mulFN- γ BVD4-1D11rat IgG1mulL-17TC11-18H10.1rat IgG1	biotin none biotin; PE-Cy7 FITC biotin: ADC.Cv7 ET	R & D Systems R & D Systems BD Biosciences BD Biosciences C PF-Cv7 BD Biosciences	ELISA ELISA IHC-Fr, FC
hu/muCXCL12 79018 mouse IgG1 muCD4 H129.19 rat IgG2a muCD4 GK1.5 rat IgG2a muCD8 53-6.7 rat IgG2a muCD11c HL3 Armenian hamster Ig muCD31 MEC13.3 rat IgG2a muCD31 MEC13.3 rat IgG2a muCD11c HL3 Armenian hamster Ig muCD11c HL3 rat IgG2a muCD31 MEC13.3 rat IgG2a muCXCL16 n/a goat polyclonal muCXCL16 n/a rat IgG2b muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rat IgG2b muCXCL16 n/a rat IgG2b muCXCR4 2B11 rat IgG2b muIFN-y XMG1.2 rat IgG1 muIL-4 BVD4-1D11 rat IgG1 muIL-17 TC11-18H10.1 rat IgG1	none biotin; PE-Cy7 FITC biotin: ADC Cv7 ET	R & D Systems BD Biosciences BD Biosciences C PF-Cv7 BD Biosciences	ELISA IHC-Fr; FC FC
muCD4 H129.19 rat IgG2a muCD4 GK1.5 rat IgG2b muCD8 53-6.7 rat IgG2b muCD11c HL.3 Armenian hamster Ig muCD31 MEC13.3 rat IgG2a muCD11c HL.3 Armenian hamster Ig muCD11c HL3 rat IgG2a muCD11c HL3 rat IgG2a muCD11c 142417 rat IgG2a muCXCL16 n/a goat polyclonal muCXCL16 n/a rat IgG2b muCXCL16 n/a rat IgG2b muCXCR4 2B11 rat IgG2b muCXCR6 n/a rat IgG2b muLN-Y XMG1.2 rat IgG1 muIL-4 BVD4+1D11 rat IgG1 muIL-4 BVD4+1D11 rat IgG1	biotin; PE-Cy7 FITC biotin: ADC Cv7 ET	BD Biosciences BD Biosciences C PF-Cv7 BD Biosciences	IHC-Fr; FC FC
muCD4 GK1.5 rat IgG2b muCD8 53-6.7 rat IgG2a muCD11c HL3 Armenian hamster Ig muCD31 MEC13.3 rat IgG2a muCD31 MEC13.3 rat IgG2a muCD11c HL3 Armenian hamster Ig muCD31 MEC13.3 rat IgG2a muCXCL16 1.42417 rat IgG2a muCXCL16 n./a goat polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rat IgG2b muCXCR4 2B11 rat IgG1b muLN-y XMG1.2 rat IgG1 muIL-4 BVD4-1D11 rat IgG1 muIL-4 BVD4-1D11 rat IgG1	FITC kintin: ADC-CV7 ET	C PF-Cv7 BD Biosciences	
muCD8 53-6.7 rat IgG2a muCD11c HL3 Armenian hamster Igt muCD31 MEC13.3 rat IgG2a muCD11c HL3 Armenian hamster Igt muCD31 MEC13.3 rat IgG2a muCXL16 142417 rat IgG2a muCXCL16 n/a goat polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rat IgG2b muCXCR4 2B11 rat IgG2b muLSN-y XMG1.2 rat IgG1 mulFN-y XMG1.2 rat IgG1 mulL-4 BVD4-1D11 rat IgG1 mulL-17 TC11-18H10.1 rat IgG1	Piotin: ADC_CW7 EI	C PF-Cv7 BD Biosciences	L C
muCD11cHL3Armenian hamster IgmuCD31MEC13.3rat IgG2amuCXCL16142417rat IgG2amuCXCL16n./agoat polyclonalmuCXCL16n/arabbit polyclonalmuCXCL16n/arabbit polyclonalmuCXCL16n/arabbit polyclonalmuCXCR42B11rat IgG2bmuCXCR6n/arabbit polyclonalmuLXCR6n/arat IgG2bmulFN-yXMG1.2rat IgG1mulL-4BVD4-1D11rat IgG1mulL-17TC11-18H10.1rat IgG1	v_{10} v_{11} , v_{12} , v_{23} , v_{11}		IHC-Fr; FC
muCD31 MEC13.3 rat IgG2a muCXCL16 142417 rat IgG2a muCXCL16 n./a goat polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCR4 2B11 rat IgG2b muCXCR6 n/a rat IgG1b muIFN-y XMG1.2 rat IgG1 muIL-4 BVD4-1D11 rat IgG1b muIL-17 TC11-18H10.1 rat IgG1b	amster IgG1 biotin	BD Biosciences	IHC-Fr
muCXCL16 142417 rat IgG2a muCXCL16 n./a goat polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCL16 n/a rabbit polyclonal muCXCR4 2B11 rat IgG2b muCXCR6 n/a rabbit polyclonal muLFN-y XMG1.2 rat IgG1 mulL-4 BVD4-1D11 rat IgG2b mulL-17 TC11-18H10.1 rat IgG1	none	BD Biosciences	IHC-Fr
muCXCL16n./agoat polyclonalmuCXCL16n/arabbit polyclonalmuCXCR42B11rat IgG2bmuCXCR6n/arabbit polyclonalmulFN-yXMG1.2rat IgG1mulL-4BVD4-1D11rat IgG2bmulL-17TC11-18H10.1rat IgG1	none	R & D Systems	ELISA
muCXCL16n/arabbit polyclonalmuCXCR42B11rat IgG2bmuCXCR6n/arabbit polyclonalmuLFN-yXMG1.2rat IgG1mulL-4BVD4-1D11rat IgG2bmulL-17TC11-18H10.1rat IgG1	onal biotin	R & D Systems	ELISA
muCXCR42B11rat IgG2bmuCXCR6n/arabbit polyclonalmulFN-yXMG1.2rat IgG1mulL-4BVD4-1D11rat IgG2bmulL-17TC11-18H10.1rat IgG1	lonal none	lab preparation	FC
muCXCR6 n/a rabbit polyclonal mulFN-y XMG1.2 rat IgG1 mulL-4 BVD4-1D11 rat IgG2b mulL-17 TC11-18H10.1 rat IgG1	biotin	BD Biosciences	FC
mulFN-y XMG1.2 rat IgG1 mulL-4 BVD4-1D11 rat IgG2b mulL-17 TC11-18H10.1 rat IgG1	lonal none	lab preparation	FC
mulL-4 BVD4-1D11 rat IgG2b mulL-17 TC11-18H10.1 rat IgG1	FITC, PE	BD Biosciences	FC
mull17 TC11-18H10.1 rat lgG1	PE	BD Biosciences	FC
	PE	BD Biosciences	FC
muLYVE-1 223322 rat IgG2a	none	R & D Systems	IHC-Fr
rabbit IgG n/a goat polyclonal	onal Alexa Fluor 488	Invitrogen	FC
rabbit IgG n/a goat polyclonal	onal biotin	DAKO	FC
rat IgG n/a donkey polyclonal	/clonal HRP	Rockland	IHC-Fr

Reagent ^a	Vendor	Application ^b
hydroxystilbamidine nucleic acid stain	Invitrogen	FC
mouse gamma globulin	Rockland	FC, IHC-Fr
recombinant mouse CXCL12	$Peprotech^{c}$	ELISA
mouse CXCL16	donation (Prof. Ian Clark-Lewis, Vancouver, Canada)	ELISA
streptavidin-conjugated HRP	Rockland	ELISA, IHC-Fr
streptavidin-conjugated PE	Rockland	FC
 a. PE, phycoerythrin; HRP, horseradish peroxidase. b. ELISA, enzyme-linked immunosorbent assay; FC, c. Peprotech, Rocky Hill, NJ, USA. 	flow cytometry; IHC-Fr, immunohistochemistry on frozer	1 sections.

CHAPTER 2: Materials & Methods

Table 2.3Other reagents used in this study.



Figure 2.1 *The pEF-IRES-puro6 (pEF) mammalian expression vector.*

The pEF-IRES-puro6 (pEF) plasmid contains the puromycin-resistance gene (puro marker) downstream of an internal ribosome entry site (IRES) and a multiple cloning site (MCS). Expression of cloned DNA inserts is driven by the strong, constitutive human Elongation Factor-1 α promoter (pEF-1a). In addition, the vector carries the β -lactamase gene (amp marker) for selection during manipulations in bacteria. LEGEND: *black*, origin of replication; *green*, promoter; *khaki*, regulatory sequence; *orange*, selectable marker; *blue*, unique restriction site.



CHAPTER 3 Generation of Chemokine Constructs & Cell Lines



<u>CHAPTER 3: GENERATION OF CHEMOKINE</u> <u>CONSTRUCTS & CELL LINES</u>

3.1 Overview

The initial aims of this project were to clone the coding sequences for CXCL12 and CXCL16 and their antagonists into a mammalian expression vector and to transfect the resulting DNA constructs into 4T1.2 cells. The 4T1.2 model was selected for investigation of the influence of these chemokines on breast cancer progression as it recapitulates many of the aspects of the human disease and represents perhaps the most complete orthotopic model of breast cancer available for study. It was thus necessary to characterise 4T1.2 cells in terms of their endogenous CXCL12 and CXCL16 expression, as well as to determine the expression levels of the receptors for these chemokines. Following transfection of 4T1.2 cells with the DNA constructs, chemokine and antagonist expression, as well as *in vitro* growth characteristics of the transfected cell lines, were assessed.

3.2 Generation of chemokine constructs

3.2.1 Chemokine constructs used in this study

To study the role of CXCL12 in the progression of breast cancer, a DNA construct encoding wild-type murine CXCL12 was generated, as well as a construct encoding an antagonist of CXCL12 function, CXCL12_(P2G) (Table 3.1). This CXCL12 antagonist is identical in peptide sequence to wild-type CXCL12 except that the second proline residue of the mature peptide is mutated to a glycine residue. CXCL12_(P2G) has only 3-fold lower binding affinity for CXCR4 than wild-type CXCL12, however the amino acid substitution results in a complete loss of activity, as shown by the inability of the analogue to induce calcium mobilization in CEM cells (a human T cell line). Moreover, CXCL12_(P2G) is a potent antagonist of CXCL12 activity capable of inhibiting chemotaxis of CEM cells towards CXCL12¹³¹. Sequence alignments of the CXCL12 constructs are shown in Figure 3.1A.

For investigation of the role of CXCL16 in breast cancer, a construct encoding murine wild-type membrane-bound CXCL16 and a construct encoding CXCL16₍₉₋₂₂₀₎, an N-terminally truncated variant of CXCL16, were produced (Table 3.1 and Figure 3.1B).

The CXCL16₍₉₋₂₂₀₎ construct lacked the coding sequence for the first eight amino acids of the mature CXCL16 protein (but retained the signal sequence) to generate a protein that would antagonise wild-type CXCL16 function by binding to the CXCR6 receptor without inducing signalling. Previous work in our laboratory has shown the CXCL16₍₉₋₂₂₀₎ protein to inhibit the chemotaxis of CXCR6-expressing 300.19 cells towards CXCL16 (Scott Townley, unpublished results).

The CXCL12_(Ala) construct was designed as a control for non-specific effects of chemokine overexpression that were independent of chemokine activity. CXCL12_(Ala) was a variation of wild-type murine CXCL12 in which the first two conserved cysteines (Cys9 and Cys11) of the mature peptide were replaced with alanine residues (Table 3.1 and Figure 3.1A). The putative effect of these amino acid substitutions was destruction of the tertiary structure of CXCL12, and hence its biological function, as has been reported for other chemokines in which the conserved cysteine residues are mutated²⁷⁵. Since CXCL12_(Ala) lacked any specific biological activity, it was deemed suitable as a control for overexpression of both the CXCL12 and CXCL16 constructs. All five chemokine constructs used in this study were designed to encode six histidine residues at the carboxyl-terminus of the protein so that the engineered constructs could be distinguished from endogenous protein *in vivo*.

3.2.2 Cloning of chemokine constructs

The coding sequences for the CXCL12, CXCL12_(P2G) and CXCL12_(Ala) constructs, including the C-terminal histidine tag, had been cloned previously into the cloning vector pCR2.1-TOPO in this laboratory and were engineered such that they were flanked by a 5' EcoRI restriction site and a 3' SpeI site. To subclone the His-tagged CXCL12 and CXCL12_(P2G) constructs, the CXCL12 cDNA cassettes in pCR2.1-TOPO were excised using EcoRI and SpeI restriction enzymes. The CXCL12 cDNA cassettes were separated from the pCR2.1-TOPO vector backbone by agarose gel electrophoresis and then purified. They were then ligated into the pEF-IRES-puro6 (pEF) vector backbone, which had been similarly digested and purified (Figure 3.2A).

Direct subcloning of the $CXCL12_{(Ala)}$ construct from pCR2.1-TOPO to pEF was not possible as a KDEL motif for protein retention in the endoplasmic reticulum had been

previously engineered, C-terminal to the His tag. This had to be removed for the present study. Primers complementary to the $CXCL12_{(Ala)}$ sequence, including the histidine tag, were used to amplify the $CXCL12_{(Ala)}$ cassette without the nucleotides encoding the KDEL motif. These primers also incorporated EcoRI and SpeI restriction sites which were used to clone the PCR product into pEF (Figure 3.2B). All three CXCL12 constructs were sequenced to confirm the absence of mutations and, in the case of $CXCL12_{(Ala)}$, the successful removal of the twelve nucleotides encoding the KDEL motif.

The coding sequence of full-length wild-type mouse CXCL16 had been cloned previously into the pEF vector in this laboratory. However, to distinguish engineered expression of full-length CXCL16 from endogenous expression, PCR primers were designed to add a nucleotide sequence encoding a six histidine tag at the 3' end of the chemokine coding sequence. This tag could not be used to distinguish soluble CXCL16 derived from engineered and endogenous sources, as the tag is retained intracellularly when the chemokine domain of CXCL16 is cleaved. Since the N-terminus of chemokines is a crucial determinant of receptor binding, the histidine tag could not be engineered 5' of the CXCL16 coding sequence. Should it become necessary to distinguish engineered versus endogenous expression of secreted CXCL16, a construct encoding only the chemokine domain (amino acid residues 1 to 88) with a C-terminal histidine tag could be used. In addition to the histidine tag, the CXCL16 primers were designed to incorporate EcoRI and EcoRV restriction sites to facilitate cloning of the PCR product into the pEF vector. CXCL16-His was amplified using these primers from the CXCL16::pEF template generated previously in the laboratory. The resulting PCR product was digested with EcoRI and EcoRV and ligated into pEF that had been similarly digested (Figure 3.2C). The new construct was then sequenced to confirm that no mutations had been introduced into the coding sequence and that the histidine tag had been successfully appended during the cloning process.

To generate the CXCL16₍₉₋₂₂₀₎ construct, an overlap extension PCR approach was taken (Figure 3.3). To this end, PCR primers were designed and used to amplify the signal sequence of CXCL16 and the nucleotide sequence encoding amino acid residues 9 to 220. Sequence analysis software (SignalP v3.0; http://www.cbs.dtu.dk/services/SignalP/^{273, 274}) was employed to verify that deletion of the eight N-terminal amino acids of the CXCL16 protein would not alter the cleavage site of the signal peptide. The primers for the 3' end of

the signal sequence and the 5' end of the antagonist portion were designed such that they included a complementary region that would facilitate annealing of the two PCR products when combined in one reaction. The combined PCR products were then subjected to a second round of PCR with the primers used to amplify wild-type CXCL16 above. This produced the CXCL16₍₉₋₂₂₀₎ construct complete with signal sequence and C-terminal histidine tag and flanked by EcoRI and EcoRV restriction sites. This construct was then digested and cloned into pEF as described for the wild-type CXCL16 construct above (Figure 3.2D). The construct was sequenced to confirm that the correct sequence had been cloned.

Detailed vector maps of the chemokine constructs generated for this study are shown in Figure 3.4.

3.3 The 4T1.2 model system

To study the role of CXCL12 and CXCL16 in the progression of breast cancer, the 4T1.2 mouse model of breast cancer was chosen²⁷⁶. The 4T1.2 cell line was derived by single cell cloning from the 4T1 cell line, which originated from a spontaneously arising mammary carcinoma in a Balb/c mouse²⁷⁷. The 4T1.2 cell line was selected for its propensity to metastasise to lung, bone and other sites, a pattern of metastatic spread that closely resembles that of human breast cancer²⁷⁸. It is a particularly good model of spontaneously metastatic mammary carcinoma because unlike some metastatic breast cancer models, in which metastases are forced by injecting tumour cells intravenously, metastases in this model form by spontaneous migration of cells from the primary tumour. Primary mammary tumours can be generated by an orthotopic injection of 4T1.2 cells into the fat pad of syngeneic Balb/c mice.

Preliminary experiments were conducted to characterise the endogenous expression of CXCL12, CXCR4, CXCR7, CXCL16 and CXCR6 in the 4T1.2 model. There was no detectable expression of CXCL12 messenger RNA in total RNA extracted from the cell line or from 4T1.2 tumours (Figure 3.5A). Similar assessment of CXCL16 expression by RT-PCR produced a faint but distinct band for both the cell line and tumour samples (Figure 3.6A). The results of the RT-PCR were confirmed at the protein level by ELISA analysis of supernatants from cultured 4T1.2 cells. CXCL12 production by the 4T1.2 cell

line was below the lower limit of detection of the assay (Figure 3.5B). A low level of endogenous CXCL16 shed by 4T1.2 cells into culture supernatants was detectable by ELISA, however expression of membrane-bound CXCL16 could not be detected by flow cytometry (Figure 3.6B and C). Thus it seems that the expression level of endogenous CXCL16 is too low for chemokine retained on the cell surface to be detectable.

The expression of the CXCL12 receptor CXCR4 in the 4T1.2 cell line was examined by RT-PCR and flow cytometry. CXCR4 mRNA was expressed in 4T1.2 tumours, however no CXCR4 message was detected in the 4T1.2 cell line. Despite this, flow cytometric analysis indicated that the receptor was present on the surface of cells. It may be that very low levels of mRNA expression are sufficient for translation of the CXCR4 protein detected on the cell surface, and that the RT-PCR approach used here was not sensitive enough to detect such low levels of CXCR4 transcripts. Indeed, CXCR4 expression in 4T1.2 cells has been detected by semi-quantitative real-time PCR (Robin Anderson, personal communication). No functional response was detected when cells were stimulated with CXCL12 in calcium mobilisation assays. Due to the adherent nature of 4T1.2 cells, some small spikes in fluorescence were observed at random intervals during the test period due to the propensity of 4T1.2 cells to form clumps when in suspension. Thus greater variation in baseline fluorescence was observed for these cells than for the control 300.19 cells, which grow naturally in suspension.

Another receptor for CXCL12, CXCR7, has been identified recently, a receptor that does not induce mobilisation of intracellular calcium stores upon ligation by CXCL12. Expression of this receptor by 4T1.2 cells was analysed by RT-PCR, as no antibodies to mouse CXCR7 suitable for flow cytometric analysis are currently commercially available. Low levels of CXCR7 mRNA were detected in both the 4T1.2 cell line and 4T1.2 tumour samples (Figure 3.7A). This may explain the absence of calcium mobilisation in response to CXCL12 stimulation: CXCL12 may bind preferentially to CXCR7 rather than CXCR4 on these cells.

To determine if 4T1.2 cells expressed CXCR6, which could form a potential autocrine loop with endogenously produced CXCL16, the mRNA content of the cells was examined by RT-PCR. A faint band corresponding to a portion of the CXCR6 sequence could be detected (Figure 3.8A). However, flow cytometric analysis with an anti-CXCR6 antibody

demonstrated the absence of surface CXCR6 protein (Figure 3.8B). This was confirmed by a calcium mobilisation assay to detect receptor function: no mobilisation of intracellular calcium stores was seen in response to stimulation with recombinant CXCL16 protein, in contrast to positive control 300.19 cells transfected with a construct expressing CXCR6 (BCX6 cells; Figure 3.8C). Endogenous CXCL16 secreted by 4T1.2 cells may be sufficient to bind and induce internalisation of CXCR6, hence the apparent absence of CXCR6 surface expression and function, despite the presence of CXCR6 mRNA.

In addition to the *in vitro* analyses described above, expression of all three chemokine receptors in 4T1.2 tumours was analysed by RT-PCR, and a specific band for each receptor was produced (Figure 3.7A and Figure 3.8A). Whether this represents expression of the receptors by the tumour cells, by tumour-infiltrating cells, or both could not be determined.

3.4 In vitro generation and characterisation of chemokine constructexpressing cell lines

3.4.1 Generation of chemokine construct-expressing cell lines

The chemokine constructs generated as described in Section 3.1 above were transfected into 4T1.2 cells by electroporation. Stably transfected populations were screened by ELISA for expression of the protein of interest (Figure 3.9). Clones were isolated by limiting dilution from those populations expressing the highest amount of the desired protein. These clones were then screened individually by ELISA for protein production (Figure 3.10 and Figure 3.11). Five clones from each transfected population were selected and pooled to produce cell lines expressing the highest levels of protein. The exception to this was the selection of five clones from the 4TX12 2 population expressing levels of CXCL12 similar to those of the 4TX12 1 clones. This was to facilitate comparisons between the two 4TX12 cell lines generated. Cell lines were derived from pooled clones because formation of stable transfectants involves insertion of the transfected construct into the genome of the cell, with insertion occurring at a random point. Thus pooling multiple clones also minimised the chance that any changes in the general characteristics of the cell line were due to the specific effects of the chemokine expressed, and not to the genomic insertion point of the expression construct in any particular clone.

Six cell lines were generated: 4TX12 1 and 4TX12 2 (expressing the wild-type CXCL12 construct), 4T12P2G (expressing the CXCL12_(P2G) construct), 4T16 (expressing the wild-type CXCL16 construct), $4T\Delta16$ (expressing the CXCL16₍₉₋₂₂₀₎ construct) and 4T12Ala (expressing the control CXCL12_(Ala) construct). All six were screened for expression of their respective constructs by RT-PCR and ELISA (Figure 3.12 and Figure 3.13). All the cell lines expressed their respective constructs significantly over and above the endogenous levels of chemokine, as determined by ELISA, with the exception of the 4T12Ala line. The disruption of the tertiary structure of the CXCL12_(Ala) protein is likely to render its detection by ELISA less efficient. It may also more readily target the protein for destruction, resulting in reduced secretion. However, the RT-PCR data clearly indicate that the CXCL12_(Ala) construct is robustly expressed at the mRNA level, similar to the other constructs.

In addition to the RT-PCR and ELISA analyses, the 4T16 and 4T Δ 16 lines were assessed for surface expression of their respective CXCL16 constructs by flow cytometry (Figure 3.13C). Small but clear increases in fluorescence intensity were observed for both cell lines. Together with the ELISA analysis, these data indicate that both the membranebound and secreted forms of CXCL16 are produced by these cells.

3.4.2 In vitro characterisation of chemokine construct-expressing cell lines

The *in vitro* growth characteristics of the newly generated cell lines were assessed in two ways: assessment of proliferation on plastic and assessment of colony formation in semisolid agar. The first assay compared the proliferation rates of the eight cell lines with the parental 4T1.2 line, while the second was used to compare the anchorage-independent growth (a measure of tumorigenicity) of the cells.

The proliferative capacity was assessed by measuring the reduction of XTT by metabolically active cells growing in medium containing 10%, 1% or 0.1% FCS. No significant differences in cell growth properties were observed between the parental and CXCL12 or CXCL16 construct-expressing cell lines (Figure 3.14 and Figure 3.15). Similarly, there were no significant differences in the number of colonies formed by the cells in soft agar (Figure 3.16). These data suggested that neither the process involved in

generating stable transfectants, nor the production of CXCL12 and CXCL16, had a direct effect on the growth of 4T1.2 cells, at least in an *in vitro* setting.

3.5 Summary of findings and preliminary discussion

In summary, no endogenous CXCL12 expression was detected in 4T1.2 cells. Although no CXCR4 mRNA was detected, surface CXCR4 was clearly present on 4T1.2 cells as determined by flow cytometry. A low level of CXCR7 mRNA was revealed by RT-PCR, but this could not be confirmed at the protein level due to the lack of reagents available for detecting CXCR7 protein. Neither the expression of CXCR4 nor CXCR7 correlated with a functional response to CXCL12 stimulation. It should be noted that only mobilisation of intracellular calcium stores was assessed as an indicator of receptor signalling, thus CXCL12-induced signalling through CXCR7 would not be detected using this method. In the case of CXCR4, it was hypothesised that the lack of calcium mobilisation in response to CXCL12 was due to competition of CXCR7 with CXCR4 for CXCL12 binding, as CXCR7 seems to have a higher affinity for CXCL12 than CXCR4¹²⁹. Since CXCR7 also binds the CXCR3 ligand CXCL11¹²⁹, one way to determine whether or not CXCL12 binds to CXCR7 on 4T1.2 cells is to conduct competitive binding studies. Displacement of CXCL12 by CXCL11 would indicate that CXCL12 binds CXCR7 rather than CXCR4 on 4T1.2 cells.

A low level of CXCL16 was detected in 4T1.2 cells using both RT-PCR and ELISA. A very low level of CXCR6 mRNA expression was observed, but this did not correlate with detectable surface protein expression. The lack of calcium mobilisation by 4T1.2 cells in response to CXCL16 was not surprising given the absence of surface-expressed CXCR6. It may be that the lack of surface CXCR6 was a consequence of the low level of mRNA expression, though clearly CXCR4 protein was detectable despite very low transcript levels. Alternatively, internalisation of CXCR6 may be induced by endogenous CXCL16, perhaps forming an autocrine signalling loop. Intracellular staining for CXCR6 could be employed to resolve this question.

Chemokine and chemokine antagonist production was clearly detectable in the transfected cell lines. In an attempt to confirm the functionality of the expressed proteins, culture supernatants from the transfectants were collected and used as stimuli in calcium mobilisation and chemotaxis assays with CXCR4- or CXCR6-expressing 300.19 cells. However, chemokine concentrations achieved in the supernatants were too low for responses to be observed *in vitro* (data not shown). High background due to the presence of serum in the culture medium could not be avoided due to the sensitivity of 4T1.2 cells to serum-starvation. Specially formulated serum-free medium may be required for generation of culture supernatants with minimal background effects. It may be also necessary to purify and concentrate the secreted chemokines using the histidine tag prior to testing. In the case of the CXCL16 constructs, where the histidine tag is lost when the chemokine is cleaved from the cell surface, culture supernatants from cell lines expressing the soluble chemokine domain with a C-terminal histidine tag could be used for functional assays.

Five clones from each transfected cell population were pooled to produce cell lines for further study. It may seem counterproductive to clone out a cell line only to pool clones together again. However, as discussed above, multiple clones were pooled to obtain high-level chemokine expression, as well as to ensure that any changes in the general characteristics of the cell line were due to the specific effects of the chemokine expressed. Ideally, a panel of individual clones would be selected for *in vivo* study, but this is a very labour intensive approach. Cultured cell lines were discarded and fresh aliquots defrosted on a regular basis to prevent a preponderance of any one particular clone developing and altering the properties of the cell line as a whole.

No differences in the *in vitro* growth characteristics of the transfected cell lines and the 4T1.2 cell line were observed. This suggests that chemokine or antagonist expression does not directly affect tumour cell growth properties. Additional tests *in vitro*, for instance migration and invasion assays, could be employed to further characterise the effect of chemokine expression on tumour cells. However, a more complete picture of the effect of chemokine expression on tumour progression could be gained in an *in vivo* system where the influence of host components can be studied as well. The experiments described in the following chapter examine the effects of chemokine expression on tumour growth *in vivo*.

This page left blank intentionally.

Table 3.1 Chemokine constructs and their functions.

Construct	Mutation	Function
CXCL12	None	Wild-type chemokine
CXCL12 _(P2G)	$Pro2 \rightarrow Gly$	CXCL12 antagonist
CXCL16	None	Wild-type chemokine
CXCL16 ₍₉₋₂₂₀₎	Deletion of residues 1-8	CXCL16 antagonist
CXCL12(Ala)	Cys9 → Ala; Cys11→ Ala	Biologically inactive control

Figure 3.1 Amino acid sequences of chemokine constructs.

The amino acid sequences of the chemokine constructs used in this study are shown in alignment. The sequences shown are those of the precursor proteins. **(A)** CXCL12 constructs. The first 21 amino acids *(blue and underlined)* form the signal peptide that targets the protein for secretion; these are cleaved from the mature protein. Each construct carries a six histidine residue tag *(green italics)* at its C-terminus. The glycine substitution in the CXCL12_(P2G) antagonist is highlighted in red, as are the alanine substitutions in the biologically inactive CXCL12_(Ala) control construct. **(B)** CXCL16 constructs. The first 26 amino acids *(blue and underlined)* form the signal peptide that targets the protein to the cell membrane; these are cleaved from the mature protein. Each construct carries a six histidine residue tag *(green italics)* at its C-terminus. The eight amino acids deleted in the CXCL16₍₉₋₂₂₀₎ antagonist are represented by red dots.

CXCL12	1 MD	JAK VVAVLALVLAALCISDGKPVSLSYRCPCBFFESHIARANVKHLKILNTPNCALOIVARLKNNNROVCIDPKLK
CXCL12 (P2G)	1	DAKVVAVLALVLAALCISDGK <mark>G</mark> VSLSYRCPCRFFESHIARANVKHLKILNTPNCALQIVARLKNNNRQVCIDFKLK
CXCL12 (Ala)	1 MD	DAKVVAVLALVLAALCISDGKPVSLSYRAPARFESHIARANVKHLKILNTPNCALQIVARLKNNNRQVCIDPKLK
CXCL12	78 WI	-Qeylekalnkrlkm <i>hhhhh</i>
CXCL12 (P2G)	78 WI	:Qeylekalnkrlkm <i>hhhhh</i>
CXCL12 (Ala)	78 WI	Qeylekalnkrlkm <i>hhhhh</i>
ď		
CXCL16	1 MR	<pre>KGFGPLSLAFFLFLLALLTLPGDGNQGSVAGSCSCDRTISSGTQIPQGTLDHIRKYLKAFHRCPFFIRFQLQSKS</pre>
CXCL16(9-220)	1 <u>MR</u>	<pre>krgfgplslafflflalltlpgdg</pre> cscdrtissgtgipggtldhirkylkafhrcpffirfglgsks
CXCL16	78VCG	5GSQDQWVRELVDCFER KECGTGHGKSFHHQKHLPQASTQTPEAAEGTPSDTSTPAHSQSTQHSTLPSGALSLNK
CXCL16(9-220)	78 VC	CGGSQDQWVRELVDCFERKECGTGHGKSFHHQKHLPQASTQTPEAAEGTPSDTSTPAHSQSTQHSTLPSGALSLNK
CXCL16	155 EH	ITQPWEMTTLPSGYGLEARPEAEANEKQQDDRQQEAPGAGASTPAWVPVLSLLAIVFFLTAAMAYVLCNRRATQQN
CXCL16(9-220)	155 EH	ITQPWEMTTLPSGYGLEARPEAEANEKQQDDRQQEAPGAGASTPAWVPVLSLLAIVFFLTAAMAYVLCNRRATQQN
CXCL16	232SAG	ЗІДІМҮТРVЕРКР ННННН

۷

CXCL16 (9-220)

232SAGLQLWYTPVEPRP ННННН

Figure 3.2 Cloning of chemokine constructs.

(A) The CXCL12 and CXCL12_(P2G) coding sequences were excised from the pCR2.1-TOPO vector with the restriction enzymes EcoRI and SpeI. The resulting fragments were separated by agarose gel electrophoresis and the desired bands were excised and purified. The purified chemokine DNA was then ligated into pEF-IRES-puro6 which had been similarly digested and purified. (B) The CXCL12_(Ala) coding sequence was amplified by PCR to eliminate the KDEL motif previously engineered at the 3' end of the template._The PCR product was digested with EcoRI and SpeI and ligated into pEF-IRES-puro6. (C) The CXCL16 coding sequence was amplified by PCR with primers to append nucleotides encoding a six histidine tag at the 3' end. The PCR product was digested with EcoRI and EcoRV and ligated into pEF-IRES-puro6 which had been similarly digested. (D) The CXCL16₍₉₋₂₂₀₎ construct was generated using overlap extension PCR as outlined in Figure 3.3 below. The final PCR product was digested with EcoRV and ligated into pEF-IRES-puro6.



Figure 3.3 Generation of the $CXCL16_{(9-220)}$ -His construct by overlap extension *PCR*.

The CXCL16₍₉₋₂₂₀₎ construct was amplified from wild-type CXCL16 template using overlap extension PCR. (A) PCR primers (blue and purple arrows) complementary to the signal sequence and primers complementary to the nucleotide sequence encoding amino acid residues 9 to 220 of CXCL16 were designed. The primers for the 3' end of the signal sequence and the 5' end of the antagonist portion were designed such that they included a complementary region (purple). (B) The CXCL16 signal sequence and antagonist portion were amplified in separate polymerase chain reactions. (C) The PCR products from (B) were combined in a third polymerase chain reaction. The complementary region appended to the signal sequence in the first round of PCR enabled the two PCR products to anneal to each other. Thus they formed a template from which a single product could be amplified by the external PCR primers used in the first round of PCR. The external primers contained EcoRI and EcoRV restriction sites to facilitate cloning of the final product into the pEF-IRES-puro6 vector. (D) CXCL16(9-220)-His was purified by gel electrophoresis. The purified product was then digested with EcoRI and EcoRV and cloned into pEF-IRESpuro6. The construct was sequenced to confirm that the overlap extension PCR had produced the desired sequence.





Figure 3.4 Vector maps of chemokine constructs.

(A) CXCL12-His::pEF-IRES-puro6.
(B) CXCL12_(P2G)-His::pEF-IRES-puro6.
(C) CXCL16₍₉₋₂₂₀₎-His::pEF-IRES-puro6.
(D) CXCL16FL-His::pEF-IRES-puro6.
(E) CXCL16₍₉₋₂₂₀₎-His::pEF-IRES-puro6. The nature and function of the various elements of each construct are indicated in the legend.







Figure 3.5 Expression of CXCL12 by 4T1.2 cells.

(A) Total RNA was extracted from 4T1.2 cells (5 x 10⁶) or excised 4T1.2 tumours using TRIzol[®] reagent according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Section 2.4.2. The resulting cDNA was subjected to 35 cycles of PCR with mouse CXCL12 primers or glutaraldehyde 3-phosphate dehydrogenase (GAPDH) primers as a control. +, positive control; -, negative control. For CXCL12 primers, H₂O was added to the negative control instead of template. For GAPDH negative controls: *1*, reverse transcription control with no reverse transcriptase added; *2*, reverse transcription control with no template added; *3*, PCR control with no template added. (B) 4T1.2 cells were cultured for 48 hours in complete medium. Culture supernatant was harvested, passed through a 2µm filter and protease inhibitor cocktail was added at a 1:100 dilution. The supernatant was then assayed for CXCL12 content by ELISA as described in Section 2.4.3. Culture medium alone served as a background control for comparison.

This page left blank intentionally.





(A) Total RNA was extracted from 4T1.2 cells (5 x 10^6) or excised 4T1.2 tumours using TRIzol[®] reagent according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Section 2.4.2. The resulting cDNA was subjected to 35 cycles of PCR with mCXCL16 primers or GAPDH primers as a control. +, positive control; -, negative control. For CXCL16 primers, H₂O was added to the negative control instead of template. For GAPDH negative controls: 1, reverse transcription control with no reverse transcriptase added; 2, reverse transcription control with no template added; 3, PCR control with no template added. (B) 4T1.2 cells were cultured for 48 hours in complete medium. Culture supernatant was harvested, passed through a 2µm filter and protease inhibitor cocktail was added at a 1:100 dilution. The supernatant was then assayed for CXCL16 content by ELISA as described in Section 2.4.3. Culture medium alone served as a background control for comparison. (C) 4T1.2 cells were stained with a biotinylated anti-mouse CXCL16 antibody or isotype-matched control followed by streptavidin-conjugated phycoerythrin as described in Section 2.4.1. Fluorescing cells were quantified on the same day using an LSRII flow cytometer. Shaded region, anti-CXCL16 antibody. Solid line, isotype-matched control antibody.

Figure 3.7 Expression of CXCR4 and CXCR7 by 4T1.2 cells.

(A) CXCR4 and CXCR7 mRNA expression in 4T1.2 cells. Total RNA was extracted from 10⁷ cells using TRIzol[®] reagent according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Section 2.4.2. The resulting cDNA was subjected to 35 cycles of PCR with mCXCR4 or mCXCR7 primers, or GAPDH primers as a control. +, positive control; -, negative control. For CXCR4 and CXCR7 primers, H₂O was added to the negative control instead of template. For GAPDH negative controls: 1, reverse transcription control with no reverse transcriptase added; 2, reverse transcription control with no template added; 3, PCR control with no template added. (B) Cell surface expression of CXCR4. Cells were fixed in 4% formaldehyde in PBS for 10 minutes prior to staining (300.19 cells) or left unfixed (4T1.2 cells) and incubated with a monoclonal rat anti-mouse CXCR4 antibody or isotype-matched control followed by streptavidin-conjugated phycoerythrin as described in Section 2.4.1. Fluorescing cells were quantified on the same day using a FACScan flow cytometer. Histograms are shown for 4T1.2 (25.72% of cells CXCR4 positive after subtracting background determined from isotype control) and positive control 300.19 cells (22.49% of cells CXCR4 positive after subtracting background determined from isotype control). Shaded region, anti-CXCR4 antibody. Solid line, isotype-matched control antibody. Representative data from one experiment are shown. (C) Intracellular calcium mobilisation in response to CXCL12 stimulation of 4T1.2 cells. Approximately 10⁷ cells were incubated with 2µM FURA-2AM for 30 minutes at 37°C in 5% CO₂. Cells were washed once in calcium buffer (Section 2.6.2.4) and then stimulated at the indicated time points with buffer alone, control peptide (CCL2_(Ala) at 300ng/ml) or mouse CXCL12 (300ng/ml). 300.19 cells were used as a positive control. Data are normalized to background fluorescence readings for each sample.













Figure 3.8 Expression of CXCR6 by 4T1.2 cells.

(A) CXCR6 mRNA expression in 4T1.2 cells. Total RNA was extracted from 10^7 cells using TRIzol® reagent (Invitrogen) according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Chapter 2: Materials & Methods. The resulting cDNA was subjected to 35 cycles of PCR with mCXCR6 primers. +, positive control; -, negative control. For CXCR6 primers, H₂O was added to the negative control instead of template. For GAPDH negative controls: *1*, reverse transcription control with no reverse transcriptase added; *2*, reverse transcription control with no template added; 3, PCR control with no template added. (B) Cell surface expression of CXCR6. Cells were incubated with a polyclonal rabbit antimouse CXCR6 antibody or isotype-matched control followed by streptavidin-conjugated phycoerythrin as described in Section 2.4.1. Fluorescing cells were quantified on the same day using a FACScan flow cytometer. Histograms are shown for 4T1.2 (-2.94% of cells CXCR6 positive after subtracting background determined from isotype control) and positive control CXCR6-expressing 300.19 ('BCX6') cells (84.88% of cells CXCR6 positive after subtracting background determined from isotype control). Shaded region, anti-CXCR6 antibody. Solid line, isotype-matched control antibody. Representative data from one experiment are shown. (C) Intracellular calcium mobilisation in response to CXCL16 stimulation of 4T1.2 cells. Approximately 10^7 cells were incubated with 2µM FURA-2AM for 30 minutes at 37°C in 5% CO₂. Cells were washed once in calcium buffer (Section 2.6.2.4) and then stimulated at the indicated time points with buffer alone, control peptide (CCL2_(Ala) at 300ng/ml) or recombinant mouse CXCL16 (300ng/ml). BCX6 cells were used as a positive control. Data are normalized to background fluorescence readings for each sample.









С







94





* = clones selected for pooling



Figure 3.11 Level of CXCL16 construct production by individual clones from transfected 4T1.2 populations.

Individual clones from each of the CXCL16-producing transfected populations in Figure 3.9 were isolated by limiting dilution and plated in 96-well trays. Supernatants from wells containing individual colonies were harvested and assayed for CXCL16 content by ELISA as described in Section 2.4.3. The five highest CXCL16 construct-expressing clones from each transfected population *(indicated by asterisks)* were selected and pooled to generate high-expressing cell lines.


Figure 3.12 CXCL12 construct expression by cell lines derived from pooled clones.

(A) Total RNA was extracted from 5×10^6 cells from each of the cell lines derived from the pooled clones in Figure 3.10 using TRIzol[®] reagent according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Section 2.4.2. The resulting cDNA was subjected to 35 cycles of PCR with mCXCL12 primers or GAPDH primers as a control. +, positive control CXCL12::pEF-IRES-puro6 plasmid; -, negative control. For CXCL12 primers, H₂O was added to the negative control instead of template. For GAPDH primers, the negative controls were, from left to right: reverse transcription control with no reverse transcriptase added, reverse transcription control with no template added, and PCR control with no template added. (B) Cell lines derived from the pooled clones in Figure 3.10 were cultured in complete medium. Culture supernatant was harvested, passed through a 2µm filter and protease inhibitor cocktail was added at a 1:100 dilution. The supernatant was then assayed for CXCL12 content by ELISA as described in Section 2.4.3. Culture supernatants from the parental 4T1.2 line served as a background control for comparison (upper graph). ***, p<0.001; **, p<0.01, significantly different from control 4T1.2 culture supernatant, Kruskal-Wallis non-parametric ANOVA.

Figure 3.13 CXCL16 construct expression by cell lines derived from pooled clones.

(A) Total RNA was extracted from 5×10^6 cells from each of the lines derived from the pooled clones in Figure 3.11 using TRIzol[®] reagent according to the manufacturer's instructions. Five micrograms of total RNA were subjected to DNase treatment and reverse-transcribed as described in Section 2.4.2. The resulting cDNA was subjected to 35 cycles of PCR with mCXCL16 primers or GAPDH primers as a control. +, positive control CXCL16::pEF-IRES-puro6 plasmid; -, negative control. For CXCL16 primers, H₂O was added to the negative control instead of template. For GAPDH primers, the negative controls were, from left to right: reverse transcription control with no reverse transcriptase added, reverse transcription control with no template added, and PCR control with no template added. (B) Cell lines derived from the pooled clones in Figure 3.11 were cultured in complete medium. Culture supernatant was harvested, passed through a 2µm filter and protease inhibitor cocktail was added at a 1:100 dilution. The supernatant was then assayed for CXCL16 content by ELISA as described in Section 2.4.3. Culture supernatants from the parental 4T1.2 line or control 4T12Ala line served as background controls for comparison. ***, significantly different from control 4T1.2 and 4T12Ala culture supernatants, p<0.001, One Way ANOVA. (C) To detect surface-bound CXCL16, cells from each of the lines derived from the pooled clones in Figure 3.11 were stained with a biotinylated anti-mouse CXCL16 antibody or isotype-matched control followed by streptavidin-conjugated phycoerythrin as described in Section 2.4.1. Fluorescing cells were quantified on the same day using an LSRII flow cytometer. Shaded region, anti-CXCL16 antibody. Solid line, isotype-matched control antibody.



CXCL16

99



Figure 3.14 In vitro proliferation rates of CXCL12-expressing 4T1.2 cell lines.

In vitro proliferation rate of CXCL12-expressing cell lines. Cells were prepared at 10^5 cell/ml in serum-free medium and aliquots of 25µl were added to wells containing medium with 0.1%, 1% or 10% FCS. Cells were then allowed to proliferate at 37°C in 5% CO₂. After the indicated time points, the number of viable cells in each well was assessed by reduction of XTT (1mg/ml) added in a 50:1 ratio with PMS (1.25mM) to each well. Absorbance at 490nm (reference 650nm) was determined using a microplate reader, and background XTT reduction (determined from wells containing no cells) was subtracted. (A) Comparison of 4T1.2, 4T12Ala, 4TX12 1 and 4T12P2G lines. (B) Comparison of 4'T1.2, 4T12Ala, 4TX12 2 lines. Data points represent mean ± SD of 2 (A) or 3 (B) independent experiments. Differences between data points were not significant.



Figure 3.15 In vitro proliferation rates of CXCL16-expressing 4T1.2 cell lines.

In vitro proliferation rate of CXCL16-expressing cell lines. Cells were prepared at 10^5 cell/ml in serum-free medium and aliquots of 25µl were added to wells containing medium with (A) 0.1%, (B) 1% or (C) 10% FCS. Cells were then allowed to proliferate at 37°C in 5% CO₂. After the indicated time points, the number of viable cells in each well was assessed by reduction of XT1 (1mg/ml) added in a 50:1 ratio with PMS (1.25mM) to each well. Absorbance at 490nm (reference 650nm) was determined using a microplate reader, and background XTT reduction (determined from wells containing no cells) was subtracted. Data points represent mean ± SEM of 2 independent experiments. Differences between data points were not significant.



Figure 3.16 In vitro tumorigenicity of chemokine construct-expressing cell lines.

The wells of 6-well trays were coated with 0.7% agar made up in complete IMDM (Section 2.6.1.3), which was allowed to set. A layer of 0.3% agar containing 500 cells was then overlaid and allowed to solidify. Cells were incubated at 37° C in 5% CO₂ for 10 days. The number of colonies per well was then counted under a dissecting microscope. (A & B) CXCL12 construct-expressing cell lines. Bars represent mean ± SEM of 3-4 independent experiments with duplicate or triplicate determinations. Differences between columns were not significant, P=0.2059 (A) and P=0.3218 (B), One-Way ANOVA. (C) CXCL16 construct-expressing cell lines. Bars represent mean ± SEM of 2 independent experiments with duplicate determinations. Differences between not significant, P=0.2824, One-Way ANOVA.



CHAPTER 4 The Effect of chemokine expression on breast cancer *in vivo*

Ŵ

CHAPTER 4: THE EFFECT OF CHEMOKINE EXPRESSION ON BREAST CANCER *IN VIVO*

4.1 Overview

The ultimate aim of this project was to investigate the effects of CXCL12, CXCL16 and their antagonists on the progression of breast cancer *in vivo*. Thus growth and metastasis of tumours in syngeneic mice was examined to determine the ability of CXCL12 and CXCL16 to potentiate or inhibit these processes. These experiments provided a holistic picture of the effect of chemokine expression on tumour cells, stromal cells and infiltrating cells present in tumours *in situ*.

4.2 Effects of exogenous chemokine expression on primary tumour growth

To investigate the effect of CXCL12, CXCL16 and their antagonists on tumour growth, construct-expressing 4T1.2 cells and parental or control 4T12Ala cells were injected into the mammary fat pads of female Balb/c mice. Tumour growth was monitored for approximately one month, during which time tumour diameter was measured every two days in two perpendicular directions using electronic callipers. Tumour volume was calculated as width² x length/2, where width was the smaller diameter and length the larger. Tumours were also weighed at the end of the experiment to confirm the data generated from the calliper measurements.

Initial experiments compared the growth of CXCL12 construct-expressing tumours with that of control parental 4T1.2 tumours (Figure 4.1) or control 4T12Ala tumours (Figure 4.2). Growth of CXCL12-expressing tumours was significantly reduced compared with that of the parental 4T1.2 and control 4T12Ala tumours. This tumour-inhibitory effect of CXCL12 was confirmed in a subsequent experiment in which the growth of a second independently generated CXCL12-expressing cell line (4TX12 2) was compared with that of the original 4TX12 1 line (Figure 4.3). The 4TX12 2 tumours exhibited reduced growth compared with control 4T12Ala tumours, similar to 4TX12 1 tumours.

In contrast to the CXCL12-expressing tumours, growth of the CXCL12_(P2G)-expressing tumours was similar to controls (Figure 4.1 and Figure 4.2). This was apparent in terms of both the rate of tumour growth and final tumour weight, and was confirmed in subsequent experiments. A representative experiment is shown in Figure 4.4. These data raised two possibilities: one, that CXCL12 was not directly involved in primary tumour growth; or two, that the 4T12P2G cell line did not express CXCL12_(P2G) at sufficiently high levels to inhibit the effects of endogenous CXCL12 *in vivo*. The results obtained from study of metastasis in this model indicate that the former was the case (see below).

Growth of CXCL16-expressing and CXCL16₍₉₋₂₂₀₎-expressing tumours was also compared with that of control 4T12Ala tumours. Initial experiments suggested a slight trend for growth of 4T16 tumours to be enhanced and $4T\Delta16$ tumours to be reduced compared with controls (not shown). However, further experiments conducted to increase statistical power showed no significant differences in the rate of primary tumour growth between the control and 4T16 and $4T\Delta16$ lines (Figure 4.5). This was confirmed by both the calliper measurements and comparison of the final tumour weights at the end of the experiment.

4.3 Effects of CXCL12 construct expression on metastasis

To determine if expression of any of the constructs affected metastasis of 4T1.2 tumours, the lungs of tumour-bearing mice were harvested after approximately one month. To determine the frequency of metastasis, lungs were fixed and examined under the dissecting microscope for the presence of surface metastases.

Enumeration of surface metastatic nodules revealed that 4TX12 1 tumours metastasised spontaneously to the lungs significantly less frequently than control tumours (Figure 4.6), a finding that was not surprising given the inhibitory effect of CXCL12 on primary tumour growth. Unexpectedly, 4T12P2G tumours also spontaneously metastasised to the lungs less frequently than control tumours, despite the fact that CXCL12_(P2G) had no discernible effect on primary tumour growth. These results also confirmed that CXCL12_(P2G) was being expressed at sufficient levels to exert a biological effect in mice.

CXCL12 and CXCR4 are pro-angiogenic, and both have been shown to be expressed in $EC^{111, 117}$. It was therefore postulated that expression of $CXCL12_{(P2G)}$ might inhibit

angiogenesis within the tumour, thus reducing the opportunity for tumour cells to enter the bloodstream and metastasise to distant sites. Hence, angiogenesis was examined in 4T1.2 tumours to determine if this was a potential mechanism through which CXCL12_(P2G) might be acting to inhibit metastasis. To quantify the degree of vascularisation of CXCL12expressing tumours, the haemoglobin content of primary tumour homogenates was compared. CXCL12_(P2G)-expressing tumours contained similar levels of haemoglobin to control tumours (Figure 4.7). Immunohistochemical analysis of CD31 expression (a marker of EC) was also employed to assess vascularisation of 4T1.2 tumours. Considerable variability in the vascular density between tumours within the same group was observed. However, the blood vessel density was not found to be obviously decreased in CXCL12_(P2G)-expressing tumours compared with controls (Figure 4.8), in agreement with the haemoglobin results. Interestingly, overexpression of CXCL12 induced an increase in the haemoglobin content of tumour tissues; however, this increase was not statistically significant. Moreover, the vascular density of CXCL12-expressing tumours appeared reduced compared with control tumours (Figure 4.8). The implications of these apparently contradictory results will be discussed in Section 4.5 below and in Section 6.2.

Since $CXCL12_{(P2G)}$ did not seem to affect the haematogenous route of tumour metastasis, investigative efforts were directed towards the lymphatic route. The effect of $CXCL12_{(P2G)}$ expression on lymphangiogenesis in the primary tumour was assessed using immunohistochemical staining for LYVE-1, a marker of lymphatic endothelium. Again, considerable variability in the lymphatic vascular density between tumours within the same group was observed. However, no general reduction in the lymphatic vessel density of $CXCL12_{(P2G)}$ -expressing tumours compared with control tumours was apparent (Figure 4.9). Similarly, overexpression of CXCL12 did not affect lymphangiogenesis within primary tumours (Figure 4.9). Together, these data indicate that $CXCL12_{(P2G)}$ -mediated inhibition of 4T1.2 tumour metastasis is not a result of reduced vascularisation or lymphangiogenesis within the primary tumour.

Since there are multiple stages in the metastatic process at which $CXCL12_{(P2G)}$ could exert its inhibitory effect, experiments were undertaken to broadly ascertain when and where $CXCL12_{(P2G)}$ was acting. 4T12P2G cells, or control 4T12Ala cells for comparison, were injected intravenously into Balb/c mice. This bypassed the need for the cells to form a primary tumour and then escape from the tumour mass to metastasise. Similar to the results of the spontaneous metastasis experiments, $CXCL12_{(P2G)}$ expression resulted in the formation of fewer lung metastases than observed in controls as determined by enumeration of surface lung nodules (Figure 4.10). Given that $CXCL12_{(P2G)}$ had no effect on primary tumour growth, it seems most plausible that $CXCL12_{(P2G)}$ inhibits survival of metastatic cells in the circulation and/or homing to and/or extravasation at secondary sites. It is important to note that these data do not rule out a role for $CXCL12_{(P2G)}$ in escape of cells from the primary tumour. Of note, intravenously injected 4TX121 cells also produced fewer lung metastases than control tumour cells, indicating that the anti-tumour properties of CXCL12 are still in effect once metastatic cells leave the primary tumour. The implications of these findings will be discussed in more detail in Chapter 6.

4.4 Effects of CXCL16 construct expression on metastasis

When spontaneous metastasis to the lungs of mice bearing control, 4T16 and 4T Δ 16 tumours was compared, no statistically significant differences in the incidence of lung metastasis was observed (Figure 4.11A). Interestingly, upon intravenous injection, the 4T Δ 16 cell line formed significantly fewer lung metastases than the control 4T12Ala line (Figure 4.11B). In light of this, it is surprising that CXCL16₍₉₋₂₂₀₎ expression had no effect on spontaneous metastasis. A possible explanation for this apparent paradox is discussed in Section 6.4. It should be noted that due to time constraints, the metastasis assays were performed once only. Additional experiments would provide more statistical power and confirm the observed effect of CXCL16₍₉₋₂₂₀₎ on metastasis.

4.5 Summary and preliminary discussion

In summary, these experiments show that CXCL12 overexpression has a profound suppressive effect on primary tumour growth and metastasis in the 4T1.2 model. Given the lack of effect of CXCL12 on growth of 4T1.2 cells *in vitro*, it seems probable that CXCL12 exerts its effect through host components, rather than through direct effects on tumour cells themselves. However, the possibility that tumour cells become sensitive to CXCL12 signalling under growth conditions encountered *in vivo* cannot be ruled out. Analysis of the effect of CXCL12 on tumour angiogenesis indicated that there was an increase, though not statistically significant, in the haemoglobin content of CXCL12-expressing tumours. This result made a stark contrast with the apparent reduction in the

vascular density of these same tumours. Possible explanations for these divergent results are discussed further in Section 6.2, but it seems likely that any effects of CXCL12 on the tumour vasculature play a minor role, if any, in this model. The precise mechanisms by which CXCL12 influences the host are further investigated in the next chapter.

Expression of the CXCL12 antagonist, CXCL12_(P2G), inhibited metastasis of tumour cells to the lungs of tumour-bearing mice, despite having no overt inhibitory effect on primary tumour growth, angiogenesis or lymphangiogenesis. However, the potential for CXCL12_(P2G) to influence angiogenesis and lymphangiogenesis should not be discounted, for reasons to be discussed further in Section 6.3. Of particular interest, CXCL12_(P2G) was able to inhibit lung metastasis of tumour cells injected directly into the circulation, indicating the importance of endogenous CXCL12 in metastatic steps following detachment of cells from the primary tumour and intravasation. Blockade of CXCL12 signalling through CXCR4 may be responsible for the effects observed, but it should not be forgotten that 4T1.2 cells also express CXCR7. The ability of CXCL12_(P2G) to bind CXCR7 and to block CXCR7-associated functions (e.g. enhancement of adhesion to EC) needs to be further investigated to fully dissect the mechanism behind CXCL12_(P2G) inhibition of metastasis.

Neither CXCL16 nor its antagonist had any discernible effect on primary tumour growth or spontaneous metastasis. Surprisingly, $4T\Delta 16$ cells metastasised significantly less frequently to the lungs when injected intravenously, despite the expression of CXCL16₍₉₋₂₂₀₎ having no effect on spontaneous metastasis. As these metastasis experiments were conducted only once, further experiments to increase statistical power are required before any solid conclusions can be drawn. However, a hypothesis that could explain the reduced metastasis of intravenously injected $4T\Delta 16$ cells is put forward in Section 6.4.



Figure 4.1 Comparison of the growth of CXCL12 construct-expressing 4T1.2 mammary tumours and parental 4T1.2 tumours in vivo.

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SD from one representative experiment of two performed (4T1.2, *n*=24; 4TX12 1 and 4T12P2G, *n*=15). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight ± SD from one representative experiment of two performed (4T1.2, *n*=24; 4TX12 1, *n*=24; 4TX12 1,



Figure 4.2 Comparison of the growth of CXCL12 construct-expressing 4T1.2 mammary tumours and control 4T12Ala tumours in vivo.

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SD from one representative experiment of two performed (*n*=12 mice per group). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight ± SD from one representative experiment of two performed (*n*=12 mice per group). ****, P<0.0001 (Kruskal-Wallis), significantly different compared with control 4T12Ala and 4T12P2G tumours.





Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume \pm SD (*n*=12 mice per group). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight \pm SD (*n*=12 mice per group). Final tumour weights were significantly different, P<0.0001 (Kruskal-Wallis). ***, P<0.001 and *, P<0.05 significantly different compared with control 4T12Ala tumours (Dunn's post test).



Figure 4.4 Growth of CXCL12_(P2G)-expressing 4T1.2 mammary tumours in vivo.

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume \pm SD (*n*=12 mice per group). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight \pm SD (*n*=12 mice per group). Final tumour weights were not significantly different (P=0.7125, Mann-Whitney U test).





Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SEM from 4 independent experiments (*n*=6, 10, 6 and 12, respectively). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight ± SEM from 4 independent experiments (*n*=6, 10, 6 and 12, respectively). Final tumour weights were not significantly different (P=0.1183, Kruskal-Wallis).



Figure 4.6 Spontaneous metastasis to the lungs of 4TX12 1 and 4T12P2G tumourbearing mice.

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 cells. After 3-4 weeks, lungs were harvested and fixed overnight in 4% formaldehyde in PBS. Lungs were washed twice for at least two hours in 30% sucrose solution and were stored in 70% ethanol at 4°C until analysis. The five lobes of each lung were separated and examined under a dissecting microscope for surface metastatic nodules. Data points represent the number of surface metastatic nodules in the lungs for individual mice from two independent experiments. Bars represent the median. Medians were significantly different, P=0.0003 (Kruskal-Wallis). ***, P<0.001; *, P<0.05 (Dunn's multiple comparisons post test), compared with control 4T1.2 tumour-bearing mice.



Figure 4.7 Haemoglobin content of CXCL12-construct expressing tumours.

Tumours were snap frozen in liquid nitrogen and homogenized to a fine powder. The powdered tumour was weighed before being resuspended in 500µl deionised water and sonicated for approximately 30 seconds. Debris was removed by centrifugation and the resulting supernatant was then used for the haemoglobin assay. Drabkin's reagent was prepared containing Brij 35 Solution as per the manufacturer's instructions. Fifty microlitres per well of tumour supernatants were transferred to a 96-well tray and 200µl/well of Drabkin's solution was added. The plate was incubated for 15 minutes at room temperature to allow colour to develop and then the absorbance at 562nm was determined. A standard curve was generated using haemoglobin from bovine blood dissolved in deionised water.



Figure 4.8 Vascularisation of CXCL12 construct-expressing tumours.

CD31 staining of EC in sections of CXCL12 construct-expressing 4T1.2 tumours. Sections were stained with rat anti-mouse CD31 antibody followed by HRP-conjugated anti-rat IgG antibody. Binding of antibodies was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs showing areas of greatest vascular density of individual tumours are shown. Magnification: 100x. (A) Isotype control. (B-D) 4T12Ala tumours. (E-G) 4TX12 1 tumours. (H-J) 4T12P2G tumours.

This page left blank intentionally.



Figure 4.9 Lymphangiogenesis within CXCL12 construct-expressing tumours.

LYVE-1 staining of lymphatic vasculature in sections of CXCL12 construct-expressing 4T1.2 tumours. Sections were stained with rat anti-mouse LYVE-1 antibody followed by HRP-conjugated anti-rat IgG antibody. Binding of antibodies was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs showing areas of greatest lymphatic vascular density of individual tumours are shown. Magnification: 100x. (A) Isotype control. (B) Positive control staining of a LN. (C-E) 4T12Ala tumours. (F-H) 4TX12 1 tumours. (I-K) 4T12P2G tumours.

This page left blank intentionally.



Figure 4.10 *Experimental metastasis of CXCL12-expressing tumour cell lines to the lungs of Balb/c mice.*

Female 6-8 week old Balb/c mice were injected in the tail vein with 5×10^5 cells (A) or 7.5 x 10^5 cells (B) in PBS. After approximately 2 weeks, lungs were harvested and fixed overnight in 4% formaldehyde in PBS. Lungs were washed twice for at least two hours in 30% sucrose solution and were stored in 70% ethanol at 4°C until analysis. The five lobes of each lung were separated and examined under a dissecting microscope for surface metastatic nodules. Data points represent the number of surface metastatic nodules in the lungs of individual mice. Bars represent the median. (A) Medians were significantly different, P=0.0013 (Kruskal-Wallis). **, P<0.01; *, P<0.05 (Dunn's multiple comparisons post test), compared with mice injected with control 4T12Ala cells. (B) Medians were significantly different, P=0.0004 (Kruskal-Wallis). **, P<0.01 (Dunn's multiple comparisons post test), compared with mice injected with control 4T12Ala cells.



Figure 4.11 Metastases in the lungs of 4T16 and $4T\Delta 16$ tumour-bearing mice.

(A) Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1 x 10⁵ cells. After 31 days, lungs were harvested and fixed overnight in 4% formaldehyde in PBS. Lungs were washed twice for at least two hours in 30% sucrose solution and were stored in 70% ethanol at 4°C until analysis. The five lobes of each lung were separated and examined under a dissecting microscope for surface metastatic nodules. Data points represent the number of surface metastatic nodules in the lungs of individual mice. Bars represent the median. Medians were not significantly different, P=0.3231 (Kruskal-Wallis). (B) Female 6-8 week old Balb/c mice were injected in the tail vein with 7.5 x 10^5 cells in PBS. After approximately 2 weeks, lungs were harvested and fixed overnight in 4% formaldehyde in PBS. Lungs were washed twice for at least two hours in 30% sucrose solution and were stored in 70% ethanol at 4°C until analysis. The five lobes of each lung were separated and examined under a dissecting microscope for surface metastatic nodules. Data points represent the number of surface metastatic nodules in the lungs of individual mice. Bars represent the median. Medians were significantly different, P=0.0318 (Kruskal-Wallis). *, P<0.05 (Dunn's multiple comparisons post test), compared with mice injected with control 4T12Ala cells.



CHAPTER 5 The Effect of CXCL12 expression on the anti-tumour immune response

<u>CHAPTER 5: THE EFFECT OF CXCL12 EXPRESSION ON</u> <u>THE ANTI-TUMOUR IMMUNE RESPONSE</u>

5.1 Overview

As described in the previous chapter, CXCL12 expression had a striking inhibitory effect on the growth of 4T1.2 tumours. Since CXCL12 did not have any discernible effect on 4T1.2 cell growth *in vitro*, it was concluded that CXCL12 was most likely to be exerting its inhibitory effects by influencing host cells, rather than tumour cells. CXCL12 is proangiogenic, but the data presented in the previous chapter indicated that CXCL12 was unlikely to be influencing vascularisation in this model. Given that CXCL12 induces responses in a variety of leukocytes, it seemed most likely that tumour growth was being inhibited by a CXCL12-induced immune response. In this chapter, experiments to identify leukocyte subsets involved in the CXCL12-induced anti-tumour response, as well as the mechanisms by which they mediate their effects, are presented.

5.2 Identification of leukocyte subsets important for the anti-tumour effect of CXCL12

5.2.1 The role of T cells

The data presented in Section 4.2 suggested that the inhibitory effects of CXCL12 were not a consequence of its direct action on tumour cells. Rather, it seemed likely that it was exerting its activity on the tumour stroma and accessory cells, in particular, the cells of the immune system. To elucidate the mechanism by which growth of CXCL12-expressing tumours was inhibited, the CXCL12 construct-expressing cell lines were injected into the mammary fat pads of SCID mice, which lack both functional T and B cells. Growth of CXCL12-expressing tumours in these mice was comparable to controls, implying that a functional adaptive immune response is required for the tumour-inhibitory effects of CXCL12 (Figure 5.1).

To further clarify the role of particular lymphocyte subsets in the tumour inhibitory effect of CXCL12, Balb/c mice carrying the nu/nu mutation (nude mice) and lacking functional T cells were injected with control CXCL12_(Ala)-expressing or CXCL12-expressing tumour cells. There was a significant inhibition in the growth rate of CXCL12-expressing tumours

compared with control tumours in these mice (Figure 5.2). However, the reduction in 4TX12 1 tumour growth was less pronounced in nude mice compared with that observed in wild-type mice (Table 5.1). This partial abrogation of tumour growth inhibition indicates that T cells are involved in mediating the anti-tumour effect of CXCL12. However, the incomplete loss of the anti-tumour response suggests that other cell subsets are also important for CXCL12-mediated tumour inhibition.

Further support for the role of T cells in the anti-tumour response came from correlative analysis of final tumour weight and the percentages and absolute numbers of both CD4⁺ and CD8⁺ splenocytes (Figure 5.3 and Table 5.2). For both 4T12Ala and 4TX12 1 tumours, the total number of splenocytes positively correlated with tumour weight (correlation coefficients 0.5874 and 0.8231, respectively). Additionally, a significant negative correlation of the percentage CD4⁺ splenocytes and tumour weight was observed (i.e. smaller tumour weight corresponded with a higher percentage of CD4⁺ cells in the spleen). However, this correlation was much stronger for 4TX12 1 tumours (correlation coefficient = -0.8462) than for control tumours (correlation coefficient = 0.6503) of the absolute number of CD8⁺ splenocytes and tumour weight was observed for 4TX12 1 tumours. Why an increased number of CD8⁺ splenocytes should be associated with enhanced tumour growth remains unclear.

T cells from tumour-DLN and tumours were isolated to characterise their surface marker phenotype (Figure 5.4). DLN and tumours were taken 7 days post-injection of tumour cells and single-cell suspensions were prepared for flow cytometric analysis. T cells were identified as CD4 or CD8 positive, and the expression of the markers CD25 and CD62L was examined. CD25 is the high affinity IL-2 receptor α -chain, which is up-regulated by T cells upon activation, although when expressed together with CD4, it also defines a population of T_{reg} cells. CD62L or L-selectin, is an adhesion molecule expressed on LN-homing cells, and is generally down-regulated upon activation. On the whole, no obvious differences in CD25 or CD62L expression were observed on either CD4 or CD8 cells isolated from LN or tumours. The exception to this was the dramatic reduction in CD4⁺CD25⁻ and CD4⁺CD62L⁻ cells seen in DLN compared with naïve mice. This reduction in the proportion of CD4⁺ cells did not seem to be specific to a particular tumour

group; rather, it appeared to be a phenomenon common to all tumour-bearing mice. A reduction in $CD8^+$ cells in the DLN of tumour-bearing mice was also noted; however, the difference was not as dramatic or as consistent as that observed for $CD4^+$ cells.

In another approach, sections of primary tumours were examined for infiltration of $CD4^+$ and $CD8^+$ T cells to assess the relative contribution of these T cell subsets to the reduction of 4TX12 1 tumour growth. However, no difference in the level of T cell infiltration into CXCL12-expressing and control tumours was apparent at either early (Figure 5.5) or late stages of tumour growth (Figure 5.6 and Figure 5.7).

5.2.2 The role of dendritic cells

The data outlined above suggested that CXCL12 was not acting by altering T cell infiltration of tumours, and this led to the hypothesis that CXCL12 was acting to enhance the activation state of those cells that enter the tumour. DC are potent antigen-presenting cells and are responsible for the activation of T cells during the initiation of an immune response. CD11c is the α_x chain of the α_x - β_2 integrin, and is expressed predominantly on conventional DC. To determine if an increase in DC priming and hence activation of T cells was occurring, immunohistochemistry to detect CD11c in tumour-DLN and tumours was undertaken. Tumour-DLN (Figure 5.8) from 4TX12 1 tumour-bearing mice, but not the tumours (Figure 5.9), were found to harbour an increased proportion of CD11c⁺ cells compared with controls by day 6 post-injection of tumour cells. This suggests that CXCL12 overexpression by the tumour enhances DC accumulation in the DLN, which in turn may promote the activation of T cells.

5.2.3 The role of NKT cells

As nude mice do not lack natural killer T (NKT) cells^{279, 280}, while SCID mice do²⁸¹, and since these cells have documented tumour cytotoxic activity^{279, 280}, the role of NKT cells in the CXCL12-induced anti-tumour effect was investigated. The CXCL12 construct-expressing cell lines were injected into the mammary fat pads of J α 18^{-/-} mice. These mice lack Type I NKT cells, which express an invariant TCR containing the V α 14-J α 18 α -chain, however Type II CD1d-dependent V α 14-J α 18-independent NKT cells are still present²⁸². There was a statistically significant inhibition of the growth of primary CXCL12-

expressing tumours in $J\alpha 18^{-/-}$ mice, similar to wild-type mice (Figure 5.10). This suggests that Type I NKT cells are not required for the anti-tumour effects of CXCL12 overexpression.

5.3 The role of cell-mediated immunity

It was postulated that increased activation of T cells in 4TX121 tumour-bearing mice could lead to an enhanced cell-mediated anti-tumour immune response. Since T cells are capable of inducing cytolysis directly, the quality of the cell-mediated cytotoxic response against control and CXCL12-expressing tumours was compared using an in vitro cytotoxicity assay (Figure 5.11). Lymphocytes were isolated from naïve mice or mice bearing control 4T12Ala or 4TX12 1 tumours for use as effectors. Wild-type 4T1.2 cells were used as targets. In the initial experiment (Figure 5.11A), effector cells were obtained from mice bearing late-stage tumours (30 days post injection) by pooling lymphocytes isolated from spleens and tumour-draining (i.e. inguinal and brachial) LN. In general, the level of cytotoxicity towards 4T1.2 targets was relatively low. However, at a 10:1 effectorto-target ratio, effector cells from 4TX121 tumour-bearing mice exerted significantly enhanced cytolytic activity compared with naïve effector cells and effector cells derived from 4T12Ala tumour-bearing mice. At lower effector-to-target cell ratios, no significant differences in cytotoxicity were observed between effector cells derived from the two tumour groups. Interestingly, at a 30:1 effector-to-target ratio, effector cells from 4T12Ala tumour-bearing mice exerted significantly less cytotoxicity than did effectors from 4TX121 tumour-bearing mice and naïve mice, suggesting that the basal level of cytotoxicity of effector cells from control tumour-bearing mice is suppressed.

At such a late stage of tumour growth, it was possible that the relatively weak cytotoxicity observed was a result of tumour-mediated immunosuppression mechanisms, and indeed the systemic immunosuppressive properties of 4T1 tumours have been documented²⁸³. Additionally, CXCL12-expressing tumours grew more slowly than controls from the outset, suggesting that the immune response to the tumour was becoming engaged early in tumour progression. Hence, effector cells from mice with early-stage tumours were tested for cytolytic function against 4T1.2 cells *in vitro* (Figure 5.11B). Considerable variability in the cytolytic activity of effector cells from individual mice was observed in these experiments. This was to be expected because at this early time point, the inherent

variability of the burgeoning immune response in individual mice is more apparent than at later stages when the immune response has become established. However, it was clear that the cytotoxic response of lymphocytes from mice with early-stage tumours was enhanced compared with that of lymphocytes from naïve mice. Moreover, there was a trend, though not significant, for effector cells from 4TX12 1 tumour-bearing mice to exhibit greater cytotoxicity than those from 4T12Ala tumour-bearing mice. Attempts were made to enhance the cytotoxic response of effector cells *in vitro* by re-stimulating them with mitomycin C-treated tumour cells prior to their addition to the cytotoxicity assay. However, the cytotoxic response induced by the conditions used was too strong for differences between effectors from the two tumour groups to be observed (data not shown). Further optimisation of the re-stimulation conditions (e.g. method of inactivating tumour cells, number of stimulators, length of re-stimulation time) may better highlight differences in the cytotoxic capabilities of effector cells from mice bearing 4T12Ala and 4TX12 1 tumours.

Further experiments were conducted to elucidate the mechanism by which effector cells from 4TX12 1 tumour-bearing mice were mediating their cytotoxic effects. Lymphocytemediated cytolysis is generally thought to proceed through two main pathways: the perforin/granzyme pathway or via members of the TNF death receptor family. Perforin and granzymes are major components of the cytotoxic granules of CTL and NK cells. On their release during granule-mediated killing, perforin prompts uptake of granzymes by target cells, which then induce cell death by initiating apoptotic signalling pathways²⁸⁴. Of the TNF family members, TRAIL is thought to have evolved to induce cell death in virally-infected cells whilst sparing normal cells, and is thought to be the main contributor to the elimination of transformed cells²⁸⁵.

Perforin (pfp)-deficient mice were obtained to determine the contribution of the perforin/granzyme pathway to CXCL12-mediated tumour growth inhibition, while TRAIL-deficient mice were obtained to determine the contribution of the TNF death receptor pathway (Figure 5.12). As in wild-type mice, there was a trend for 4TX12 1 tumour growth to be slower compared with 4T12Ala tumours in TRAIL^{-/-} mice. Though differences in final tumour weight were not statistically significant, differences in tumour volume were significant at most late time points. Two mice from the 4T12Ala group had to be culled earlier than the other mice in that group as they were displaying overt signs of

morbidity, which may have affected the statistical analysis. In pfp^{-/-} mice, while control 4T12Ala tumour growth was considerably reduced compared with that observed in TRAIL knockout and wild-type mice, the growth rate and final weight of these tumours was still significantly greater than that of 4TX12 1 tumours. Interestingly, for both pfp^{-/-} and TRAIL^{-/-} mice, the reduction in 4TX12 1 tumour growth was less pronounced compared with that observed in wild-type mice (Table 5.1). This partial abrogation of tumour inhibition indicates that perforin and TRAIL are involved in mediating the anti-tumour effect of CXCL12. However, the incomplete loss of the anti-tumour response suggests that any anti-tumour cytotoxic activity against CXCL12-expressing tumours is not solely dependent on either of these two cell death pathways alone. Deficiency in both cell death pathways may lead to complete abrogation of the CXCL12-mediated anti-tumour effect. Alternatively, other cell death pathways not tested here may be utilised in the response against the tumour, and should not be discounted.

5.4 The role of cytokines

While T cells may mediate the cytotoxic response against CXCL12-expressing tumours directly, they may also exert anti-tumour functions indirectly through production of cytokines that mobilise other important effector cell types, e.g. macrophages and NK cells. The signature cytokine produced in a cell-mediated immune response is interferon- γ (IFN- γ), which plays a critical role in the effector function of T cells, as well as NK cells and NKT cells³⁴. It exerts numerous effects particularly pertinent to anti-tumour immunity including activating the cytotoxic and phagocytic activity of macrophages²⁸⁶ and increasing the cytotoxic activity of NK cells³⁴. As shown in Figure 5.12, the cytotoxic activity of LN cells from 4TX12 1 tumour-bearing mice in vitro was relatively low, which suggested that other cells, such as innate immune cells, were responsible for cytotoxic killing of tumour cells. For this reason, it was postulated that T cells in the LN were exerting their major function through production of cytokines and in particular, IFN- γ . To determine whether IFN- γ was important for the anti-tumour effect of CXCL12, 4T12Ala and 4TX12 1 cells were injected into IFN- $\gamma^{-/-}$ mice and the development of tumours was observed (Figure 5.13). The pattern of 4T12Ala and 4TX12 1 tumour growth was similar in IFN- $\gamma^{-/-}$ mice to that in wild-type mice, in that the rate of tumour growth was slower for CXCL12-expressing tumours than for controls. However, tumours grew to be generally

larger in size compared with tumours grown in wild-type mice. Additionally, while CXCL12-expressing tumours were smaller compared with control tumours, this difference was not statistically significant. Furthermore, the ratio of 4TX12 1 tumour weight to 4T12Ala tumour weight was higher in IFN- $\gamma^{-/-}$ mice than in wild-type mice (Table 5.1), indicating that the tumour inhibitory effect of CXCL12 was partly lost in the absence of IFN- γ . However, the fact that 4TX12 1 tumours continued to grow more slowly than control 4T12Ala tumours demonstrates that IFN- γ is clearly not the sole contributor to the anti-tumour effect of CXCL12.

A recent report has described the importance of IL-23 in the promotion of tumour growth²⁸⁷. IL-23 plays a major role in the maintenance of Th17 cells which, as the name suggests, are major producers of the pro-inflammatory cytokine IL-17²⁸⁸. Inflammation is thought to be pro-tumorigenic²⁸⁹ and the requirement for IL-23 in tumour promotion implies a role for inflammatory IL-17-producing cells in this process. To investigate the hypothesis that CXCL12 may block the generation of IL-17⁺ cells, an ELISPOT assay for IL-17-producing cells was performed (Figure 5.14). Mice were injected with either control or CXCL12-expressing tumour cells and seven days later the tumour-DLN were removed. Lymphocytes were extracted by mechanical disruption of the LN, and the lymphocytes obtained were cultured overnight in the presence of PMA and ionomycin in an anti-IL-17coated filter plate. The following day, cells were washed from the plate and IL-17 secreted by the cells was detected with a secondary antibody and chromogenic substrate. 'Spots' of colour that developed in the bottom of the wells represented individual IL-17-producing cells. Enumeration of the spots in wells containing cells from 4T12Ala and 4TX121 tumour-bearing mice revealed that fewer IL-17-producing cells were present in the DLN of 4TX12 1 tumour-bearing mice.

The requirement of IFN- γ for the CXCL12-induced anti-tumour response and the reduction of IL-17-producing cells in mice bearing CXCL12-expressing tumours suggested that CXCL12 might serve to bias effector T cell differentiation away from an inflammatory IL-17-producing lineage to an IFN- γ -producing Type I lineage able to induce anti-tumour cell-mediated cytotoxicity. Differentiation of Type I IFN- γ -producing T cells also inhibits differentiation of Type II IL-4-producing T cells which promote humoral immunity and resistance to multicellular parasites²⁸⁸. To analyse the balance of .IFN- γ , IL-17 and IL-4

cytokine production in the 4T1.2 model, intracellular cytokine staining and flow cytometric analysis was employed (Figure 5.15). Cells were prepared from tumour-bearing mice as described for the ELISPOT assay above and cultured overnight with PMA and ionomycin to enhance cytokine production. The following day, cells were stained for cytokine production and analysed by flow cytometry. CD4⁺ and CD8⁺ cells were gated, and the expression of the aforementioned cytokines was assessed in each of these cell subsets. For CD4⁺ cells, there was a general reduction in the percentage of cells expressing each cytokine in mice bearing 4TX12 1 tumours compared with those bearing control tumours, though this only reached statistical significance for IL-4. A similar trend was observed within the CD8⁺ cell subset, although this did not reach statistical significance for any of the cytokines tested. Interestingly, a significant reduction in the proportion of IFN- γ^+ LN cells in general was observed in mice bearing 4TX12 1 tumours compared with those bearing control tumours (Figure 5.15C). The intensity of cytokine staining was also assessed to determine if cytokine-producing cells from different tumour groups were producing different amounts of cytokine. However, no significant differences between the groups were found.

The general reduction in cytokine-producing cells observed in DLN of 4TX12 1 tumours compared with the DLN of 4T12Ala tumours as described above was unexpected. Thus an alternative approach using a cytokine bead array to quantify cytokine secretion by DLN cells was employed. The range of cytokines analysed was expanded to include other effector cytokines, as well as anti-inflammatory cytokines, to provide a more complete picture of the cytokine milieu induced during the anti-tumour immune response. A panel of ten cytokines was assessed, including IFN- γ , IL-17 and IL-4, as well as TNF α , GM-CSF, IL-1 α , IL-2, IL-5, IL-6 and IL-10. Lymphocytes were extracted from the DLN of tumours 7 days post injection and cultured overnight in the presence of PMA and ionomycin to enhance cytokine secretion. Culture supernatants were then analysed using the cytokine bead array kit. Of the ten cytokines tested, the most notable differences in cytokine expression between tumour groups were observed for TNF- α and IL-4 (Figure 5.16). TNF- α secretion by DLN cells from 4TX12 1 tumour-bearing mice was elevated compared with that secreted by cells from control tumour-DLN, however the difference did not reach statistical significance. Production of IL-4 was increased in cells from 4T12Ala DLN compared with cells from naïve LN and 4TX12 1 DLN, but again, the difference was not statistically significant.

5.5 Summary of findings and preliminary discussion

In this chapter, it was established that CXCL12-mediated tumour growth inhibition was absent in SCID mice, and partly abrogated in nude mice. These experiments highlighted the importance of the adaptive immune response, and particularly T cells, for the effect of CXCL12. It should be pointed out that the extent of metastasis was not formally assessed in these mice, although there was an indication in nude mice that the number of lung metastases was reduced in the 4TX12 1 tumour-bearing group, in parallel with the reduction of primary tumour size. It will be interesting in future to determine if metastasis in SCID mice similarly parallels primary tumour growth. Such experiments would give an indication of the contribution of both immune- and non-immune-mediated mechanisms to the decreased metastatic capability of 4TX12 1 tumours.

Correlative analysis further supported the importance of T cells for CXCL12-induced antitumour immunity. This was apparent in the correlation of reduced tumour weight with an increased proportion of splenic CD4⁺ cells, particularly in mice bearing 4TX12 1 tumours. Curiously, the total number of splenic CD8⁺ cells in 4TX12 1 tumour-bearing mice correlated positively with tumour weight. Perhaps this is indicative of a failure of CD8⁺ T cells, which have been successfully activated, to leave the spleen and migrate to the tumour site. The lack of a similar accumulation of CD8⁺ cells in 4T12Ala spleens may reflect a relative absence of T cell activation and subsequent proliferation. Further study is required, however, to confirm this hypothesis and to elucidate the reason why CD8⁺ cells are retained in the spleens of some 4TX12 1 tumour-bearing mice.

Apart from implicating T cells in the CXCL12-induced anti-tumour immune response, the studies of tumour growth in immunocompromised mice indicated that other cell subsets present in nude mice, but not in SCID mice, were required. These were not Type I NKT cells, as determined from an experiment in Ja18^{-/-} mice lacking these cells. This by no means rules out a role for NKT cells; Type II NKT cells, which are CD1d-dependent but do not express the invariant V α 14-J α 18 chain, are still present in Ja18^{-/-} mice. It may be that Type II, rather than Type I, NKT cells play a part in CXCL12-induced tumour

growth inhibition. As B cells are also lacking in SCID, but not nude, mice, it is possible that they too are involved in the anti-tumour response. However, time constraints prevented investigation of their input in this study. The potential contribution of NKT and B cells to the anti-tumour effect of CXCL12 is discussed further in Section 6.2.

Tumour expression of CXCL12 was found to increase the number of $CD11c^+$ cells in tumour-DLN, suggesting that there was an increase of DC infiltrating the DLN to prime T cells. This discovery led to the hypothesis that T cell activation was enhanced in mice bearing CXCL12-expressing tumours. In line with this, cell-mediated cytotoxicity was found to be enhanced in mice bearing 4TX12 1 tumours, as demonstrated by the partial requirement for the cytotoxic mediators perforin and TRAIL and by *in vitro* cytotoxicity assays.

IFN- γ was also shown to be required for CXCL12-induced anti-tumour immunity through the partial loss of CXCL12-mediated tumour growth inhibition in IFN- γ -deficient mice. Further analysis of cytokine expression within tumour-DLN in wild-type mice indicated that IL-17⁺ and IL-4⁺ cells were reduced in 4TX12 1 tumour-DLN compared to 4T12Ala tumour-DLN. Together, these findings suggest that 4T1.2 tumour growth is promoted through bias of the immune system away from a more productive cell-mediated antitumour response, and that expression of CXCL12 may serve to tip the scales back in favour of this form of immunity. Paradoxically, the proportion of IFN- γ^+ cells in 4TX12 1 DLN was reduced as assessed by intracellular cytokine staining, and cytokine bead array analysis did not reveal any difference in the level of IFN- γ production by DLN cells from the two tumour groups. Possible explanations for these discrepancies are discussed in Section 6.2, but may reflect limitation of the analysis to the DLN. Assessment of systemic cytokine levels or cytokine levels within the tumour may give a more complete picture of their role in the immune response to 4TX12 1 tumours.

As presented in this chapter, intriguing insights into the anti-tumour immune response induced by CXCL12 have been gained. However, there remain many questions to be answered about the cells involved and the mechanisms by which they exert their effects. Much information might be gained by study of cells and cytokines within tumours, and this represents an avenue of investigation to be followed in future. Attempts were made to
further characterise T cells and CD11c⁺ cells infiltrating tumours, as well as to identify differences in the levels of innate immune cells within tumours (Figure 5.4B and data not shown). However, time constraints prevented sufficient optimisation of the protocols used and further experiments are required for meaningful conclusions to be drawn. The contribution of innate immune cells, and particularly macrophages, to the anti-tumour response is of particular interest, as is the nature of immunosuppressive mechanisms induced by the tumour. Study of these aspects of the immune response will provide valuable insight into the mechanism of CXCL12-induced anti-tumour immunity and build on the work presented in this thesis.

This page left blank intentionally.

Strain	Experiment –	Mean tumour weight (g)		4TX12 1	% inhibition of
		4T12Ala	4TX12 1	tumour weight as % of control	4TX12 1 tumour growth
Wild-type	1	0.83	0.20	24.2	75.8
	2	0.90	0.21	22.9	77.1
Nude	1	0.81	0.66	81.1	18.9
	2	0.91	0.48	52.7	43.3
pfp ^{-/-}	1	0.79	0.49	61.4	38.6
TRAIL-/-	1	0.78	0.44	57.1	42.9
IFN-γ ^{-/-}	1	1.21	0.64	52.9	47.1

Table 5.1 Comparison of tumour growth inhibition by CXCL12 in wild-type and nude mice.

 Table 5.2 Correlation of splenic T cell populations with final tumour weight.

	Spearman p	P value	Significance
4T12Ala			
Total splenocytes	0.5874	0.0446	*
% CD4 ⁺ splenocytes	-0.5849	0.0457	*
Number of CD4 ⁺ splenocytes	0.4755	0.1182	ns
% CD8 ⁺ splenocytes	-0.3298	0.2951	ns
Number of CD8 ⁺ splenocytes	0.4266	0.1667	ns
4TX12 1			
Total splenocytes	0.8231	0.0010	**
% CD4 ⁺ splenocytes	-0.8462	0.0005	***
Number of CD4 ⁺ splenocytes	0.07692	0.8122	ns
% CD8 ⁺ splenocytes	-0.5464	0.0660	ns
Number of CD8 ⁺ splenocytes	0.6503	0.0220	*



Figure 5.1 Growth of CXCL12-expressing tumours in SCID mice.

Female 6-8 week old SCID mice were injected in the fourth mammary fat pad with 1×10^5 tumour cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume \pm SD (*n*=5 mice per group). One representative experiment of two is shown. No significant differences between groups (Mann-Whitney U test). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight \pm SEM (*n*=10 mice per group from 2 independent experiments). No significant differences between groups. (P=0.9985, Student's t test.)



Figure 5.2 Growth of CXCL12-expressing tumours cells in nude mice.

Female 6-8 week old Balb/c mice carrying the nude mutation were injected in the fourth mammary fat pad with 1 x 10⁵ tumour cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume \pm SD. *, P<0.01; #, P<0.001, Mann-Whitney U test, *n*=11-12 mice per group. (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight \pm SD. ***, P<0.0001, Mann-Whitney U test, *n*=11-12 mice per group.

Figure 5.3 *Correlation of the number and proportion of splenic T cells subsets with tumour weight.*

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 tumour cells. Final weights of tumours were determined at the termination of the experiment 29 days later. Spleens were removed and single cell suspensions were prepared for flow cytometric labelling with phycoerythrin-conjugated rat anti-mouse CD4, CD8, or matched isotype control. Fluorescing cells were quantified on the same day using an LSRII flow cytometer. Percentages and absolute numbers of CD4⁺, CD8⁺ and total splenocytes were determined and correlated with final tumour weight for each mouse. Data points represent the measurements for individual mice from two independent experiments. (A-C) 4T12Ala tumour-bearing mice. **(D-F)** 4TX12 1 tumour-bearing mice. (A & D) Correlation weight with total number of tumour of splenocytes. (B & E) Correlation of tumour weight with total number (filled squares) and percentage (open squares) of CD4⁺ splenocytes. (C & F) Correlation of tumour weight with total number (filled circles) and percentage (open circles) of CD8⁺ splenocytes. For a summary of the Spearman's p correlation coefficients and P values, refer to Table 5.2.





Figure 5.4 *Characterisation of T cell subsets from early stage tumours and tumour draining lymph nodes.*

Female 6-8 week old Balb/c mice were injected in the hind foot pad with 1×10^6 tumour cells or PBS (naïve controls). Seven days later, draining (popliteal) LN and tumours were removed and single cell suspensions were prepared for flow cytometric labelling. Cells were stained with fluorochrome-conjugated rat anti-mouse CD4, CD8, CD25 and CD62L antibodies or matched isotype controls. Fluorescing cells were quantified on the same day using an LSRII flow cytometer. (A) Quantitation of the proportion of cells obtained from tumour-DLN expressing the indicated markers. (B) Quantitation of the proportion of cells obtained from (n=3-4 mice per group) from one representative experiment of three independent experiments.





CD4 and CD8 staining of infiltrating T cells in sections of early stage CXCL12 constructexpressing 4T1.2 tumours. Sections of tumours obtained either two (A, B, E, F, I, J) or six (C, D, G, H, K, L) days post tumour cell injection were stained with biotinylated rat antimouse CD4 or CD8 antibody followed by streptavidin-conjugated HRP. Binding of antibodies was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs of individual tumours are shown. Magnification: 100x. (A, C, E, G, I, K) Control 4T12Ala tumours. (B, D, F, H, J, L) 4TX12 1 tumours. (A-D) Isotype control staining. (E-H) CD4 staining. (I-L) CD8 staining. This page left blank intentionally.



Figure 5.6 *CD4⁺ T cell infiltration of late stage CXCL12-expressing 4T1.2 tumours.*

CD4 staining of infiltrating T cells in sections of late stage CXCL12 construct-expressing 4T1.2 tumours. Sections of tumours obtained approximately 4 weeks post tumour cell injection were stained with rat anti-mouse CD4 antibody followed by streptavidin-conjugated HRP. Binding of the antibody was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs of two individual tumours from each group are shown. Magnification: 200x. (A) Isotype control staining. (B & C) Control 4T12Ala tumours. (D & E) 4TX121 tumours.

This page left blank intentionally.



Figure 5.7 *CD8*⁺ *T cell infiltration of late stage CXCL12-expressing 4T1.2 tumours.*

CD8 staining of infiltrating T cells in sections of late stage CXCL12 construct-expressing 4T1.2 tumours. Sections of tumours obtained approximately 4 weeks post tumour cell injection were stained with rat anti-mouse CD8 antibody followed by streptavidin-conjugated HRP. Binding of the antibody was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs of two individual tumours from each group are shown. Magnification: 100x. (A) Isotype control staining. (B & C) Control 4T1.2 tumours. (D &E) 4TX12 1 tumours.

This page left blank intentionally.



Figure 5.8 Accumulation of $CD11c^+$ cells in the tumour-draining lymph nodes.

 $CD11c^+$ cells in sections of 4T1.2 tumour-DLN. Sections of tumour-draining (inguinal **[B-D]** and brachial **[E-G]**) LN were obtained six days post tumour cell injection. Sections were stained with biotinylated hamster anti-mouse CD11c antibody followed by streptavidin-conjugated HRP. Binding of antibody was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs of individual LN are shown. Magnification: 100x. (A) Isotype control. (B & E) LN from naïve mice. (C & F) DLN from control 4T12Ala tumour-bearing mice. (D & G) DLN from 4TX12 1 tumour-bearing mice.

This page left blank intentionally.



Figure 5.9 *CD11c*⁺ *cells within CXCL12-expressing tumours.*

 CD11c^+ cells in sections of CXCL12 construct-expressing 4T1.2 tumours. Sections of tumours obtained six days post tumour cell injection were stained with biotinylated hamster anti-mouse CD11c antibody followed by streptavidin-conjugated HRP. Binding of antibody was revealed by incubation with DAB substrate and cell nuclei were counterstained with haematoxylin. Sections were mounted and examined by light microscopy. Representative micrographs of individual tumours are shown. Magnification: 100x. (A) Isotype control. (B) Naïve skin. (C) Control 4T12Ala tumour. (D) 4TX12 1 tumour.

This page left blank intentionally.



Figure 5.10 *Growth of CXCL12-expressing tumours in invariant NKT cell-deficient mice.*

Female 6-8 week old $J\alpha 18^{-/-}$ mice were injected in the fourth mammary fat pad with 1 x 10⁵ tumour cells. Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SD (*n*=7 mice per group). *, P<0.05; **, P<0.01; ***, P<0.001, Mann-Whitney U test.



Figure 5.11 Cytotoxic activity of lymphocytes against 4T1.2 tumour cells in vitro.

(A) Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad with 1×10^5 tumour cells. After approximately 5 weeks, single cell suspensions were prepared from the pooled spleen and tumour-draining (brachial) LN for each mouse and used as effector cells in a cytotoxicity assay as described in Section 2.4.9. Target 4T1.2 cells were labelled with calcein, then plated at 2.5 x 10⁴ cells/well in 96-well trays and allowed to adhere prior to the addition of effector cells. Four days later, non-adherent cells were washed away, and fluorescence of remaining target cells was measured on a Molecular Imager FX. Percentage cytotoxicity was calculated as 100-(fluorescence of target cells/fluorescence of target cells only x 100). *, P<0.05, significantly different from naïve effectors and effectors from 4T12Ala tumour-bearing mice (Kruskal-Wallis). #, P<0.05, significantly different from naïve effectors and effectors from 4TX12 1 tumour-bearing mice (Kruskal-Wallis). (B) Female 6-8 week old Balb/c mice were injected in both hind foot pads with 1×10^6 cells per foot pad. Seven days later, lymphocytes were isolated from the draining popliteal LN and used as effector cells in a cytotoxicity assay. 4T1.2 cells plated at 1×10^4 cells/well were used as targets. After 3 days, non-adherent cells were washed away, and remaining target cells were labelled with calcein. Fluorescence was measured on a Molecular Imager FX and percentage cytotoxicity was calculated as 100-(fluorescence of target cells/fluorescence of target cells only x 100).



Figure 5.12 Growth of CXCL12-expressing tumours in perforin- and TRAILdeficient mice.

Female 6-8 week old $pfp^{-/-}$ or TRAIL^{-/-} mice were injected in the fourth mammary fat pad with 1 x 10⁵ tumour cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SD (n=5-7 mice per group). For pfp^{-/-} mice, *, P<0.05; **, P<0.01, Mann-Whitney U test. For TRAIL^{-/-} mice, #, P<0.05; ##, P<0.01, Mann-Whitney U test. GB) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight ± SD (n=5-7 mice per group). **, significantly different from control 4T12Ala tumour weight, P=0.0022, Mann-Whitney U test.





Female 6-8 week old IFN- $\gamma^{-/-}$ mice were injected in the fourth mammary fat pad with 1 x 10⁵ tumour cells. (A) Tumour diameter was measured every 2 days once tumours became palpable. Tumour volume in mm³ was calculated according to the equation width² x length/2. Data points represent the mean tumour volume ± SD (*n*=7 mice per group). No significant differences between groups (Mann-Whitney U test). (B) Final weights of tumours were determined at the termination of the experiment. Bars represent the mean final tumour weight ± SD (*n*=7 mice per group). No significant difference between groups (P=0.0530, Mann-Whitney U test).



Figure 5.14 *Quantitation of IL-17-producing cells in tumour-draining lymph nodes by ELISPOT assay.*

Female 6-8 week old Balb/c mice were injected in both hind foot pads with 1 x 10^6 tumour cells each. Seven days later, tumour-draining popliteal LN were obtained and macerated through a 70µm nylon sieve to generate single cell suspensions. Cells were prepared in complete medium containing PMA and ionomycin and aliquoted at 1.2×10^5 cells/well into the wells of a 96-well filter plate coated with anti-mouse IL-17 antibody. Cells were then incubated overnight at 37°C. The following day, cells were washed away and bound IL-17 was detected with biotinylated anti-mouse IL-17 followed by HRP-conjugated streptavidin. BCIP/NBT chromogenic substrate was used to visualise spots, and then plate was washed and allowed to dry. Spots representing individual IL-17-producing cells were counted under a dissecting microscope. Bars represent mean (\pm SD) percentage of LN cells expressing IL-17 (*n*=3 mice per group). *, P<0.05, Kruskal-Wallis test.



Figure 5.15 Intracellular cytokine staining of lymphocytes isolated from tumourdraining lymph nodes.

Female 6-8 week old Balb/c mice were injected in both hind foot pads with 1 x 10⁶ tumour cells each. Seven days later, draining popliteal LN were removed and single cell suspensions were prepared. Cells were incubated overnight with PMA and ionomycin at 37°C. Five hours prior to the end of the stimulation, GolgiStopTM was added to the cells to arrest cytokine secretion. Cells were then fixed, permeabilised and stained with fluorochrome-conjugated rat anti-mouse CD4 or CD8, and anti-mouse IFN- γ , IL-17 or IL-4. Fluorescing cells were quantified using an LSRII flow cytometer. Bars represent the mean \pm SEM (*n*=10 mice per group from 2 independent experiments). (A-C) Percentage of cells positive for cytokine expression. (D-F) Mean fluorescence index of cytokine positive cells. (A & D) Cytokine positive CD4⁺ cells. (B & E) Cytokine positive CD8⁺ cells. (C & F) Total cytokine positive cells. *, 4TX12 1 significantly different from control 4T12Ala group, P<0.05, Mann-Whitney U test.



Figure 5.16 *Cytokine production by cells isolated from tumour-draining lymph nodes.*

Female 6-8 week old Balb/c mice were injected in the fourth mammary fat pad (both flanks) with 1 x 10⁵ tumour cells. Six days later, draining (inguinal and brachial) LN were removed and single cell suspensions were prepared. Cells were incubated overnight with PMA and ionomycin at 37°C. The following day, culture supernatants were harvested and filtered through a 0.2µm filter and analysed by cytokine bead array as described in Section 2.4.12. Samples were acquired within 24 hours on a FACSCanto flow cytometer and analysed using the Bender MedSystems FlowCytomix software. Bars represent mean \pm SEM (*n*=7 mice per group from 2 independent experiments). No significant differences between groups. (A) Concentration of IL-2, IFN- γ , and IL-17 in culture supernatants from cells derived from tumour-DLN. (B) Concentration of TNF- α , GM-CSF and IL-4 in culture supernatants from cells derived from tumour-DLN. (C) Concentration of IL-1 α , IL-5, IL-6 and IL-10 in culture supernatants from cells derived from tumour-DLN.



CHAPTER 6

Discussion



CHAPTER 6: DISCUSSION

6.1 Introduction

The importance of CXCL12 and CXCR4 in cancer has become increasingly apparent over the last few years. However, while CXCR4 expression has been correlated with tumour malignancy and metastatic disease in multiple cancer types^{6, 127, 178-195}, the precise mechanisms by which it promotes cancer progression are still poorly understood. Much attention has been focused on immunotherapy as a strategy for treating cancer, and chemokines, with their ability to modulate leukocyte migration and activation, hold much promise as potential therapeutics. A handful of studies have reported anti-tumour activity of CXCL12 when overexpressed in fibrosarcoma²¹⁵, melanoma, lung carcinoma²¹⁸ and leukaemia²¹⁶, but at the time this research was conducted, the anti-tumour effects of CXCL12 in breast cancer had not been investigated.

The relatively newly identified chemokine CXCL16, which acts as a chemoattractant for activated polarised Type I T cells and other effectors of cell-based immunity, is a possible candidate for enhancing anti-tumour immune responses. Very little has been published regarding the role of this chemokine in cancer, and the few studies that have been recorded have documented differing effects. As yet, no reports about the potential use of CXCL16 as a therapeutic agent in cancer have been published.

For the reasons outlined above, this study was conducted to assess the ability of CXCL12 and CXCL16 and their respective antagonists, CXCL12_(P2G) and CXCL16₍₉₋₂₂₀₎, to influence breast tumour growth and metastasis. A number of significant findings were made, specifically that (1) overexpression of CXCL12 induced a significant anti-tumour immune response that was in part dependent on T cells and the accumulation of CD11c⁺ cells within the tumour-DLN, (2) expression of the antagonist CXCL12_(P2G) did not affect primary tumour growth but inhibited both spontaneous and experimental metastasis, and (3) overexpression of CXCL16 did not affect tumour growth or metastasis, but expression of the CXCL16₍₉₋₂₂₀₎ antagonist inhibited experimental metastasis. Attempts were made to elucidate the mechanisms behind these key observations; however, time constraints prevented detailed characterisation of the mode of action of the chemokines and their antagonists in this model. Therefore, the following discussion will address the major

findings of this work, but also will include speculation as to the mechanistic details of chemokine/antagonist-mediated inhibition of tumour growth and metastasis. In addition, suggestions as to how these mechanistic details might be investigated in future are discussed.

6.2 The effects of CXCL12 overexpression on breast tumour progression

As demonstrated in the preceding chapters, CXCL12 had a profound suppressive effect on 4T1.2 tumour growth *in vivo*. This was despite any apparent effect on 4T1.2 cell growth *in vitro*. Subsequent experiments to elucidate the mechanism of CXCL12-mediated tumour inhibition showed that (1) the inhibitory effect was completely abrogated in SCID mice, partly abrogated in nude mice and intact in Ja18^{-/-} mice, (2) the inhibitory effect was partly dependent on IFN- γ , perforin and TRAIL, and (3) CXCL12 overexpression in the tumour increased the proportion of CD11c⁺ cells, but reduced the proportion of IFN- γ^+ , IL-17⁺ and IL-4⁺ cells in the tumour-DLN.

6.2.1 The effects of CXCL12 on tumour cells and the tumour vasculature

The fact that CXCL12 could inhibit tumour growth *in vivo*, without affecting the *in vitro* growth characteristics of 4T1.2 cells, strongly suggested that CXCL12 was mediating its effects through modulation of host components, rather than acting directly on the tumour. This does not rule out the possibility that CXCL12 is able to regulate tumour cell processes that are required only for growth *in vivo*, and that are dispensable for growth *in vitro*. However, direct stimulation of cells with CXCL12 generally activates functions that would be expected to promote tumour growth, such as secretion of degradative enzymes²⁰⁷ (which could assist the growing tumour to invade surrounding tissue as well as supply growth factors and nutrients freed from the ECM) and activation of integrins²⁹⁰ (which could enhance anchorage of tumour cells). In light of the reduction of tumour growth seen in this model, it seems unlikely that direct action of CXCL12 on tumour cells plays a significant role here.

From reports of the functions of CXCL12 in the literature, two of the major ways in which CXCL12 could influence the host are through induction of tumour angiogenesis^{117, 196, 210}, and through augmentation of the anti-tumour immune response²¹⁵⁻²¹⁸. Generally,

neovascularisation is thought to be a tumour-promoting process, providing oxygen and nutrients to the rapidly growing tumour mass. CXCL12 is a pro-angiogenic chemokine¹¹⁷, and as such, it might be expected that its expression would enhance tumour angiogenesis. However, the data obtained in this study indicated that the vascularity of CXCL12-expressing tumours was reduced compared with control tumours. Thus it seems likely that the decrease in vascular density was not a direct result of overexpressing CXCL12, but rather an indirect consequence of the inhibition of tumour growth. Despite the reduced vascularity of CXCL12-expressing tumours, there was a trend for the haemoglobin concentration within these tumours to be higher. It is possible for vascular permeability to be increased without any change in blood vessel density, as has been reported in a model of lung carcinoma²⁹¹, and it may be that CXCL12 directly affects only this aspect of the vasculature in the 4T1.2 model. However, it seems unlikely that increasing vascular permeability and instability is a mechanism by which CXCL12 inhibits tumour progression as it has been shown that inducing maturation and stabilisation of tumour blood vessels blocks tumour expansion²⁹².

6.2.2 The importance of T cells and dendritic cells

The actions of CXCL12 on the tumour vasculature and on tumour cells directly would be expected to have pro-tumour rather than anti-tumour effects, as discussed above. Thus it seemed most likely that CXCL12 was exerting its anti-tumour function by influencing the immune response to 4T1.2 tumours. For this reason, analysis of the effect of CXCL12 on the anti-tumour immune response was a significant focus of this work. Experiments with strains of mice lacking various leukocyte subsets were carried out to clarify which cell populations could be mediating tumour inhibition by CXCL12. SCID mice have a defective *Prkdc* gene and therefore cannot repair the DNA double strand breaks required for antigen receptor gene rearrangement. These mice lack T, B and NKT cells²⁹³. Nude mice have a mutation in the *Foxn1* gene, which encodes a transcription factor essential for development of the thymus. These mice are deficient in T cells that are dependent on the thymus for their development, though NKT cells and some thymus-independent T cells are still present^{294, 295}.

The complete loss of CXCL12-mediated tumour inhibition in SCID mice indicated that T cells, B cells and/or NKT cells were required for this effect. The experiments in nude

mice confirmed a role for T cells in CXCL12-mediated tumour inhibition. This is in agreement with other studies that demonstrated the importance of T cells in CXCL12-mediated growth inhibition of fibrosarcoma²¹⁵, melanoma, colon carcinoma, lung carcinoma²¹⁸ and leukaemia²¹⁶. Further support for the role of T cells in this model came from correlative analysis in the present study, which indicated that the greater the proportion of CD4⁺ splenic T cells, the smaller the tumour, particularly when the tumours expressed CXCL12.

Another significant finding of this work was that CXCL12 expression by 4T1.2 tumours resulted in increased numbers of CD11c⁺ cells in the tumour-DLN. CD11c is the α_x chain of the α_x - β_2 integrin, and is expressed predominantly on conventional DC. It can also be expressed on activated T cells²⁹⁶. It is plausible that the CD11c detected by immunohistochemistry in the DLN was expressed by cells from either or both populations. However, no obvious differences in CD11c⁺ cell infiltration of tumours from the different groups was apparent. In an effective immune response, DC activated at the tumour site mature and migrate to the tumour-DLN. There, they prime naïve T cells, which become activated and migrate back to the tumour site to mediate their effector functions^{297, 298}. Given the increase of CD11c⁺ cells within the DLN, but not the tumours, it seems likely that these cells represent DC rather than activated T cells. This is in agreement with a previous report that CXCL12 overexpression in colon carcinoma increased the numbers of CD11c⁺ DC in the tumour-DLN²¹⁸.

Based on the observations discussed above, it seems likely that CXCL12 acts to promote T cell activation by increasing the proportion of DC in the tumour-DLN. There, DC can prime naïve T cells which then exert anti-tumour effector functions. One way in which CXCL12 may promote migration of DC to the tumour-DLN is through induction of their maturation. CXCL12 may enhance infiltration of the tumour by innate inflammatory cells, which could then provide the requisite 'danger signals' to induce maturation of immunostimulatory DC. Interestingly, responsiveness to CXCL12 increases during the maturation process of DC¹⁵⁰, and a very recent publication has documented the role of CXCR4 in directly promoting survival of DC, particularly those of a mature phenotype²⁹⁹. The ability of CXCL12 to enhance the survival of mature DC may be critical to the

generation of sufficient mature DC for priming of anti-tumour T cells and may represent a key aspect of the anti-tumour response in this model.

Upon maturation, DC up-regulate CCR7 expression³⁰⁰, increasing their responsiveness to CCL21 and CCL19 expressed in lymphatic vessels and LN and facilitating their migration to LN^{301, 302}. Though the chemotactic responsiveness of mature DC to CXCL12 is also increased¹⁵⁰, it may be that signals from CCL21 and CCL19 are sufficient to overcome CXCL12 signalling, or the high levels of CXCL12 within the tumour desensitise CXCR4 so that DC are free to respond to chemotactic signalling by CCL21 and CCL19. Alternatively, since chemokines are readily transported through the lymphatics and accumulate in DLN³⁰³, it is possible that CXCL12 expressed by the tumour drains into surrounding LN. Together with other LN chemokines, CXCL12 may facilitate directly the migration of activated DC into the DLN to initiate the anti-tumour immune response. CXCL12 filtering into the tumour-DLN may also serve an additional function. Costimulation of T cells by CXCL12 enhances activation of cell signalling pathways commonly associated with cell survival and promotes secretion of $IL-2^{81}$. It may be that in this model, CXCL12 enhances survival and proliferation of T cells receiving signals from incoming DC in the DLN, thus facilitating their optimum activation for anti-tumour activity.

A logical next step would be to further narrow down the T cell subsets important for the CXCL12-mediated anti-tumour response. To this end, experiments were conducted in which growth of control and CXCL12-expressing tumours was monitored in mice depleted of CD4⁺ and CD8⁺ lymphocytes using specific antibodies. Unfortunately, the preparation of anti-CD8 antibody did not efficiently deplete CD8⁺ cells, and time constraints prevented the production of sufficient fresh antibody for further experiments. The data from mice treated with anti-CD4 antibody, however, indicated that CD4⁺ cells were not required for CXCL12-mediated tumour inhibition. Growth of CXCL12-expressing tumours was inhibited to the same extent in mice depleted of CD4⁺ cells as in mice treated with control IgG (data not shown). These findings strongly suggest that CD8⁺ T cells are the mediators responsible for the anti-tumour effects of CXCL12, and it is expected that this will be confirmed in future experiments with an effective anti-CD8 depleting antibody.

If CD8⁺ T cells are confirmed to be the main effector population in this model, then it is most likely that they are activated by direct cross-presentation of tumour cell antigens by mature DC in the tumour-DLN. This hypothesis fits with the observed increase of $CD11c^+$ cells in tumour-DLN and the finding that CD4⁺ T cells are dispensable for the CXCL12induced anti-tumour response. DC that acquire tumour antigens at the tumour site are able to process these molecules and present them on MHC class I molecules to CD8⁺ T cells. It has been shown that cross-presentation of tumour antigens to CD8⁺ T cells can occur efficiently in the absence of CD4⁺ T cell help³⁰⁴. However, in mice lacking tumour-DLN due to ablation of peripheral LN, CD4⁺ T cell help was crucial for efficient CD8⁺ T cell activation³⁰⁴. It was postulated by the authors that CD4⁺ T cell help became necessary only when antigens were limiting or DC were unable to interact efficiently with CD8⁺ T cells, such as may occur at sites outside of the DLN. This could explain why CD4⁺ T cells do not seem to be required for the CXCL12-induced anti-tumour response described in this thesis, as DLN are intact in this model. It also provides an explanation for the correlation of reduced tumour weight with an increased proportion of splenic CD4⁺ cells. Yu et al. showed that while dispensable for priming in the DLN, CD4⁺ T cell help was required for efficient priming of $CD8^+$ T cells in the spleen. It is possible that in the 4T1.2 model, $CD4^+$ cell-dependent priming of effector cells occurs in the spleen in response to tumour CXCL12 expression, but the predominant site of activation is the DLN.

6.2.3 The role of cell-mediated cytotoxicity

Despite the evidence for the involvement of T cells in CXCL12-induced tumour inhibition, increased infiltration of CXCL12-expressing tumours with either CD4⁺ or CD8⁺ cells was not apparent. It has been reported that CXCR4 is down-regulated on activated T cells and that this correlates with a decreased migratory response to CXCL12^{156, 157}. However, highly activated CD4⁺ T cells, which become hypersensitive to apoptotic signals, were found to be partially protected from apoptosis by CXCL12³⁰⁵. Thus expression of CXCL12 within the tumour microenvironment, while not enhancing T cell infiltration, may in fact enhance survival of activated T cells within the tumour mass.

Another potential explanation for the enhanced immune response to CXCL12-expressing tumours, despite unchanged T cell infiltration, is that T cells entering the tumour were in an enhanced activation state. However, flow cytometric analysis of CD4⁺ and CD8⁺ cells

isolated from tumours did not reveal any significant differences in the activation marker profile between tumour groups. As an alternative to analysis of phenotypic markers of T cell activation, functional assessment of the activation state of T cells within the different tumour groups was employed. In vitro cytotoxicity assays suggested that the cytotoxic activity towards 4T1.2 targets of LN cells from 4TX121 tumour-bearing mice was enhanced compared with LN cells from control tumour-bearing mice. The cytotoxic activity observed was relatively low (around 20-30%), in keeping with previous reports that 4T1 tumours (from which 4T1.2 cells are derived²⁷⁶) are very poorly immunogenic²⁸³. It should be considered also that the source of effector cells (i.e. the secondary lymphoid organs) may have yielded cells that were not yet fully activated for cytotoxic function; activated lymphocytes may not realise their full cytotoxic capabilities until they reach the tumour site. Another point for consideration is that the method of obtaining effector cells (i.e. mechanical disruption of tissues) favours the isolation of non-adherent cells, such as lymphocytes, rather than more adherent cells, such as macrophages. It is possible that the main function of T effector cells is not direct cytotoxic activity against tumour cells, but stimulation of the cytotoxic activity of innate immune cells, such as macrophages and NK cells, through production of cytokines such as IFN- γ . Indeed, it was found in the current study that CXCL12-mediated inhibition of tumour growth was partially lost in IFN-γ-deficient mice (discussed further below). Interestingly, a recent study demonstrated that IFN- γ required for tumour immunity can be produced by innate immune cells, but only if they are provided with help, most likely mediated by IL-2, from T cells³⁰⁶. Such a scenario may be the case in the 4T1.2 model studied here. CD8⁺ Tc1 cells would be the most likely source of IL- 2^{307} , since depletion of CD4⁺ cells in the present work did not seem to affect CXCL12-mediated tumour growth inhibition (discussed in Section 6.2.2 above). If innate immune cells are the direct effectors of cytotoxicity in this model, they are more likely to be found in the tumour than in the secondary lymphoid organs. Thus the true extent of the cytotoxic response to 4T1.2 tumours might be better demonstrated if effector cells collected from the tumour, rather than the LN and spleen, were to be used in the *in vitro* assay.

Mice deficient in common mediators of $CD8^+$ T cell cytotoxic function were used to provide insight into the mechanism by which T cells were exerting their effects. The inhibitory effect of CXCL12 was partially lost in both perforin-, and TRAIL-deficient

mice. This suggests that both the perforin/granzyme cytolytic pathway and the TRAILdependent pathway are important for the CTL response against 4TX121 tumours and provides further evidence for the role of CD8⁺ T cells in this model. However, it is important to note that both perforin³⁰⁸ and TRAIL³⁰⁹ are important for the cytolytic properties of NK cells. NK cells also express CXCR4 and migrate in response to CXCL12¹⁵³, and may be preferentially recruited to tumours where they can be activated by IFN- γ -producing cells to exert their cytotoxic functions. It has been shown previously that the cytotoxicity of liver and spleen mononuclear cells towards 4T1 cells in vitro is due solely to the activity of NK cells³¹⁰. Moreover, that cytotoxic activity was dependent on both perforin and TRAIL. Comparison of the growth of 4T1 tumours in wild-type and TRAIL^{-/-} mice in that study revealed that tumour growth was enhanced in the absence of TRAIL. Importantly, however, enhanced 4T1 tumour growth in TRAIL^{-/-} mice was only observed when the tumour burden was small (i.e. an inoculum size of 1×10^4 cells or less), presumably because at larger tumour cell doses, the NK cell response was overwhelmed. In the experiments conducted in the present study, a tumour cell inoculum size of 1×10^5 cells was used, ten times greater than that used when TRAIL deficiency was observed to have an effect on 4T1 tumour growth. Interestingly, in mice inoculated with $2.5 \times 10^4 \text{ 4T1}$ cells and treated with α -galactosylceramide, tumour growth was inhibited in the presence of TRAIL³¹⁰. This indicates that further activation of NK cells can be induced to effectively overcome larger tumour cell doses. Since IFN- γ is another factor that can enhance the cytotoxic activity of NK cells³⁴, and since experiments in IFN- γ -deficient mice in the present study revealed that the anti-tumour effect of CXCL12 is partly dependent on IFN- γ , it is possible that NK cells augment the cytotoxicity of CTL towards 4TX12 1 tumours observed here. It is also interesting to note that NK-like cytolytic functions have been attributed to DC, and that these cells use TRAIL to kill their targets^{311, 312}. DC also produce large amounts of IFN- γ . Following activation and initial cytotoxic activity, these 'interferon-producing killer dendritic cells (IKDC)' diminish their cytolytic capacity and up-regulate antigen presentation functions³¹³. An intriguing possibility is that the DC accumulating in the DLN in this model might have been derived from such an IKDC population. This could be further investigated by flow cytometric analysis of CD11c⁺ cells isolated from the DLN to determine if these cells express the characteristic CD11c^{int}B220⁺Gr1⁻CD49b⁺ IKDC surface marker phenotype.

6.2.4 The role of cytokines

Clearly, the production of IFN- γ is important for the inhibition of 4T1.2 tumour growth mediated by CXCL12, as demonstrated by the experiments in IFN- γ -deficient mice. IFN- γ production is a major function of effector $CD8^+$ T cells³¹⁴, which suggests that T cells in the 4T1.2 model may exert not only direct cytotoxic function against tumour cells, but also indirect effects via IFN-y-mediated activation of other effector cells. Following the observation that CXCL12-induced tumour inhibition was partly lost in IFN- $\gamma^{-/-}$ mice, it was somewhat surprising to find that the proportion of IFN- γ^+ cells within 4TX12 1 tumour-DLN was reduced compared with control tumour-DLN. Indeed, the percentage of cytokine-positive cells in general within 4TX12 1 tumour-DLN tended to be reduced compared with that in controls. Furthermore, no significant difference in the amount of cytokine secreted by LN cells from either tumour group was apparent. Again, however, it should be noted that only cytokine-secreting cells from DLN were assessed, and the proportion of cytokine-secreting cells within the tumours themselves may be quite different. This being said, it would be expected that a reduction in the proportion of IL- 17^+ and IL-4⁺ cells within the DLN would reflect enhanced anti-tumour immunity. IL-23 (and by extension IL-17) is thought to be important for the promotion of tumour progression²⁸⁷, presumably through induction of inflammation, which can be pro-tumorigenic³¹⁵. IL-17 also inhibits the generation of a Th1-type immune response, which is thought to be the most effective anti-tumour response. Similarly, IL-4 inhibits Th1 responses and instead promotes Th2 humoral immunity, which by itself is generally ineffective against most tumours. Thus a reduction in the numbers of cells positive for these cytokines in mice bearing 4TX12 1 tumours would not be surprising. On the other hand, the reduction of IFN- γ^+ cells in 4TX12 1 DLN is perhaps unexpected. However, if the majority of effector cells produce IFN- γ and migrate from the DLN back to the tumour, then a reduction in IFN- γ^+ cells in the DLN is not so surprising. An increase in IFN- γ^+ cells in DLN from 4TX12 1 tumour-bearing mice might be observed if the LN are taken at earlier time points, prior to the egress of effector cells. In addition, it should be considered that not only effector CD8⁺ T cells, but also innate immune cells present in the tumour, could be sources of IFN- γ . Analysis of cytokine levels in the tumour or systemic levels in the blood may give a better indication of the contribution of IFN- γ , and indeed other cytokines, to the immune response against 4TX12 1 tumours.

6.2.5 The potential effects of CXCL12 on suppressor cells

Another finding from the analysis in this thesis was that increased total splenocytes correlated with tumour weight. It has been shown previously that mice bearing 4T1 tumours have elevated levels of splenic MSC, which are potent inhibitors of CD4⁺ and $CD8^+$ T cell effector functions³¹⁶. In that study it was found that in Stat6^{-/-} mice, which are unable to transduce signals induced by IL-4, reduction of MSC and resistance to metastatic disease occurred following removal of the primary tumour. Consistent with observations in the present study, CD8⁺, but not CD4⁺, cells were required for this effect, which was also IFN- γ -dependent³¹⁶. Low doses of IFN- γ combined with TNF- α were found to reduce MSC numbers in a model of lung cancer by forcing their differentiation into mature macrophages³¹⁷. These data may have implications for the 4T1.2 model described in this thesis. Here, IFN- γ was found to be required in part for the CXCL12-mediated inhibition of tumour growth, and secretion of TNF- α by DLN cells was elevated in mice bearing CXCL12-expressing tumours. CXCL12 stimulation can induce TNF- α production by ovarian cancer cells³¹⁸ and astrocytes^{319, 320}, and it is possible that CXCL12 can induce TNF- α production by cells within 4T1.2 tumours and DLN in a similar manner. Thus reduction in 4TX12 1 tumour growth may be the result of CXCL12-induced IFN- γ and TNF- α forcing maturation of MSC and abrogating their immunosuppressive effects.

Another observation from the studies of Sinha et al. was that cytotoxic nitric oxide (NO)producing M1 macrophages were required for resistance to metastatic disease³¹⁶. Macrophage infiltration of 4TX12 1 tumours and 4T12Ala tumours was examined by immunohistochemical staining of tumour sections for the macrophage marker F4/80, but no differences in macrophage infiltration between tumour groups were evident (data not shown). However, the phenotype of tumour-associated macrophages was not assessed, and it may be that there are differences in the bias of cytotoxic NO-producing M1 macrophages versus immunosuppressive arginase-producing M2 macrophages in 4TX12 1 and 4T12Ala tumours. Differentiation of macrophages towards an M1 phenotype is thought to be dependent on IFN- γ , as STAT6^{-/-} mice produce M1 macrophages, whereas wild-type IL-4responsive Balb/c mice produce M2 macrophages³²¹. Thus IFN- γ production in the 4T1.2 model may have a dual role in the anti-tumour response by reducing MSC numbers and inducing their maturation towards a cytotoxic M1 macrophage phenotype. It is also interesting to note that gp120 stimulation of CXCR4-expressing macrophages induces their
activation and local inflammation³²². Whether CXCL12 exerts a similar function in the context of the 4T1.2 tumour, to induce macrophage activation and cytotoxic function directly, remains to be determined.

Aside from MSC, Treg cells represent another immunosuppressive cell type that has the potential to inhibit anti-tumour immunity. Elevated levels of T_{reg} cells have been documented in the peripheral blood, tumours and regional LN of breast cancer patients⁴⁶ and depletion of these cells enhances anti-tumour immunity in mouse models of the disease^{47, 48}. Previous work has shown that 4T1 tumours produce large amounts of TGF- β^{42} , a major inducer of regulatory T cell (T_{reg}) differentiation³⁸. As yet, however, there have been no studies published documenting increased T_{reg} cell numbers in 4T1 tumours. Two reports that Treg migrate in response to CXCL12 have been published^{323, 324}, but it seems unlikely that CXCL12 overexpression in 4T1.2 tumours is acting in this capacity in the present work. If this were the case, it would be expected that tumour growth would be enhanced, rather than inhibited as observed here. Moreover in the current study, tumour growth in mice depleted of CD4⁺ cells (including, presumably, CD4⁺CD25⁺ T_{reg} cells) was not inhibited compared with tumour growth in mock-depleted mice (data not shown). This suggests that T_{reg} cells are not a major contributor to tumour-induced immunosuppression in this model.

6.2.6 Other potential mediators of the effects of CXCL12

As discussed above, the experiments in SCID and nude mice confirmed a role for T cells in CXCL12-mediated anti-tumour immunity. The complexity of the response to the tumour became apparent following the experiments in nude mice, which revealed that cell types other than T cells must be involved as well, because some growth retardation of CXCL12-expressing tumours was still observed. NKT cells and B cells, being present in nude mice but not in SCID mice, were potential co-mediators of tumour inhibition. Much attention has focused on the potential anti-tumour activity of NKT cells, in particular those expressing V α 14-J α 18 and responding to α -galactosylceramide³²⁵. J α 18^{-/-} mice lack the J α 18 joining segment of the TCR α chain and are unable to produce Type I NKT cells, which require expression of the invariant V α 14-J α 18 TCR α chain for their development^{282, 326}. However, in the current study, the growth of CXCL12-expressing

4T1.2 cells in J α 18^{-/-} mice was similarly inhibited as in wild-type mice, indicating that deficiency of Type I NKT cells did not affect the CXCL12-mediated suppression of tumour growth. It is possible that Type II NKT cells, which are still present in J α 18^{-/-} mice, could mediate the anti-tumour effects of CXCL12 to some extent. However, recent work suggests that this subtype of NKT cells exerts an immunosuppressive effect that promotes tumour growth³²⁷. Thus it seems unlikely that NKT cells are the co-mediators of tumour inhibition observed here.

The role of B cells in this model must also be considered. The work of recent years has highlighted the importance of cell-mediated immunity in the eradication of tumours, and the potential of humoral immunity to contribute to the anti-cancer response has slipped from notice somewhat. However, a few studies have shown that immunotherapy of patients can induce effector cell populations that can mediate antibody-dependent cell-mediated cytotoxicity (ADCC)³²⁸⁻³³¹. Interestingly, plasma cells become increasingly sensitive to the chemoattractive properties of CXCL12 as they mature³³², a process that facilitates their migration into the bone marrow. It is possible that in the 4T1.2 model, CXCL12 overexpression by tumour cells may recruit mature plasma cells producing tumour-specific antibodies into the tumour itself, thus enhancing the antibody supply for ADCC at the tumour site. However, the absence of a requirement for CD4⁺ T cells in the anti-tumour immunity towards 4TX121 tumours has implications for the potential for B cells to contribute to this response. Since efficient class-switching of antibodies requires the help of CD4⁺ T cells, it would be expected that only anti-tumour antibodies of the IgM isotype would be produced in this system. A murine $Fc\alpha/\mu$ receptor has been identified on macrophages and B cells³³³, but the ability of this receptor to contribute to ADCC has not been determined. IgM is a potent activator of complement, though whether 4T1.2 tumour cells are sensitive to complement-mediated destruction has not been determined.

Another possibility to be considered is that CXCL12-induced anti-tumour functions of NKT or B cells only come to the fore in the absence of T cells. When T cells are present, they may be sufficient to mediate CXCL12-induced tumour growth inhibition, and the role of NKT and/or B cells becomes redundant. When T cells are lacking, as is the case in nude mice, then the anti-tumour effects of NKT and/or B cells then become evident. A role for

these cell subsets in the anti-tumour response may be revealed only upon concomitant depletion of T cells and NKT cells or of T cells and B cells.

Finally, it should not be forgotten that the nude mutation exhibits some 'leakiness'. Populations of T cells that develop independently of the thymus have been identified in the intestine, appendix, liver and epidermis³³⁴⁻³³⁹. Moreover, it has been shown that intraepithelial $\gamma\delta$ T cells are important for protection of mice from skin carcinogenesis³⁴⁰. It is thought that such cells may play a role in immunosurveillance of epithelial tissues, eliminating newly transformed cells and preventing tumour development²⁴. Whether such cells can contribute significantly to the anti-tumour response once tumours have become well established, as in this model, remains uncertain.

6.2.7 Summary and future directions

Work still needs to be done to clarify the precise components and mechanisms of the antitumour immune response induced by CXCL12. However, the data obtained thus far have given rise to the working model of the anti-tumour response shown in Figure 6.1. The model incorporates the major findings of this study, as well as some speculation on processes that remain to be elucidated. The role of mature DC and T cells, IFN- γ and perforin- and TRAIL-dependent cytotoxicity (as mediated by T cells and NK cells) is presented. In addition, the model emphasises the importance of the contribution of innate immune cells to the anti-tumour response, particularly NK cells and macrophages. The potential roles of B cells, antibodies and thymus-independent T cells in the immune response are not included because as yet, no direct evidence for their participation has been found. Similarly, anti-tumour responses generated in the spleen have not been included as they may not follow the same pattern as those occurring in the DLN.

As well as providing the basis for the model of CXCL12 action in Figure 6.1, the findings of this study raise interesting questions regarding additional details of the anti-tumour response induced by CXCL12. Confirmation of the role of CD8⁺ T cells should be the next step in the investigation of T cell involvement. Examination of the effector functions of T cells isolated from tumours will clarify whether these cells exert anti-tumour activity through direct cytotoxicity, IFN- γ production or both. It will be interesting also to further

characterise the DC infiltrating tumour-DLN. Flow cytometry could be employed for analysis of phenotypic markers of DC activation, such as MHC class II, CD80, CD86, CD83 and CCR7. Assessment of the antigen-presenting function of these DC could prove informative as well. This could be achieved by testing the ability of DC isolated from DLN of 4T12Ala or 4TX12 1 tumour-bearing mice to induce proliferation of anti-tumour T cells isolated from 4T1.2 tumour-bearing mice *in vitro*.

Future experiments involving anti-asialoGM1 depletion of NK cells and liposomemediated depletion of macrophages³⁴¹⁻³⁴³ will provide further insight into the role of innate immune cells in anti-tumour immunity. Quantitation of NO within tumours will also give an indication of the importance of macrophage bias (i.e. NO-producing M1 versus arginase-producing M2 macrophages) to CXCL12-mediated tumour inhibition. Another important aspect of this model to be investigated is the contribution of MSC to the growth of 4T1.2 tumours, and what effect CXCL12 overexpression has on this cell population. Thus cell population(s) in the peripheral blood and infiltrating the spleens of tumourbearing mice should be characterised to determine if these cells are Gr1⁺CD11b⁺ MSC. Comparison of the numbers of MSC in 4T12Ala versus 4TX12 1 tumour-bearing mice would give an indication of whether or not reduction of MSC levels are important for the anti-tumour effect of CXCL12.

In summary, multiple significant effects of CXCL12 overexpression on 4T1.2 tumour growth have been demonstrated in this work. CXCL12 induces a potent anti-tumour immune response, an effect not previously observed in breast cancer. This immune response is dependent on T cells, increased infiltration of tumour-DLN by DC, the cytotoxic effectors perforin and TRAIL and the cytokine IFN- γ . The findings of this study also point to the involvement of additional cell types in the CXCL12-induced anti-tumour response and open up exciting new avenues for future investigation. What is clear from the data obtained thus far is the complexity of the anti-tumour immune response in this model. The ability of CXCL12 to mobilise so many components of the immune system testifies to its potential as an immunotherapeutic agent.

6.3 The effects of $\text{CXCL12}_{(\text{P2G})}$ overexpression on breast tumour progression

Although the present study demonstrates that overexpression of CXCL12 enhances the anti-tumour immune response in a model of breast cancer, reports in the literature indicate that endogenous CXCL12 can act in a pro-tumorigenic capacity. *In vitro* and *in vivo* studies suggest that CXCL12 can enhance the survival²⁰², proliferation¹⁹⁶⁻¹⁹⁹ and invasion^{197, 204, 205, 208, 209} of breast tumour cells and promote the vascularisation of breast tumours^{196, 210}. In addition, blockade of CXCR4 using a neutralising antibody has been shown to inhibit human breast tumour cell metastasis in a SCID mouse model⁶. More recent work has demonstrated that the novel CXCL12 receptor CXCR7 also has pro-tumorigenic functions¹²⁹, though these effects are yet to be described in breast cancer *in vivo*. Despite the data implicating CXCL12 and its receptors in the progression of breast cancer, the details of the mechanism by which they do so remain elusive. Many of the tumour-promoting functions of CXCL12 have been conjectured from the results of *in vitro* studies, and there is very little direct evidence for CXCL12 and its receptors enhancing specific aspects of breast tumorigenesis and metastasis *in vivo*.

In this study, expression of the CXCL12 antagonist, $CXCL12_{(P2G)}$, did not have any obvious effect on the growth of primary 4T1.2 tumours. However, spontaneous metastasis of 4T1.2 cells from the primary tumour to the lungs was significantly inhibited by $CXCL12_{(P2G)}$, in a manner that did not depend on reduced tumour angiogenesis or lymphangiogenesis. Moreover, $CXCL12_{(P2G)}$ was capable of inhibiting metastasis to the lungs even when tumour cells were injected directly into the circulation, obviating the need for cells to form a primary tumour and then escape from the tumour mass. These data highlight the role of CXCL12 in metastatic steps occurring following the entrance of tumour cells into the circulation, though a role for CXCL12 in earlier metastatic steps cannot be ruled out.

Several studies have reported the inhibition of metastasis of human MDA-MB-231 breast cancer cells in SCID mice using anti-CXCR4 antibodies^{6, 344}, RNAi^{197, 205}, or peptide antagonists of CXCR4^{344, 345}. The data presented in this thesis extend these findings by demonstrating that blockade of CXCR4 inhibits both spontaneous and experimental metastasis in a more biologically relevant syngeneic, orthotopic model of breast cancer.

The results obtained in the 4T1.2 model are in agreement with those obtained in a similar study by Smith et al.³⁴⁶ in which RNAi-mediated knockdown of CXCR4 inhibited both spontaneous and experimental breast cancer metastasis. However, in that study, CXCR4 knockdown was also found to inhibit primary tumour growth. In contrast, expression of CXCL12_(P2G) in the present work did not affect 4T1.2 tumour growth, indicating the metastasis-specific effect of this antagonist and confirming a role for CXCL12 in metastasis.

While the vascular density of primary 4T12P2G tumours was not reduced compared with control tumours, the potential for CXCL12_(P2G) to inhibit metastasis by influencing the tumour vasculature should not be discounted. Criscuoli and colleagues found that in Src^{-/-} mice, reduction of spontaneous metastasis was the result of decreased vascular permeability, though microvessel density was unchanged²⁹¹. The permeability of tumour blood vessels was not assessed in the 4T1.2 model, but represents a possible future avenue of investigation. The extravasation of intravenously injected FITC-dextran into tumours has been used to quantify tumour vascular permeability, and may reveal reduced vascular permeability in 4T12P2G tumours. With regard to tumour lymphangiogenesis, it should be noted that only the density of intratumoral lymphatic vessels was assessed in this study. It has been shown that the presence of lymphatic vessels in the tumour margin (defined as <100µm from the tumour edge) may be a more accurate predictive factor for lymphatic metastasis³⁴⁷. Examination of the effects of CXCL12_(P2G) on lymphangiogenesis within this region may give a better indication of whether or not CXCL12_(P2G) affects the lymphatic route of metastasis.

An essential feature of metastatic cells is the ability to evade anoikis, a form of apoptosis induced in the absence of anchorage-dependent survival signals^{20, 348}. Data from our laboratory have demonstrated a role for CXCL12 in preventing anoikis of detached breast cancer cells (M. Kotchetkova and S. McColl, manuscript in preparation). Thus $CXCL12_{(P2G)}$ expression in the 4T1.2 model may serve to block the survival of metastatic cells in the circulation. Given that 4T1.2 cells do not express CXCL12, it seems most likely that survival signals would come from host-derived plasma CXCL12, the basal levels of which have been reported to be 2-3ng/ml in Balb/c mice³⁴⁹. Further investigation of the potential of CXCL12_(P2G) to block 4T1.2 anoikis would require confirmation of the

ability of CXCL12 to enhance survival of detached 4T1.2 cells *in vitro*, followed by experiments to determine if $CXCL12_{(P2G)}$ could block this effect.

Another function of CXCL12 that may be pertinent to the metastasis of malignant cells is its ability to induce adhesion to endothelium and ECM components via activation of integrins. Various studies have demonstrated the activation of the integrin very late antigen (VLA)-4 on leukocytes^{350, 351}, haematopoietic progenitor cells³⁵² and multiple myeloma cells³⁵³ by CXCL12. The increased affinity of VLA-4 promotes firm adhesion of cells to EC expressing vascular cell adhesion molecule (VCAM)-1 and ECM components such as fibronectin. Such interactions in the context of 4T1.2 metastasis may facilitate arrest and adhesion of metastatic cells at secondary sites followed by extravasation into the underlying tissue. It is interesting to note that MDA-MB-435 cells (recently redefined as melanoma-derived cells²⁰³) transfected with a CXCR7-expressing DNA construct exhibited enhanced adhesion to EC¹²⁹. The ability of CXCL12_(P2G) to bind and antagonise the function of CXCR7 has not been determined, but it is possible that CXCL12_(P2G) inhibits tumour cell adhesion to endothelium through interaction with this receptor.

In a study of genes that mediate breast cancer metastasis to the lungs, MMP2 was identified as a gene that was expressed in highly aggressive lung metastatic cells but was not required for primary breast tumour growth³⁵⁴. Interestingly, CXCL12 has been shown to induce MMP2 production by prostate cancer cells³⁵⁵, rhabdomyosarcoma cells¹⁹³ and human CD34⁺ haematopoietic progenitor cells³⁵⁶. Thus it is possible that CXCL12_(P2G) expression by 4T1.2 cells blocks production of MMP2 and hinders their ability to invade lung tissue. Some preliminary experiments were conducted to compare the level of MMP2 mRNA expression in 4T12P2G and 4T12Ala cells by real time RT-PCR, but no significant differences were detected between the cell lines. However, MMP2 needs to be post-translationally cleaved to become activated and exert its proteolytic function³⁵⁷. Therefore, functional assays such as zymography to detect MMP2 activity may provide a more accurate picture of the role of the enzyme in this system and uncover differences between the control and CXCL12_(P2G)-expressing cell lines.

Of course, the defining function of CXCL12, as a chemotactic factor, should not be forgotten. Chemokines are major inducers of cell migration into and within tissues. CXCL12 is highly expressed in the lung⁶, potentially creating a chemotactic gradient to

attract metastatic tumour cells out of the circulation and into the tissue. Antagonism of CXCL12 by $CXCL12_{(P2G)}$ expression may neutralise the ability of tumour cells to sense and respond to the CXCL12 gradient within the lung tissue, preventing their establishment in a growth-conducive environment. Given that $CXCL12_{(P2G)}$ had no effect on primary tumour growth, it seems unlikely that CXCL12 promotes tumour growth once metastatic cells are established in the lungs. Still, it is possible that tumour cells encounter different physiological conditions in this tissue and require CXCL12 to proliferate and survive in their new environment, as has been shown for colon carcinoma metastases³⁵⁸.

In summary, the data presented in this thesis show that antagonism of CXCL12 specifically inhibits spontaneous metastasis in a syngeneic, orthotopic model of breast cancer, without affecting primary tumour growth. These data confirm and extend previous findings that blockade or knockdown of CXCR4 inhibits metastasis of human breast cancer cells in xenogeneic mouse models^{6, 197, 205, 344, 345}. In addition, the present work demonstrates that CXCL12 is important for metastatic steps that take place following entry of malignant cells into the circulation, as CXCL12(P2G) was able to inhibit metastasis of intravenously injected tumour cells. A working model of the action of CXCL12_(P2G) during metastasis is presented in Figure 6.2. The potential role for endogenous CXCL12 signalling to enhance anchorage-independent survival in the circulation and/or arrest, adhesion and extravasation at secondary sites is highlighted. The potential points at which CXCL12_(P2G) could block metastasis are also shown. It is possible that CXCL12 contributes to escape of metastatic cells from the primary tumour, and/or assists survival and expansion of tumour cells in the lung environment. However, the experimental evidence obtained thus far has only confirmed a role for CXCL12 in metastatic steps post tumour escape, and the lack of effect of CXCL12 on primary tumour growth suggests that it is unlikely to be required for growth of secondary tumours. It should also be noted that inhibition of metastasis by CXCL12_(P2G) could be the result of interaction with CXCR4 or CXCR7 expressed on the tumour cell surface. This raises interesting questions as to the contribution of each receptor to the metastatic process, and presents a future avenue of research.

6.4 The effects of CXCL16 construct overexpression on breast tumour progression

Some of the physiological functions of the atypical chemokine CXCL16 have relevance to the pathology of breast tumour development. CXCL16 is expressed by vascular EC^{222, 226} and at low levels by aortic smooth muscle cells^{222, 225}, and promotes proliferation, chemotaxis and tube formation by HUVEC²⁴⁹. Thus it might be expected that CXCL16 facilitates vascularisation of 4T1.2 tumours *in vivo*. Furthermore, CXCL16 can act as a scavenger receptor of oxidised low density lipoprotein (oxLDL)^{234, 250}. LDL has been shown to enhance proliferation of various cancer cells, including breast cancer cells^{359, 360}. Thus it is possible that CXCL16 expression by tumour cells could serve to promote their uptake of LDL, and hence their proliferation, *in vivo*. On the other hand, CXCR6 is expressed on polarised Type 1 T cells²³¹ and NK cells²²⁹, therefore it might be expected that CXCL16 would enhance recruitment of such cells to the tumour to mediate cytotoxic activity.

The initial in vitro characterisation of the 4T1.2 cell line revealed a low level of endogenous CXCL16 expression, suggesting that production of this chemokine might be required for the growth of 4T1.2 tumours. Overexpression of CXCL16 or the CXCL16 antagonist, however, did not affect primary tumour growth, indicating that endogenous CXCL16 expression is dispensable for this aspect of 4T1.2 tumour development. It might have been expected that expression of CXCL16 would enhance the immune response to 4T1.2 tumours, but clearly, CXCL16 did not exert such a function in this case. It may be that simply increasing CXCL16 levels in the tumour would be sufficient to produce an effective anti-tumour response, but other complicating factors could be in play. CXCR6⁺ cells may be recruited, but their anti-tumour activity may be suppressed at the tumour site. CXCR6 expression on T cells is induced only once they have been activated²³¹, so if T cells are not being primed, then CXCL16 cannot recruit them to the tumour. CXCL16 may still be able to recruit CXCR6⁺ NK cells, but as discussed above, the large dose of tumour cells used in the *in vivo* experiments $(1 \times 10^5 \text{ cells})$ may be such as to overwhelm any anti-tumour activity of these cells. It is possible that inhibition of CXCL16-expressing tumour growth would occur when a smaller inoculum of tumour cells (e.g. 5×10^3 cells) is used.

Curiously, CXCL16₍₉₋₂₂₀₎ expression was found to significantly reduce metastasis of intravenously injected tumour cells to the lungs. This was despite a lack of effect on spontaneous metastasis of cells from primary tumours. Since intravenously injected cells have to complete the same metastatic steps as spontaneously metastatic cells that have reached the circulation, an explanation for the different outcomes of the two experiments is not immediately apparent. However, there is potential for the quandary to be resolved if the host immune response is taken into account. It has been reported that while T cells are important for the control of tumours and metastases established within tissues, metastatic cells in transit are targeted by NK cells^{361, 362}. Moreover, MSC inhibit the cytotoxic activity of NK cells⁵⁹. In light of this, it should be considered that the timeframe of the experimental metastasis assay is such that there is insufficient time for MSC to accumulate before NK cells are able recognise and destroy tumour cells arrested in the vasculature. This in contrast to the spontaneous metastasis setting where MSC accumulate and begin to suppress NK cell function during the initial phase of primary tumour growth, before metastatic cells are released into the circulation. As discussed above, the size of the tumour cell inoculum in the spontaneous metastasis assay was such as to overwhelm NK cell activity, hence the apparent lack of effect of CXCL16 and CXCL16₍₉₋₂₂₀₎ in that setting. However, a significantly smaller inoculum of tumour cells was used in the experimental metastasis assay. Moreover, intravenously injected cells would become dispersed by the circulation, and these isolated cells would be more readily targeted by NK cells. This is in contrast to cells injected into the mammary fat pad, which would remain as a bolus and thus have the protection of neighbouring tumour cells, making it easier for them to overwhelm the local NK cell response.

Since NK cells express CXCR6²²⁹, it might be expected that CXCL16-expressing tumour cells would attract NK cells, effectively bringing about their own doom. Thus it seems paradoxical that 4T1.2 cells express endogenous CXCL16 and furthermore, that CXCL16₍₉₋₂₂₀₎ expression renders intravenously injected cells more sensitive to destruction. However, *in vitro* cytotoxicity assays performed in this laboratory have shown that 4T1.2 target cells overexpressing CXCL16 are protected to some extent from cytolytic attack (O. Fahy, unpublished data). Thus expression of the antagonist may block the protective effects of endogenously expressed CXCL16 *in vivo*, resulting in the reduction of lung metastasis of intravenously injected 4T1.2 cells. Overexpressing CXCL16 did not enhance lung metastasis, suggesting that the level of endogenous expression is sufficient to protect

metastatic cells from destruction *in vivo*, and that further increasing CXCL16 expression does not augment their resistance to attack. How CXCL16 mediates its protective effects is as yet undetermined and is the subject of ongoing investigation in this laboratory. However, an interesting possibility to consider is that in normal physiology, the interaction of CXCL16 on DC with CXCR6 on CD8⁺ T cells during priming protects the DC from destruction by the activated T cell.

It should be noted that the metastasis assays with the 4T16 and 4T Δ 16 cells could be performed only once due to time constraints, and further experiments to increase statistical power are required to confirm the observations documented in this thesis. However, the data presented here provide a tantalising glimpse of a potentially novel aspect of CXCL16 function in the context of tumour biology, and possibly also in normal physiology during the generation of immune responses. *In vitro* cytotoxicity assays to confirm the modulatory effect of CXCL16 on the cytolytic killing of 4T1.2 cells by NK cells specifically should be the next step in this line of investigation. *In vivo* experiments with a smaller inoculum of 4T16 and 4T Δ 16 cells to avoid overwhelming the NK cell response may reveal differences in primary tumour growth characteristics. In addition, depletion of NK cells with antiasialoGM1 would give an indication of the significance of these cells for the effects of CXCL16 and CXCL16₍₉₋₂₂₀₎ on tumour growth and metastasis *in vivo*.

6.5 Perspectives and concluding remarks

The work presented in this thesis demonstrates that overexpression of a chemokine and its antagonist has profound effects on the progression of breast cancer in an animal model. These findings await further confirmation in other models of cancer and in human disease, but given the high degree of evolutionary conservation of CXCL12 and CXCR4 and the documented expression of CXCR4 in a wide range of cancers, it seems likely that the anti-cancer effects of this chemokine and its antagonist would not be restricted to the breast cancer model studied here.

It is interesting that both CXCL12 and its antagonist have anti-tumour properties; this demonstrates that the level and location of CXCL12 expression can profoundly affect tumorigenesis and metastatic disease. Early high-level expression of CXCL12 at the

tumour site induces anti-tumour responses that appear to depend on the adaptive arm of the immune system. On the other hand, the data obtained through the expression of CXCL12_(P2G) suggest that endogenous CXCL12 in the circulation and/or in secondary tissues can promote metastatic spread of malignant cells. The experimental data obtained confirm the findings of previous studies demonstrating a role for CXCL12/CXCR4 in tumour metastasis and extend this work to show that endogenous CXCL12 can specifically influence metastatic steps independently of any effect on primary tumour growth.

The ability of CXCL12 alone to mobilise anti-tumour immunity, especially against such a poorly immunogenic and immunosuppressive tumour as that investigated here, testifies to its ability to co-ordinate multiple necessary components of the immune response. While not able to induce complete tumour regression on its own, CXCL12 may prove to be a potent and effective adjuvant when combined with current immunotherapeutic strategies. First it will be necessary to establish the ability of CXCL12 to induce anti-tumour responses in a more realistic model of tumour therapy, where tumours and metastases have already developed. Though this study has demonstrated anti-metastatic properties of wild-type CXCL12, whether or not these were due to the inhibition of primary tumour growth, or to induction of a systemic anti-tumour response, remains to be addressed. Vaccination of mice with established disease with CXCL12-expressing tumour cells would provide insight into the effectiveness of the CXCL12-induced immune response against established tumours and metastases.

Antagonism of CXCL12 forms another potential anti-metastatic strategy, as demonstrated by the effect of CXCL12_(P2G) on breast cancer metastasis. However, whether this effect is due to the action of CXCL12_(P2G) on CXCR4, CXCR7 or both requires further investigation. To date, the majority of studies have focused on the role of CXCR4 in metastasis, and the influence of CXCR7 on this process remains a grey area. Studies to determine whether CXCL12_(P2G) can bind to CXCR7 are required to determine if there are grounds for further assessing the role of this receptor in metastasis. Should a role for CXCR7 be confirmed, approaches to anti-metastatic therapies targeting CXCR4 will need to be rethought. Currently, strategies being tested specifically target CXCR4 using small molecule inhibitors, antibodies or RNA interference. A more effective approach may be to target both CXCR4 and CXCR7 at the same time. Another factor for consideration is that targeting CXCR4, while inhibiting tumour metastasis, may also hinder anti-tumour immune responses through blockade of leukocyte recruitment to tumours. As CXCR7 expression seems to be largely restricted to embryonic tissues and transformed cells¹²⁹, it possible that specific blockade of this receptor can inhibit tumour metastasis while still leaving the way open for concomitant therapies with CXCL12 to enhance anti-tumour immunity. As specific reagents for the study of CXCR7 become available, experiments to test the viability of CXCR7 inhibition as an anti-metastatic therapeutic strategy will be possible.

The lack of effect of CXCL16 and its antagonist on primary breast tumour growth and spontaneous metastasis was somewhat surprising. More surprising was that despite this, CXCL16₍₉₋₂₂₀₎ significantly inhibited experimental metastasis. As discussed in Section 6.4 above, it may be that the major mediators of this effect are NK cells. Further work is needed confirm the role of these cells in the anti-metastatic effect of CXCL16₍₉₋₂₂₀₎, but the observations outline in this thesis indicate that antagonism of CXCL16 also represents a potential anti-metastatic therapy that can be used as an adjuvant to other anti-cancer therapies. Of course, it must be considered that treatment involving antagonism of CXCL16 may block recruitment to the tumour site of effector T cells generated by concomitant immune priming strategies. Therefore care must be taken to establish precisely how CXCL16 antagonism should be employed to maximise the effectiveness of combination therapies that incorporate this treatment.

In conclusion, this study demonstrates multiple significant roles for the chemokine CXCL12 in breast cancer progression. Endogenous host expression of CXCL12 clearly has pro-metastatic effects, as antagonism of CXCL12 activity inhibits breast cancer metastasis. However, inducing CXCL12 expression at the tumour site can enhance the anti-tumour immune response. In addition, this work has uncovered a potential pro-metastatic role for endogenously expressed CXCL16, which can be inhibited by expression of a CXCL16 antagonist. These findings provide a basis for further investigation of the roles of CXCL12 and CXCL16 in metastatic cancers of the breast, as well as other tissues. This work also highlights the potential of CXCL12, CXCL16 and inhibitors of their receptors as therapeutic agents to treat this complex disease.

Figure 6.1 Working model of CXCL12 action in primary mammary tumours.

CXCL12 expressed by the tumour recruits NK cells to the tumour mass (1), where they secrete IFN- γ and exert cytotoxic functions (2). CXCL12 also induces TNF- α production by TAM (3) which, together with IFN- γ , induces differentiation of MSC into macrophages (4) and immature DC into mature DC (5). Mature DC migrate to the DLN where they cross-present antigen to $CD8^+$ T cells (6). CXCL12 may filter from the tumour to the DLN where it co-stimulates T cell activation and their secretion of IL-2 (7). Activated $CD8^+$ T cells migrate back to the tumour (8) where they exert cytotoxic functions and secrete IFN- γ (9). IL-2 produced by these cells may also induce innate immune cells in the tumour to contribute to local IFN- γ production. IFN- γ may facilitate generation of NO-producing M1 macrophages (10) that exert cytotoxic functions against tumour cells (11). CXCL12 may also recruit NKT cells to the tumour (12) where they receive activation signals that induce their production of IFN- γ (13), though this may be a minor arm of the anti-tumour response which becomes significant only in the absence of T cells. See text for further details. DC, dendritic cell; DLN, draining lymph node; IFN- γ , interferon- γ ; MSC, myeloid suppressor cell; NK cell, natural killer cell; NO, nitric oxide; TNF- α , tumour necrosis factor- α .



Figure 6.2 Working model of CXCL12_(P2G) action in mammary tumour metastasis.

(A) CXCL12 promotes tumour cell metastasis. Metastatic cells escape from the primary tumour mass and enter the circulation, a process which may or may not require CXCL12 signalling. Circulating CXCL12 provides survival signals to detached tumour cells in the circulation and prevents anoikis. At secondary sites, such as the lungs, tumour cells arrest and adhere to the endothelium in response to CXCL12 produced in the tissues. CXCL12 may induce tumour cell secretion of MMP, thereby facilitating extravasation and invasion of the lung, as well as forming a chemotactic gradient that directs migration of malignant cells into the tissue. It is possible that CXCL12 also provides survival and proliferation signals required for tumour cells to expand in their new environment, though it does not seem to be required for proliferation of the primary tumour. **(B)** CXCL12_(P2G) blocks CXCL12-induced metastasis. Tumour cell production of CXCL12_(P2G) may block CXCL12 signalling at any or all of the stages of metastasis described in (A). This blockade may the result of interaction with CXCR4, or possibly with CXCR7, expressed on the tumour cell surface. The action of CXCL12_(P2G) inhibits metastasis to secondary sites, without affecting the growth of the primary tumour.





CHAPTER 7

References



CHAPTER 7: REFERENCES

- 1. Hanahan, D. and Weinberg, R.A. (2000). The hallmarks of cancer. *Cell*, **100**(1): 57-70.
- 2. Mellado, M., Rodriguez-Frade, J.M., Manes, S. and Martinez, A.C. (2001). Chemokine signalling and functional responses: the role of receptor dimerization and TK pathway activation. *Annu Rev Immunol*, **19**: 397-421.
- 3. Bray, F., McCarron, P. and Parkin, D.M. (2004). The changing global patterns of female breast cancer incidence and mortality. *Breast Cancer Res*, **6**(6): 229-39.
- 4. Giordano, S.H. (2005). A Review of the Diagnosis and Management of Male Breast Cancer. *Oncologist*, **10**(7): 471-479.
- 5. Nahleh, Z. and Girnius, S. (2006). Male breast cancer: a gender issue. *Nat Clin Pract Oncol*, **3**(8): 428-37.
- Muller, A., Homey, B., Soto, H., Ge, N., Catron, D., Buchanan, M.E., McClanahan, T., Murphy, E., Yuan, W., Wagner, S.N., Barrera, J.L., Mohar, A., Verastegui, E. and Zlotnik, A. (2001). Involvement of chemokine receptors in breast cancer metastasis. *Nature*, 410(6824): 50-6.
- 7. Keyomarsi, K., Tucker, S.L., Buchholz, T.A., Callister, M., Ding, Y., Hortobagyi, G.N., Bedrosian, I., Knickerbocker, C., Toyofuku, W., Lowe, M., Herliczek, T.W. and Bacus, S.S. (2002). Cyclin E and survival in patients with breast cancer. *N Engl J Med*, **347**(20): 1566-75.
- 8. Akli, S., Van Pelt, C.S., Bui, T., Multani, A.S., Chang, S., Johnson, D., Tucker, S. and Keyomarsi, K. (2007). Overexpression of the low molecular weight cyclin E in transgenic mice induces metastatic mammary carcinomas through the disruption of the ARF-p53 pathway. *Cancer Res*, **67**(15): 7212-22.
- 9. Hui, L., Zheng, Y., Yan, Y., Bargonetti, J. and Foster, D.A. (2006). Mutant p53 in MDA-MB-231 breast cancer cells is stabilized by elevated phospholipase D activity and contributes to survival signals generated by phospholipase D. *Oncogene*, **25**(55): 7305-10.
- 10. Helton, E.S. and Chen, X. (2007). p53 modulation of the DNA damage response. *J Cell Biochem*, **100**(4): 883-96.
- 11. Herbert, B.S., Wright, W.E. and Shay, J.W. (2001). Telomerase and breast cancer. *Breast Cancer Res*, **3**(3): 146-9.
- 12. Cerone, M.A., Londono-Vallejo, J.A. and Autexier, C. (2006). Telomerase inhibition enhances the response to anticancer drug treatment in human breast cancer cells. *Mol Cancer Ther*, **5**(7): 1669-75.
- 13. Ronnov-Jessen, L., Petersen, O.W. and Bissell, M.J. (1996). Cellular changes involved in conversion of normal to malignant breast: importance of the stromal reaction. *Physiol Rev*, **76**(1): 69-125.
- 14. Derynck, R., Akhurst, R.J. and Balmain, A. (2001). TGF-beta signalling in tumor suppression and cancer progression. *Nat Genet*, **29**(2): 117-29.
- 15. Leek, R.D. and Harris, A.L. (2002). Tumor-associated macrophages in breast cancer. *J Mammary Gland Biol Neoplasia*, **7**(2): 177-89.
- 16. Carmeliet, P. and Jain, R.K. (2000). Angiogenesis in cancer and other diseases. *Nature*, **407**(6801): 249-57.
- 17. Boudreau, N. and Myers, C. (2003). Breast cancer-induced angiogenesis: multiple mechanisms and the role of the microenvironment. *Breast Cancer Res*, **5**(3): 140-6.
- 18. Kossakowska, A.E., Huchcroft, S.A., Urbanski, S.J. and Edwards, D.R. (1996). Comparative analysis of the expression patterns of metalloproteinases and their

inhibitors in breast neoplasia, sporadic colorectal neoplasia, pulmonary carcinomas and malignant non-Hodgkin's lymphomas in humans. *Br J Cancer*, **73**(11): 1401-8.

- 19. Duffy, M.J. (2002). Urokinase-type plasminogen activator: a potent marker of metastatic potential in human cancers. *Biochem Soc Trans*, **30**(2): 207-10.
- 20. Derksen, P.W., Liu, X., Saridin, F., van der Gulden, H., Zevenhoven, J., Evers, B., van Beijnum, J.R., Griffioen, A.W., Vink, J., Krimpenfort, P., Peterse, J.L., Cardiff, R.D., Berns, A. and Jonkers, J. (2006). Somatic inactivation of E-cadherin and p53 in mice leads to metastatic lobular mammary carcinoma through induction of anoikis resistance and angiogenesis. *Cancer Cell*, **10**(5): 437-49.
- 21. Sun, J. and Hemler, M.E. (2001). Regulation of MMP-1 and MMP-2 production through CD147/extracellular matrix metalloproteinase inducer interactions. *Cancer Res*, **61**(5): 2276-81.
- Yang, J.M., O'Neill, P., Jin, W., Foty, R., Medina, D.J., Xu, Z., Lomas, M., Arndt, G.M., Tang, Y., Nakada, M., Yan, L. and Hait, W.N. (2006). Extracellular matrix metalloproteinase inducer (CD147) confers resistance of breast cancer cells to Anoikis through inhibition of Bim. *J Biol Chem*, 281(14): 9719-27.
- 23. Skobe, M., Hawighorst, T., Jackson, D.G., Prevo, R., Janes, L., Velasco, P., Riccardi, L., Alitalo, K., Claffey, K. and Detmar, M. (2001). Induction of tumor lymphangiogenesis by VEGF-C promotes breast cancer metastasis. *Nat Med*, 7(2): 192-8.
- 24. Pardoll, D. (2003). Does the immune system see tumors as foreign or self? *Annual Review of Immunology*, **21**(1): 807-839.
- 25. Dunn, G.P., Old, L.J. and Schreiber, R.D. (2004). The Immunobiology of Cancer Immunosurveillance and Immunoediting. *Immunity*, **21**(2): 137-148.
- Simson, L., Ellyard, J.I., Dent, L.A., Matthaei, K.I., Rothenberg, M.E., Foster, P.S., Smyth, M.J. and Parish, C.R. (2007). Regulation of carcinogenesis by IL-5 and CCL11: a potential role for eosinophils in tumor immune surveillance. *J Immunol*, 178(7): 4222-9.
- 27. Redondo, M., Garcia, J., Villar, E., Rodrigo, I., Perea-Milla, E., Serrano, A. and Morell, M. (2003). Major histocompatibility complex status in breast carcinogenesis and relationship to apoptosis. *Hum Pathol*, **34**(12): 1283-9.
- 28. Goepel, J.R., Rees, R.C., Rogers, K., Stoddard, C.J., Thomas, W.E. and Shepherd, L. (1991). Loss of monomorphic and polymorphic HLA antigens in metastatic breast and colon carcinoma. *Br J Cancer*, **64**(5): 880-3.
- 29. Mukherjee, P., Madsen, C.S., Ginardi, A.R., Tinder, T.L., Jacobs, F., Parker, J., Agrawal, B., Longenecker, B.M. and Gendler, S.J. (2003). Mucin 1-specific immunotherapy in a mouse model of spontaneous breast cancer. *J Immunother* (1997), **26**(1): 47-62.
- 30. Knutson, K.L., Lu, H., Stone, B., Reiman, J.M., Behrens, M.D., Prosperi, C.M., Gad, E.A., Smorlesi, A. and Disis, M.L. (2006). Immunoediting of cancers may lead to epithelial to mesenchymal transition. *J Immunol*, **177**(3): 1526-33.
- 31. Kmieciak, M., Knutson, K.L., Dumur, C.I. and Manjili, M.H. (2007). HER-2/neu antigen loss and relapse of mammary carcinoma are actively induced by T cell-mediated anti-tumor immune responses. *Eur J Immunol*, **37**(3): 675-85.
- 32. Manjili, M.H., Arnouk, H., Knutson, K.L., Kmieciak, M., Disis, M.L., Subjeck, J.R. and Kazim, A.L. (2006). Emergence of immune escape variant of mammary tumors that has distinct proteomic profile and a reduced ability to induce "danger signals". *Breast Cancer Res Treat*, **96**(3): 233-41.
- 33. Trinchieri, G. (2003). Interleukin-12 and the regulation of innate resistance and adaptive immunity. *Nat Rev Immunol*, **3**(2): 133-46.

- 34. Ikeda, H., Old, L.J. and Schreiber, R.D. (2002). The roles of IFN gamma in protection against tumor development and cancer immunoediting. *Cytokine Growth Factor Rev*, **13**(2): 95-109.
- 35. Mocellin, S., Marincola, F.M. and Young, H.A. (2005). Interleukin-10 and the immune response against cancer: a counterpoint. *J Leukoc Biol*, **78**(5): 1043-1051.
- 36. Seo, N., Tokura, Y., Takigawa, M. and Egawa, K. (1999). Depletion of IL-10- and TGF-beta-producing regulatory gamma delta T cells by administering a daunomycin-conjugated specific monoclonal antibody in early tumor lesions augments the activity of CTLs and NK cells. *J Immunol*, **163**(1): 242-9.
- 37. Kundu, N., Beaty, T.L., Jackson, M.J. and Fulton, A.M. (1996). Antimetastatic and antitumor activities of interleukin 10 in a murine model of breast cancer. *J Natl Cancer Inst*, **88**(8): 536-41.
- 38. Li, M.O., Wan, Y.Y., Sanjabi, S., Robertson, A.-K.L. and Flavell, R.A. (2006). Tranforming Growth Factor-beta regulation of immune responses. *Annual Review* of *Immunology*, **24**(1): 99-146.
- 39. Harthun, N.L., Weaver, A.M., Brinckerhoff, L.H., Deacon, D.H., Gonias, S.L. and Slingluff, C.L., Jr. (1998). Activated alpha 2-macroglobulin reverses the immunosuppressive activity in human breast cancer cell-conditioned medium by selectively neutralizing transforming growth factor-beta in the presence of interleukin-2. *J Immunother (1997)*, **21**(2): 85-94.
- 40. Park, J.A., Wang, E., Kurt, R.A., Schluter, S.F., Hersh, E.M. and Akporiaye, E.T. (1997). Expression of an antisense transforming growth factor-beta1 transgene reduces tumorigenicity of EMT6 mammary tumor cells. *Cancer Gene Ther*, **4**(1): 42-50.
- 41. Wu, R.S., Kobie, J.J., Besselsen, D.G., Fong, T.C., Mack, V.D., McEarchern, J.A. and Akporiaye, E.T. (2001). Comparative analysis of IFN-gamma B7.1 and antisense TGF-beta gene transfer on the tumorigenicity of a poorly immunogenic metastatic mammary carcinoma. *Cancer Immunol Immunother*, **50**(5): 229-40.
- 42. Kobie, J.J., Wu, R.S., Kurt, R.A., Lou, S., Adelman, M.K., Whitesell, L.J., Ramanathapuram, L.V., Arteaga, C.L. and Akporiaye, E.T. (2003). Transforming growth factor beta inhibits the antigen-presenting functions and antitumor activity of dendritic cell vaccines. *Cancer Res*, **63**(8): 1860-4.
- 43. Cools, N., Ponsaerts, P., Van Tendeloo, V.F.I. and Berneman, Z.N. (2007). Balancing between immunity and tolerance: an interplay between dendritic cells, regulatory T cells, and effector T cells. *J Leukoc Biol*: jlb.0307166.
- 44. Iwamoto, M., Shinohara, H., Miyamoto, A., Okuzawa, M., Mabuchi, H., Nohara, T., Gon, G., Toyoda, M. and Tanigawa, N. (2003). Prognostic value of tumorinfiltrating dendritic cells expressing CD83 in human breast carcinomas. *Int J Cancer*, **104**(1): 92-7.
- 45. Kiertscher, S.M., Luo, J., Dubinett, S.M. and Roth, M.D. (2000). Tumors promote altered maturation and early apoptosis of monocyte-derived dendritic cells. *J Immunol*, **164**(3): 1269-76.
- Liyanage, U.K., Moore, T.T., Joo, H.G., Tanaka, Y., Herrmann, V., Doherty, G., Drebin, J.A., Strasberg, S.M., Eberlein, T.J., Goedegebuure, P.S. and Linehan, D.C. (2002). Prevalence of regulatory T cells is increased in peripheral blood and tumor microenvironment of patients with pancreas or breast adenocarcinoma. *J Immunol*, 169(5): 2756-61.
- 47. Ambrosino, E., Spadaro, M., Iezzi, M., Curcio, C., Forni, G., Musiani, P., Wei, W.Z. and Cavallo, F. (2006). Immunosurveillance of Erbb2 carcinogenesis in

transgenic mice is concealed by a dominant regulatory T-cell self-tolerance. *Cancer Res*, **66**(15): 7734-40.

- 48. Knutson, K.L., Dang, Y., Lu, H., Lukas, J., Almand, B., Gad, E., Azeke, E. and Disis, M.L. (2006). IL-2 immunotoxin therapy modulates tumor-associated regulatory T cells and leads to lasting immune-mediated rejection of breast cancers in neu-transgenic mice. *J Immunol*, **177**(1): 84-91.
- 49. Wang, H.Y. and Wang, R.-F. (2007). Regulatory T cells and cancer. *Current Opinion in Immunology: Lymphocyte development/Tumour immunology*, **19**(2): 217-223.
- 50. Miyara, M. and Sakaguchi, S. (2007). Natural regulatory T cells: mechanisms of suppression. *Trends Mol Med*, **13**(3): 108-16.
- 51. Serafini, P., Borrello, I. and Bronte, V. (2006). Myeloid suppressor cells in cancer: Recruitment, phenotype, properties, and mechanisms of immune suppression. *Seminars in Cancer Biology: The Inflammation-Cancer Linkage: A Double-Edged Sword?*, **16**(1): 53-65.
- 52. Bunt, S.K., Sinha, P., Clements, V.K., Leips, J. and Ostrand-Rosenberg, S. (2006). Inflammation induces myeloid-derived suppressor cells that facilitate tumor progression. *J Immunol*, **176**(1): 284-90.
- 53. Huang, B., Pan, P.Y., Li, Q., Sato, A.I., Levy, D.E., Bromberg, J., Divino, C.M. and Chen, S.H. (2006). Gr-1+CD115+ immature myeloid suppressor cells mediate the development of tumor-induced T regulatory cells and T-cell anergy in tumor-bearing host. *Cancer Res*, **66**(2): 1123-31.
- 54. Rodriguez, P.C., Quiceno, D.G., Zabaleta, J., Ortiz, B., Zea, A.H., Piazuelo, M.B., Delgado, A., Correa, P., Brayer, J., Sotomayor, E.M., Antonia, S., Ochoa, J.B. and Ochoa, A.C. (2004). Arginase I production in the tumor microenvironment by mature myeloid cells inhibits T-cell receptor expression and antigen-specific T-cell responses. *Cancer Res*, **64**(16): 5839-49.
- 55. Ezernitchi, A.V., Vaknin, I., Cohen-Daniel, L., Levy, O., Manaster, E., Halabi, A., Pikarsky, E., Shapira, L. and Baniyash, M. (2006). TCR zeta down-regulation under chronic inflammation is mediated by myeloid suppressor cells differentially distributed between various lymphatic organs. *J Immunol*, **177**(7): 4763-72.
- 56. Nagaraj, S., Gupta, K., Pisarev, V., Kinarsky, L., Sherman, S., Kang, L., Herber, D.L., Schneck, J. and Gabrilovich, D.I. (2007). Altered recognition of antigen is a mechanism of CD8+ T cell tolerance in cancer. **13**(7): 828-835.
- 57. Mazzoni, A., Bronte, V., Visintin, A., Spitzer, J.H., Apolloni, E., Serafini, P., Zanovello, P. and Segal, D.M. (2002). Myeloid suppressor lines inhibit T cell responses by an NO-dependent mechanism. *J Immunol*, **168**(2): 689-95.
- 58. Terabe, M., Matsui, S., Park, J.M., Mamura, M., Noben-Trauth, N., Donaldson, D.D., Chen, W., Wahl, S.M., Ledbetter, S., Pratt, B., Letterio, J.J., Paul, W.E. and Berzofsky, J.A. (2003). Transforming growth factor-beta production and myeloid cells are an effector mechanism through which CD1d-restricted T cells block cytotoxic T lymphocyte-mediated tumor immunosurveillance: abrogation prevents tumor recurrence. *J Exp Med*, **198**(11): 1741-52.
- 59. Liu, C., Yu, S., Kappes, J., Wang, J., Grizzle, W.E., Zinn, K.R. and Zhang, H.G. (2007). Expansion of spleen myeloid suppressor cells represses NK cell cytotoxicity in tumor-bearing host. *Blood*, **109**(10): 4336-42.
- 60. Nahta, R. and Esteva, F.J. (2007). Trastuzumab: triumphs and tribulations. *Oncogene*, **26**(25): 3637-43.

- 61. Allan, C.P., Turtle, C.J., Mainwaring, P.N., Pyke, C. and Hart, D.N. (2004). The immune response to breast cancer, and the case for DC immunotherapy. *Cytotherapy*, **6**(2): 154-63.
- 62. Yamano, T., Kaneda, Y., Hiramatsu, S.H., Huang, S., Tran, A.N., Giuliano, A.E. and Hoon, D.S. (2007). Immunity against breast cancer by TERT DNA vaccine primed with chemokine CCL21. *Cancer Gene Ther*, **14**(5): 451-9.
- 63. Walser, T.C., Ma, X., Kundu, N., Dorsey, R., Goloubeva, O. and Fulton, A.M. (2007). Immune-mediated modulation of breast cancer growth and metastasis by the chemokine Mig (CXCL9) in a murine model. *J Immunother (1997)*, **30**(5): 490-8.
- 64. Chu, Y., Yang, X., Xu, W., Wang, Y., Guo, Q. and Xiong, S. (2007). In situ expression of IFN-gamma-inducible T cell alpha chemoattractant in breast cancer mounts an enhanced specific anti-tumor immunity which leads to tumor regression. *Cancer Immunol Immunother*, **56**(10): 1539-49.
- 65. Braun, S.E., Chen, K., Foster, R.G., Kim, C.H., Hromas, R., Kaplan, M.H., Broxmeyer, H.E. and Cornetta, K. (2000). The CC chemokine CK beta-11/MIP-3 beta/ELC/Exodus 3 mediates tumor rejection of murine breast cancer cells through NK cells. *J Immunol*, **164**(8): 4025-31.
- 66. Guiducci, C., Di Carlo, E., Parenza, M., Hitt, M., Giovarelli, M., Musiani, P. and Colombo, M.P. (2004). Intralesional injection of adenovirus encoding CC chemokine ligand 16 inhibits mammary tumor growth and prevents metastatic-induced death after surgical removal of the treated primary tumor. *J Immunol*, **172**(7): 4026-36.
- 67. Javni, J. (1999). *Suppression of Murine Tumours by MIP-1alpha*. Doctor of Philosophy Thesis, Peter MacCallum Cancer Institute.
- 68. Rossi, D. and Zlotnik, A. (2000). The biology of chemokines and their receptors. *Annu Rev Immunol*, **18**: 217-42.
- Murphy, P.M., Baggiolini, M., Charo, I.F., Hebert, C.A., Horuk, R., Matsushima, K., Miller, L.H., Oppenheim, J.J. and Power, C.A. (2000). International union of pharmacology. XXII. Nomenclature for chemokine receptors. *Pharmacol Rev*, 52(1): 145-76.
- 70. Murphy, P.M. (2002). International Union of Pharmacology. XXX. Update on chemokine receptor nomenclature. *Pharmacol Rev*, **54**(2): 227-9.
- 71. Horuk, R. (2001). Chemokine receptors. *Cytokine Growth Factor Rev*, **12**(4): 313-35.
- 72. Procko, E. and McColl, S.R. (2005). Leukocytes on the move with phosphoinositide 3-kinase and its downstream effectors. *Bioessays*, **27**(2): 153-63.
- 73. Kanda, S., Mochizuki, Y. and Kanetake, H. (2003). Stromal cell-derived factorlalpha induces tube-like structure formation of endothelial cells through phosphoinositide 3-kinase. *J Biol Chem*, **278**(1): 257-62.
- 74. Liu, L., Puri, K.D., Penninger, J.M. and Kubes, P. (2007). Leukocyte PI3Kgamma and PI3Kdelta have temporally distinct roles for leukocyte recruitment in vivo. *Blood*, **110**(4): 1191-8.
- 75. Ono, S.J., Nakamura, T., Miyazaki, D., Ohbayashi, M., Dawson, M. and Toda, M. (2003). Chemokines: roles in leukocyte development, trafficking, and effector function. *J Allergy Clin Immunol*, **111**(6): 1185-99; quiz 1200.
- 76. Cinamon, G., Grabovsky, V., Winter, E., Franitza, S., Feigelson, S., Shamri, R., Dwir, O. and Alon, R. (2001). Novel chemokine functions in lymphocyte migration through vascular endothelium under shear flow. *J Leukoc Biol*, **69**(6): 860-6.

- 77. Rittner, H.L., Labuz, D., Richter, J.F., Brack, A., Schafer, M., Stein, C. and Mousa, S.A. (2007). CXCR1/2 ligands induce p38 MAPK-dependent translocation and release of opioid peptides from primary granules in vitro and in vivo. *Brain Behav Immun*.
- 78. Green, S.R., Han, K.H., Chen, Y., Almazan, F., Charo, I.F., Miller, Y.I. and Quehenberger, O. (2006). The CC chemokine MCP-1 stimulates surface expression of CX3CR1 and enhances the adhesion of monocytes to fractalkine/CX3CL1 via p38 MAPK. *J Immunol*, **176**(12): 7412-20.
- 79. Arai, A., Aoki, M., Weihua, Y., Jin, A. and Miura, O. (2006). CrkL plays a role in SDF-1-induced activation of the Raf-1/MEK/Erk pathway through Ras and Rac to mediate chemotactic signalling in hematopoietic cells. *Cell Signal*, **18**(12): 2162-71.
- Spinetti, G., Bernardini, G., Camarda, G., Mangoni, A., Santoni, A., Capogrossi, M.C. and Napolitano, M. (2003). The chemokine receptor CCR8 mediates rescue from dexamethasone-induced apoptosis via an ERK-dependent pathway. *J Leukoc Biol*, 73(1): 201-7.
- 81. Kumar, A., Humphreys, T.D., Kremer, K.N., Bramati, P.S., Bradfield, L., Edgar, C.E. and Hedin, K.E. (2006). CXCR4 physically associates with the T cell receptor to signal in T cells. *Immunity*, **25**(2): 213-24.
- 82. Biswas, S.K. and Sodhi, A. (2002). Tyrosine phosphorylation-mediated signal transduction in MCP-1-induced macrophage activation: role for receptor dimerization, focal adhesion protein complex and JAK/STAT pathway. *Int Immunopharmacol*, **2**(8): 1095-107.
- 83. Burger, M., Hartmann, T., Burger, J.A. and Schraufstatter, I. (2005). KSHV-GPCR and CXCR2 transforming capacity and angiogenic responses are mediated through a JAK2-STAT3-dependent pathway. *Oncogene*, **24**(12): 2067-75.
- 84. Mueller, A. and Strange, P.G. (2004). CCL3, acting via the chemokine receptor CCR5, leads to independent activation of Janus kinase 2 (JAK2) and Gi proteins. *FEBS Lett*, **570**(1-3): 126-32.
- 85. Shahrara, S., Amin, M.A., Woods, J.M., Haines, G.K. and Koch, A.E. (2003). Chemokine receptor expression and in vivo signalling pathways in the joints of rats with adjuvant-induced arthritis. *Arthritis Rheum*, **48**(12): 3568-83.
- 86. Soldevila, G., Licona, I., Salgado, A., Ramirez, M., Chavez, R. and Garcia-Zepeda, E. (2004). Impaired chemokine-induced migration during T-cell development in the absence of Jak 3. *Immunology*, **112**(2): 191-200.
- 87. Stein, J.V., Soriano, S.F., M'Rini, C., Nombela-Arrieta, C., de Buitrago, G.G., Rodriguez-Frade, J.M., Mellado, M., Girard, J.P. and Martinez, A.C. (2003). CCR7-mediated physiological lymphocyte homing involves activation of a tyrosine kinase pathway. *Blood*, **101**(1): 38-44.
- Vila-Coro, A.J., Rodriguez-Frade, J.M., Martin De Ana, A., Moreno-Ortiz, M.C., Martinez, A.C. and Mellado, M. (1999). The chemokine SDF-1alpha triggers CXCR4 receptor dimerization and activates the JAK/STAT pathway. *Faseb J*, 13(13): 1699-710.
- 89. Wong, M. and Fish, E.N. (1998). RANTES and MIP-1alpha activate stats in T cells. *J Biol Chem*, **273**(1): 309-14.
- 90. Wong, M., Uddin, S., Majchrzak, B., Huynh, T., Proudfoot, A.E., Platanias, L.C. and Fish, E.N. (2001). Rantes activates Jak2 and Jak3 to regulate engagement of multiple signalling pathways in T cells. *J Biol Chem*, **276**(14): 11427-31.
- 91. Soriano, S.F., Serrano, A., Hernanz-Falcon, P., Martin de Ana, A., Monterrubio, M., Martinez, C., Rodriguez-Frade, J.M. and Mellado, M. (2003). Chemokines

integrate JAK/STAT and G-protein pathways during chemotaxis and calcium flux responses. *Eur J Immunol*, **33**(5): 1328-33.

- 92. Neel, N.F., Schutyser, E., Sai, J., Fan, G.-H. and Richmond, A. (2005). Chemokine receptor internalization and intracellular trafficking. *Cytokine & Growth Factor Reviews: Chemokines*, **16**(6): 637-658.
- 93. Minina, S., Reichman-Fried, M. and Raz, E. (2007). Control of receptor internalization, signalling level, and precise arrival at the target in guided cell migration. *Curr Biol*, **17**(13): 1164-72.
- 94. Manes, S., Gomez-Mouton, C., Lacalle, R.A., Jimenez-Baranda, S., Mira, E. and Martinez-A, C. (2005). Mastering time and space: immune cell polarization and chemotaxis. *Seminars in Immunology: Spacial Organization in Immune Cell Signalling*, **17**(1): 77-86.
- 95. Johnston, B. and Butcher, E.C. (2002). Chemokines in rapid leukocyte adhesion triggering and migration. *Seminars in Immunology*, **14**(2): 83-92.
- 96. Baggiolini, M. (1998). Chemokines and leukocyte traffic. *Nature*, **392**(6676): 565-8.
- 97. Djeu, J.Y., Matsushima, K., Oppenheim, J.J., Shiotsuki, K. and Blanchard, D.K. (1990). Functional activation of human neutrophils by recombinant monocytederived neutrophil chemotactic factor/IL-8. *J Immunol*, **144**(6): 2205-10.
- 98. Brandt, E., Petersen, F. and Flad, H.D. (1992). Recombinant tumor necrosis factoralpha potentiates neutrophil degranulation in response to host defense cytokines neutrophil-activating peptide 2 and IL-8 by modulating intracellular cyclic AMP levels. *J Immunol*, **149**(4): 1356-64.
- 99. Fujisawa, T., Kato, Y., Nagase, H., Atsuta, J., Terada, A., Iguchi, K., Kamiya, H., Morita, Y., Kitaura, M., Kawasaki, H., Yoshie, O. and Hirai, K. (2000). Chemokines induce eosinophil degranulation through CCR-3. *J Allergy Clin Immunol*, **106**(3): 507-13.
- 100. Chakrabarti, S. and Patel, K.D. (2005). Regulation of matrix metalloproteinase-9 release from IL-8-stimulated human neutrophils. *J Leukoc Biol*, **78**(1): 279-88.
- Vicari, A.P., Ait-Yahia, S., Chemin, K., Mueller, A., Zlotnik, A. and Caux, C. (2000). Antitumor effects of the mouse chemokine 6Ckine/SLC through angiostatic and immunological mechanisms. *J Immunol*, 165(4): 1992-2000.
- 102. Bernardini, G., Spinetti, G., Ribatti, D., Camarda, G., Morbidelli, L., Ziche, M., Santoni, A., Capogrossi, M.C. and Napolitano, M. (2000). I-309 binds to and activates endothelial cell functions and acts as an angiogenic molecule in vivo. *Blood*, 96(13): 4039-45.
- 103. Salcedo, R., Ponce, M.L., Young, H.A., Wasserman, K., Ward, J.M., Kleinman, H.K., Oppenheim, J.J. and Murphy, W.J. (2000). Human endothelial cells express CCR2 and respond to MCP-1: direct role of MCP-1 in angiogenesis and tumor progression. *Blood*, 96(1): 34-40.
- 104. Salcedo, R., Young, H.A., Ponce, M.L., Ward, J.M., Kleinman, H.K., Murphy, W.J. and Oppenheim, J.J. (2001). Eotaxin (CCL11) induces in vivo angiogenic responses by human CCR3+ endothelial cells. *J Immunol*, **166**(12): 7571-8.
- 105. Cheng, S.S., Lukacs, N.W. and Kunkel, S.L. (2002). Eotaxin/CCL11 suppresses IL-8/CXCL8 secretion from human dermal microvascular endothelial cells. *J Immunol*, **168**(6): 2887-94.
- 106. Hwang, J., Kim, C.W., Son, K.N., Han, K.Y., Lee, K.H., Kleinman, H.K., Ko, J., Na, D.S., Kwon, B.S., Gho, Y.S. and Kim, J. (2004). Angiogenic activity of human CC chemokine CCL15 in vitro and in vivo. *FEBS Lett*, **570**(1-3): 47-51.

- 107. Strasly, M., Doronzo, G., Capello, P., Valdembri, D., Arese, M., Mitola, S., Moore, P., Alessandri, G., Giovarelli, M. and Bussolino, F. (2004). CCL16 activates an angiogenic program in vascular endothelial cells. *Blood*, **103**(1): 40-9.
- 108. Hwang, J., Son, K.N., Kim, C.W., Ko, J., Na, D.S., Kwon, B.S., Gho, Y.S. and Kim, J. (2005). Human CC chemokine CCL23, a ligand for CCR1, induces endothelial cell migration and promotes angiogenesis. *Cytokine*, **30**(5): 254-63.
- 109. Addison, C.L., Arenberg, D.A., Morris, S.B., Xue, Y.Y., Burdick, M.D., Mulligan, M.S., Iannettoni, M.D. and Strieter, R.M. (2000). The CXC chemokine, monokine induced by interferon-gamma, inhibits non-small cell lung carcinoma tumor growth and metastasis. *Hum Gene Ther*, **11**(2): 247-61.
- 110. Salcedo, R., Resau, J.H., Halverson, D., Hudson, E.A., Dambach, M., Powell, D., Wasserman, K. and Oppenheim, J.J. (2000). Differential expression and responsiveness of chemokine receptors (CXCR1-3) by human microvascular endothelial cells and umbilical vein endothelial cells. *Faseb J*, **14**(13): 2055-64.
- 111. Salcedo, R., Wasserman, K., Young, H.A., Grimm, M.C., Howard, O.M., Anver, M.R., Kleinman, H.K., Murphy, W.J. and Oppenheim, J.J. (1999). Vascular endothelial growth factor and basic fibroblast growth factor induce expression of CXCR4 on human endothelial cells: In vivo neovascularization induced by stromal-derived factor-1alpha. *Am J Pathol*, **154**(4): 1125-35.
- 112. Yu, L., Cecil, J., Peng, S.B., Schrementi, J., Kovacevic, S., Paul, D., Su, E.W. and Wang, J. (2006). Identification and expression of novel isoforms of human stromal cell-derived factor 1. *Gene*, **374**: 174-9.
- 113. Sierra, M.D., Yang, F., Narazaki, M., Salvucci, O., Davis, D., Yarchoan, R., Zhang, H.H., Fales, H. and Tosato, G. (2003). Differential processing of stromal-derived factor-1 {alpha} and {beta} explains functional diversity. *Blood*, **2**: 2.
- 114. Feng, Y., Broder, C.C., Kennedy, P.E. and Berger, E.A. (1996). HIV-1 entry cofactor: functional cDNA cloning of a seven-transmembrane, G protein-coupled receptor. *Science*, **272**(5263): 872-7.
- 115. Gupta, S.K., Lysko, P.G., Pillarisetti, K., Ohlstein, E. and Stadel, J.M. (1998). Chemokine receptors in human endothelial cells. Functional expression of CXCR4 and its transcriptional regulation by inflammatory cytokines. *J Biol Chem*, **273**(7): 4282-7.
- 116. Mirshahi, F., Pourtau, J., Li, H., Muraine, M., Trochon, V., Legrand, E., Vannier, J., Soria, J., Vasse, M. and Soria, C. (2000). SDF-1 activity on microvascular endothelial cells: consequences on angiogenesis in in vitro and in vivo models. *Thromb Res*, **99**(6): 587-94.
- 117. Salvucci, O., Yao, L., Villalba, S., Sajewicz, A., Pittaluga, S. and Tosato, G. (2002). Regulation of endothelial cell branching morphogenesis by endogenous chemokine stromal-derived factor-1. *Blood*, **99**(8): 2703-11.
- 118. Feil, C. and Augustin, H.G. (1998). Endothelial cells differentially express functional CXC-chemokine receptor-4 (CXCR-4/fusin) under the control of autocrine activity and exogenous cytokines. *Biochem Biophys Res Commun*, 247(1): 38-45.
- 119. Crane, I.J., Wallace, C.A., McKillop-Smith, S. and Forrester, J.V. (2000). CXCR4 receptor expression on human retinal pigment epithelial cells from the blood-retina barrier leads to chemokine secretion and migration in response to stromal cell-derived factor 1 alpha. *J Immunol*, **165**(8): 4372-8.
- 120. Murdoch, C., Monk, P.N. and Finn, A. (1999). Functional expression of chemokine receptor CXCR4 on human epithelial cells. *Immunology*, **98**(1): 36-41.

- 121. Smith, J.M., Johanesen, P.A., Wendt, M.K., Binion, D.G. and Dwinell, M.B. (2004). CXCL12 activation of CXCR4 regulates mucosal host defense through stimulation of epithelial cell migration and promotion of intestinal barrier integrity. *Am J Physiol Gastrointest Liver Physiol*.
- Lazarini, F., Tham, T.N., Casanova, P., Arenzana-Seisdedos, F. and Dubois-Dalcq, M. (2003). Role of the alpha-chemokine stromal cell-derived factor (SDF-1) in the developing and mature central nervous system. *Glia*, 42(2): 139-48.
- 123. Molyneaux, K.A., Zinszner, H., Kunwar, P.S., Schaible, K., Stebler, J., Sunshine, M.J., O'Brien, W., Raz, E., Littman, D., Wylie, C. and Lehmann, R. (2003). The chemokine SDF1/CXCL12 and its receptor CXCR4 regulate mouse germ cell migration and survival. *Development*, **130**(18): 4279-86.
- 124. Ara, T., Nakamura, Y., Egawa, T., Sugiyama, T., Abe, K., Kishimoto, T., Matsui, Y. and Nagasawa, T. (2003). Impaired colonization of the gonads by primordial germ cells in mice lacking a chemokine, stromal cell-derived factor-1 (SDF-1). *Proc Natl Acad Sci U S A*, **100**(9): 5319-23.
- 125. Schioppa, T., Uranchimeg, B., Saccani, A., Biswas, S.K., Doni, A., Rapisarda, A., Bernasconi, S., Saccani, S., Nebuloni, M., Vago, L., Mantovani, A., Melillo, G. and Sica, A. (2003). Regulation of the chemokine receptor CXCR4 by hypoxia. *J Exp Med*, **198**(9): 1391-402.
- 126. Ceradini, D.J., Kulkarni, A.R., Callaghan, M.J., Tepper, O.M., Bastidas, N., Kleinman, M.E., Capla, J.M., Galiano, R.D., Levine, J.P. and Gurtner, G.C. (2004). Progenitor cell trafficking is regulated by hypoxic gradients through HIF-1 induction of SDF-1. *Nat Med*, **10**(8): 858-64.
- 127. Staller, P., Sulitkova, J., Lisztwan, J., Moch, H., Oakeley, E.J. and Krek, W. (2003). Chemokine receptor CXCR4 downregulated by von Hippel-Lindau tumour suppressor pVHL. *Nature*, **425**(6955): 307-11.
- 128. Balabanian, K., Lagane, B., Infantino, S., Chow, K.Y., Harriague, J., Moepps, B., Arenzana-Seisdedos, F., Thelen, M. and Bachelerie, F. (2005). The chemokine SDF-1/CXCL12 binds to and signals through the orphan receptor RDC1 in T lymphocytes. *J Biol Chem*.
- 129. Burns, J.M., Summers, B.C., Wang, Y., Melikian, A., Berahovich, R., Miao, Z., Penfold, M.E., Sunshine, M.J., Littman, D.R., Kuo, C.J., Wei, K., McMaster, B.E., Wright, K., Howard, M.C. and Schall, T.J. (2006). A novel chemokine receptor for SDF-1 and I-TAC involved in cell survival, cell adhesion, and tumor development. *J Exp Med*.
- 130. Infantino, S., Moepps, B. and Thelen, M. (2006). Expression and Regulation of the Orphan Receptor RDC1 and Its Putative Ligand in Human Dendritic and B Cells. *J Immunol*, **176**(4): 2197-2207.
- 131. Crump, M.P., Gong, J.H., Loetscher, P., Rajarathnam, K., Amara, A., Arenzana-Seisdedos, F., Virelizier, J.L., Baggiolini, M., Sykes, B.D. and Clark-Lewis, I. (1997). Solution structure and basis for functional activity of stromal cell-derived factor-1; dissociation of CXCR4 activation from binding and inhibition of HIV-1. *Embo J*, 16(23): 6996-7007.
- Braun, M., Wunderlin, M., Spieth, K., Knochel, W., Gierschik, P. and Moepps, B. (2002). Xenopus laevis Stromal cell-derived factor 1: conservation of structure and function during vertebrate development. *J Immunol*, 168(5): 2340-7.
- 133. Nagasawa, T., Hirota, S., Tachibana, K., Takakura, N., Nishikawa, S., Kitamura, Y., Yoshida, N., Kikutani, H. and Kishimoto, T. (1996). Defects of B-cell lymphopoiesis and bone-marrow myelopoiesis in mice lacking the CXC chemokine PBSF/SDF-1. *Nature*, **382**(6592): 635-8.

- 134. Tachibana, K., Hirota, S., Iizasa, H., Yoshida, H., Kawabata, K., Kataoka, Y., Kitamura, Y., Matsushima, K., Yoshida, N., Nishikawa, S., Kishimoto, T. and Nagasawa, T. (1998). The chemokine receptor CXCR4 is essential for vascularisation of the gastrointestinal tract. *Nature*, **393**(6685): 591-4.
- 135. Zou, Y.R., Kottmann, A.H., Kuroda, M., Taniuchi, I. and Littman, D.R. (1998). Function of the chemokine receptor CXCR4 in haematopoiesis and in cerebellar development. *Nature*, **393**(6685): 595-9.
- 136. Nagasawa, T., Kikutani, H. and Kishimoto, T. (1994). Molecular cloning and structure of a pre-B-cell growth-stimulating factor. *Proc Natl Acad Sci U S A*, **91**(6): 2305-9.
- Peled, A., Petit, I., Kollet, O., Magid, M., Ponomaryov, T., Byk, T., Nagler, A., Ben-Hur, H., Many, A., Shultz, L., Lider, O., Alon, R., Zipori, D. and Lapidot, T. (1999). Dependence of human stem cell engraftment and repopulation of NOD/SCID mice on CXCR4. *Science*, 283(5403): 845-8.
- 138. Sugiyama, T., Kohara, H., Noda, M. and Nagasawa, T. (2006). Maintenance of the hematopoietic stem cell pool by CXCL12-CXCR4 chemokine signalling in bone marrow stromal cell niches. *Immunity*, **25**(6): 977-88.
- 139. Flomenberg, N., Devine, S.M., Dipersio, J.F., Liesveld, J.L., McCarty, J.M., Rowley, S.D., Vesole, D.H., Badel, K. and Calandra, G. (2005). The use of AMD3100 plus G-CSF for autologous hematopoietic progenitor cell mobilization is superior to G-CSF alone. *Blood*, **106**(5): 1867-74.
- 140. Liles, W.C., Rodger, E., Broxmeyer, H.E., Dehner, C., Badel, K., Calandra, G., Christensen, J., Wood, B., Price, T.H. and Dale, D.C. (2005). Augmented mobilization and collection of CD34+ hematopoietic cells from normal human volunteers stimulated with granulocyte-colony-stimulating factor by single-dose administration of AMD3100, a CXCR4 antagonist. *Transfusion*, **45**(3): 295-300.
- 141. Devine, S.M., Flomenberg, N., Vesole, D.H., Liesveld, J., Weisdorf, D., Badel, K., Calandra, G. and DiPersio, J.F. (2004). Rapid mobilization of CD34+ cells following administration of the CXCR4 antagonist AMD3100 to patients with multiple myeloma and non-Hodgkin's lymphoma. *J Clin Oncol*, **22**(6): 1095-102.
- 142. Ma, Q., Jones, D. and Springer, T.A. (1999). The chemokine receptor CXCR4 is required for the retention of B lineage and granulocytic precursors within the bone marrow microenvironment. *Immunity*, **10**(4): 463-71.
- 143. Honczarenko, M., Douglas, R.S., Mathias, C., Lee, B., Ratajczak, M.Z. and Silberstein, L.E. (1999). SDF-1 responsiveness does not correlate with CXCR4 expression levels of developing human bone marrow B cells. *Blood*, **94**(9): 2990-8.
- 144. Bleul, C.C., Schultze, J.L. and Springer, T.A. (1998). B lymphocyte chemotaxis regulated in association with microanatomic localization, differentiation state, and B cell receptor engagement. *J Exp Med*, **187**(5): 753-62.
- 145. Bleul, C.C. and Boehm, T. (2000). Chemokines define distinct microenvironments in the developing thymus. *Eur J Immunol*, **30**(12): 3371-9.
- 146. Kim, C.H., Pelus, L.M., White, J.R. and Broxmeyer, H.E. (1998). Differential chemotactic behavior of developing T cells in response to thymic chemokines. *Blood*, **91**(12): 4434-43.
- 147. Plotkin, J., Prockop, S.E., Lepique, A. and Petrie, H.T. (2003). Critical role for CXCR4 signalling in progenitor localization and T cell differentiation in the postnatal thymus. *J Immunol*, **171**(9): 4521-7.
- 148. Vianello, F., Kraft, P., Mok, Y.T., Hart, W.K., White, N. and Poznansky, M.C. (2005). A CXCR4-dependent chemorepellent signal contributes to the emigration

of mature single-positive CD4 cells from the fetal thymus. *J Immunol*, **175**(8): 5115-25.

- 149. Bleul, C.C., Fuhlbrigge, R.C., Casasnovas, J.M., Aiuti, A. and Springer, T.A. (1996). A highly efficacious lymphocyte chemoattractant, stromal cell-derived factor 1 (SDF-1). *J Exp Med*, **184**(3): 1101-9.
- 150. Vecchi, A., Massimiliano, L., Ramponi, S., Luini, W., Bernasconi, S., Bonecchi, R., Allavena, P., Parmentier, M., Mantovani, A. and Sozzani, S. (1999). Differential responsiveness to constitutive vs. inducible chemokines of immature and mature mouse dendritic cells. *J Leukoc Biol*, **66**(3): 489-94.
- 151. Campbell, J.J., Qin, S., Unutmaz, D., Soler, D., Murphy, K.E., Hodge, M.R., Wu, L. and Butcher, E.C. (2001). Unique subpopulations of CD56+ NK and NK-T peripheral blood lymphocytes identified by chemokine receptor expression repertoire. *J Immunol*, **166**(11): 6477-82.
- 152. Inngjerdingen, M., Damaj, B. and Maghazachi, A.A. (2001). Expression and regulation of chemokine receptors in human natural killer cells. *Blood*, **97**(2): 367-75.
- 153. Beider, K., Nagler, A., Wald, O., Franitza, S., Dagan-Berger, M., Wald, H., Giladi, H., Brocke, S., Hanna, J., Mandelboim, O., Darash-Yahana, M., Galun, E. and Peled, A. (2003). Involvement of CXCR4 and IL-2 in the homing and retention of human NK and NK T cells to the bone marrow and spleen of NOD/SCID mice. *Blood*, **102**(6): 1951-8.
- 154. Wang, J.F., Liu, Z.Y. and Groopman, J.E. (1998). The alpha-chemokine receptor CXCR4 is expressed on the megakaryocytic lineage from progenitor to platelets and modulates migration and adhesion. *Blood*, **92**(3): 756-64.
- 155. Nanki, T. and Lipsky, P.E. (2000). Cutting edge: stromal cell-derived factor-1 is a costimulator for CD4+ T cell activation. *J Immunol*, **164**(10): 5010-4.
- 156. Peacock, J.W. and Jirik, F.R. (1999). TCR activation inhibits chemotaxis toward stromal cell-derived factor-1: evidence for reciprocal regulation between CXCR4 and the TCR. *J Immunol*, **162**(1): 215-23.
- 157. Ebert, L.M. and McColl, S.R. (2001). Coregulation of CXC chemokine receptor and CD4 expression on T lymphocytes during allogeneic activation. *J Immunol*, **166**(8): 4870-8.
- 158. Abbal, C., Jourdan, P., Hori, T., Bousquet, J., Yssel, H. and Pene, J. (1999). TCRmediated activation of allergen-specific CD45RO(+) memory T lymphocytes results in down-regulation of cell-surface CXCR4 expression and a strongly reduced capacity to migrate in response to stromal cell-derived factor-1. *Int Immunol*, **11**(9): 1451-62.
- 159. Colamussi, M.L., Secchiero, P., Gonelli, A., Marchisio, M., Zauli, G. and Capitani, S. (2001). Stromal derived factor-1 alpha (SDF-1 alpha) induces CD4+ T cell apoptosis via the functional up-regulation of the Fas (CD95)/Fas ligand (CD95L) pathway. *J Leukoc Biol*, **69**(2): 263-70.
- 160. Krammer, P.H. (2000). CD95's deadly mission in the immune system. *Nature*, **407**(6805): 789-95.
- 161. Van Parijs, L., Refaeli, Y., Lord, J.D., Nelson, B.H., Abbas, A.K. and Baltimore, D. (1999). Uncoupling IL-2 signals that regulate T cell proliferation, survival, and Fasmediated activation-induced cell death. *Immunity*, **11**(3): 281-8.
- 162. Heidemann, J., Ogawa, H., Rafiee, P., Lugering, N., Maaser, C., Domschke, W., Binion, D.G. and Dwinell, M.B. (2004). Mucosal angiogenesis regulation by CXCR4 and its ligand CXCL12 expressed by human intestinal microvascular endothelial cells. *Am J Physiol Gastrointest Liver Physiol*, **286**(6): G1059-68.

- 163. Neuhaus, T., Stier, S., Totzke, G., Gruenewald, E., Fronhoffs, S., Sachinidis, A., Vetter, H. and Ko, Y.D. (2003). Stromal cell-derived factor 1alpha (SDF-1alpha) induces gene-expression of early growth response-1 (Egr-1) and VEGF in human arterial endothelial cells and enhances VEGF induced cell proliferation. *Cell Prolif*, 36(2): 75-86.
- 164. Hiasa, K., Ishibashi, M., Ohtani, K., Inoue, S., Zhao, Q., Kitamoto, S., Sata, M., Ichiki, T., Takeshita, A. and Egashira, K. (2004). Gene transfer of stromal cellderived factor-1alpha enhances ischemic vasculogenesis and angiogenesis via vascular endothelial growth factor/endothelial nitric oxide synthase-related pathway: next-generation chemokine therapy for therapeutic neovascularization. *Circulation*, **109**(20): 2454-61.
- 165. Doitsidou, M., Reichman-Fried, M., Stebler, J., Koprunner, M., Dorries, J., Meyer, D., Esguerra, C.V., Leung, T. and Raz, E. (2002). Guidance of primordial germ cell migration by the chemokine SDF-1. *Cell*, **111**(5): 647-59.
- 166. Knaut, H., Werz, C., Geisler, R. and Nusslein-Volhard, C. (2003). A zebrafish homologue of the chemokine receptor Cxcr4 is a germ-cell guidance receptor. *Nature*, **421**(6920): 279-82.
- 167. Vasyutina, E., Stebler, J., Brand-Saberi, B., Schulz, S., Raz, E. and Birchmeier, C. (2005). CXCR4 and Gab1 cooperate to control the development of migrating muscle progenitor cells. *Genes Dev*, **19**(18): 2187-98.
- 168. Chalasani, S.H., Sabelko, K.A., Sunshine, M.J., Littman, D.R. and Raper, J.A. (2003). A Chemokine, SDF-1, Reduces the Effectiveness of Multiple Axonal Repellents and Is Required for Normal Axon Pathfinding. J. Neurosci., 23(4): 1360-1371.
- Chalasani, S.H., Sabol, A., Xu, H., Gyda, M.A., Rasband, K., Granato, M., Chien, C.-B. and Raper, J.A. (2007). Stromal Cell-Derived Factor-1 Antagonizes Slit/Robo Signalling In Vivo. J. Neurosci., 27(5): 973-980.
- 170. Dwinell, M.B., Eckmann, L., Leopard, J.D., Varki, N.M. and Kagnoff, M.F. (1999). Chemokine receptor expression by human intestinal epithelial cells. *Gastroenterology*, **117**(2): 359-67.
- 171. Dwinell, M.B., Ogawa, H., Barrett, K.E. and Kagnoff, M.F. (2004). SDF-1/CXCL12 regulates cAMP production and ion transport in intestinal epithelial cells via CXCR4. *Am J Physiol Gastrointest Liver Physiol*, **286**(5): G844-50.
- 172. Jordan, N.J., Kolios, G., Abbot, S.E., Sinai, M.A., Thompson, D.A., Petraki, K. and Westwick, J. (1999). Expression of functional CXCR4 chemokine receptors on human colonic epithelial cells. *J Clin Invest*, **104**(8): 1061-9.
- 173. Jaleel, M.A., Tsai, A.C., Sarkar, S., Freedman, P.V. and Rubin, L.P. (2004). Stromal cell-derived factor-1 (SDF-1) signalling regulates human placental trophoblast cell survival. *Mol Hum Reprod*, **10**(12): 901-9.
- 174. Wright, L.M., Maloney, W., Yu, X., Kindle, L., Collin-Osdoby, P. and Osdoby, P. (2005). Stromal cell-derived factor-1 binding to its chemokine receptor CXCR4 on precursor cells promotes the chemotactic recruitment, development and survival of human osteoclasts. *Bone*, **36**(5): 840-53.
- 175. Yu, X., Huang, Y., Collin-Osdoby, P. and Osdoby, P. (2003). Stromal cell-derived factor-1 (SDF-1) recruits osteoclast precursors by inducing chemotaxis, matrix metalloproteinase-9 (MMP-9) activity, and collagen transmigration. *J Bone Miner Res*, **18**(8): 1404-18.
- 176. Zannettino, A.C., Farrugia, A.N., Kortesidis, A., Manavis, J., To, L.B., Martin, S.K., Diamond, P., Tamamura, H., Lapidot, T., Fujii, N. and Gronthos, S. (2005). Elevated serum levels of stromal-derived factor-lalpha are associated with

increased osteoclast activity and osteolytic bone disease in multiple myeloma patients. *Cancer Res*, **65**(5): 1700-9.

- 177. Kato, M., Kitayama, J., Kazama, S. and Nagawa, H. (2003). Expression pattern of CXC chemokine receptor-4 is correlated with lymph node metastasis in human invasive ductal carcinoma. *Breast Cancer Res*, **5**(5): R144-50.
- 178. Kodama, J., Hasengaowa, Kusumoto, T., Seki, N., Matsuo, T., Ojima, Y., Nakamura, K., Hongo, A. and Hiramatsu, Y. (2006). Association of CXCR4 and CCR7 chemokine receptor expression and lymph node metastasis in human cervical cancer. *Ann Oncol.*
- 179. Kim, J., Mori, T., Chen, S.L., Amersi, F.F., Martinez, S.R., Kuo, C., Turner, R.R., Ye, X., Bilchik, A.J., Morton, D.L. and Hoon, D.S. (2006). Chemokine Receptor CXCR4 Expression in Patients With Melanoma and Colorectal Cancer Liver Metastases and the Association With Disease Outcome. *Ann Surg*, 244(1): 113-120.
- 180. Schimanski, C.C., Schwald, S., Simiantonaki, N., Jayasinghe, C., Gonner, U., Wilsberg, V., Junginger, T., Berger, M.R., Galle, P.R. and Moehler, M. (2005). Effect of chemokine receptors CXCR4 and CCR7 on the metastatic behavior of human colorectal cancer. *Clin Cancer Res*, **11**(5): 1743-50.
- 181. Yoon, Y., Liang, Z., Zhang, X., Choe, M., Zhu, A., Cho, H.T., Shin, D.M., Goodman, M.M., Chen, Z.G. and Shim, H. (2007). CXC Chemokine Receptor-4 Antagonist Blocks Both Growth of Primary Tumor and Metastasis of Head and Neck Cancer in Xenograft Mouse Models. *Cancer Res*, **67**(15): 7518-7524.
- 182. Hu, J., Deng, X., Bian, X., Li, G., Tong, Y., Li, Y., Wang, Q., Xin, R., He, X., Zhou, G., Xie, P., Wang, J.M. and Cao, Y. (2005). The expression of functional chemokine receptor CXCR4 is associated with the metastatic potential of human nasopharyngeal carcinoma. *Clin Cancer Res*, **11**(13): 4658-65.
- 183. Phillips, R.J., Burdick, M.D., Lutz, M., Belperio, J.A., Keane, M.P. and Strieter, R.M. (2003). The stromal derived factor-1/CXCL12-CXC chemokine receptor 4 biological axis in non-small cell lung cancer metastases. *Am J Respir Crit Care Med*, 167(12): 1676-86.
- 184. Scala, S., Ottaiano, A., Ascierto, P.A., Cavalli, M., Simeone, E., Giuliano, P., Napolitano, M., Franco, R., Botti, G. and Castello, G. (2005). Expression of CXCR4 predicts poor prognosis in patients with malignant melanoma. *Clin Cancer Res*, 11(5): 1835-41.
- 185. Scala, S., Giuliano, P., Ascierto, P.A., Ierano, C., Franco, R., Napolitano, M., Ottaiano, A., Lombardi, M.L., Luongo, M., Simeone, E., Castiglia, D., Mauro, F., De Michele, I., Calemma, R., Botti, G., Caraco, C., Nicoletti, G., Satriano, R.A. and Castello, G. (2006). Human melanoma metastases express functional CXCR4. *Clin Cancer Res*, **12**(8): 2427-33.
- 186. Russell, H.V., Hicks, J., Okcu, M.F. and Nuchtern, J.G. (2004). CXCR4 expression in neuroblastoma primary tumors is associated with clinical presentation of bone and bone marrow metastases. *J Pediatr Surg*, **39**(10): 1506-11.
- 187. Zhang, L., Yeger, H., Das, B., Irwin, M.S. and Baruchel, S. (2007). Tissue microenvironment modulates CXCR4 expression and tumor metastasis in neuroblastoma. *Neoplasia*, **9**(1): 36-46.
- 188. Uchida, D., Onoue, T., Tomizuka, Y., Begum, N.M., Miwa, Y., Yoshida, H. and Sato, M. (2007). Involvement of an autocrine stromal cell derived factor-1/CXCR4 system on the distant metastasis of human oral squamous cell carcinoma. *Mol Cancer Res*, 5(7): 685-94.
- 189. Uchida, D., Begum, N.M., Tomizuka, Y., Bando, T., Almofti, A., Yoshida, H. and Sato, M. (2004). Acquisition of lymph node, but not distant metastatic potentials,

by the overexpression of CXCR4 in human oral squamous cell carcinoma. Lab Invest.

- 190. Marchesi, F., Monti, P., Leone, B.E., Zerbi, A., Vecchi, A., Piemonti, L., Mantovani, A. and Allavena, P. (2004). Increased survival, proliferation, and migration in metastatic human pancreatic tumor cells expressing functional CXCR4. *Cancer Res*, **64**(22): 8420-7.
- 191. Saur, D., Seidler, B., Schneider, G., Algul, H., Beck, R., Senekowitsch-Schmidtke, R., Schwaiger, M. and Schmid, R.M. (2005). CXCR4 expression increases liver and lung metastasis in a mouse model of pancreatic cancer. *Gastroenterology*, 129(4): 1237-50.
- 192. Pan, J., Mestas, J., Burdick, M.D., Phillips, R.J., Thomas, G.V., Reckamp, K., Belperio, J.A. and Strieter, R.M. (2006). Stromal derived factor-1 (SDF-1/CXCL12) and CXCR4 in renal cell carcinoma metastasis. *Mol Cancer*, **5**: 56.
- 193. Libura, J., Drukala, J., Majka, M., Tomescu, O., Navenot, J.M., Kucia, M., Marquez, L., Peiper, S.C., Barr, F.G., Janowska-Wieczorek, A. and Ratajczak, M.Z. (2002). CXCR4-SDF-1 signalling is active in rhabdomyosarcoma cells and regulates locomotion, chemotaxis, and adhesion. *Blood*, **100**(7): 2597-606.
- 194. Taichman, R.S., Cooper, C., Keller, E.T., Pienta, K.J., Taichman, N.S. and McCauley, L.K. (2002). Use of the stromal cell-derived factor-1/CXCR4 pathway in prostate cancer metastasis to bone. *Cancer Res*, **62**(6): 1832-7.
- 195. Darash-Yahana, M., Pikarsky, E., Abramovitch, R., Zeira, E., Pal, B., Karplus, R., Beider, K., Avniel, S., Kasem, S., Galun, E. and Peled, A. (2004). Role of high expression levels of CXCR4 in tumor growth, vascularisation, and metastasis. *Faseb J.*
- 196. Orimo, A., Gupta, P.B., Sgroi, D.C., Arenzana-Seisdedos, F., Delaunay, T., Naeem, R., Carey, V.J., Richardson, A.L. and Weinberg, R.A. (2005). Stromal fibroblasts present in invasive human breast carcinomas promote tumor growth and angiogenesis through elevated SDF-1/CXCL12 secretion. *Cell*, **121**(3): 335-48.
- 197. Lapteva, N., Yang, A.G., Sanders, D.E., Strube, R.W. and Chen, S.Y. (2004). CXCR4 knockdown by small interfering RNA abrogates breast tumor growth in vivo. *Cancer Gene Ther*.
- 198. Hall, J.M. and Korach, K.S. (2003). Stromal cell-derived factor 1, a novel target of estrogen receptor action, mediates the mitogenic effects of estradiol in ovarian and breast cancer cells. *Mol Endocrinol*, **17**(5): 792-803.
- Allinen, M., Beroukhim, R., Cai, L., Brennan, C., Lahti-Domenici, J., Huang, H., Porter, D., Hu, M., Chin, L., Richardson, A., Schnitt, S., Sellers, W.R. and Polyak, K. (2004). Molecular characterization of the tumor microenvironment in breast cancer. *Cancer Cell*, 6(1): 17-32.
- 200. Kishimoto, H., Wang, Z., Bhat-Nakshatri, P., Chang, D., Clarke, R. and Nakshatri, H. (2005). The p160 family coactivators regulate breast cancer cell proliferation and invasion through autocrine/paracrine activity of SDF-1alpha/CXCL12. *Carcinogenesis*, **26**(10): 1706-15.
- 201. Rubin, J.B., Kung, A.L., Klein, R.S., Chan, J.A., Sun, Y., Schmidt, K., Kieran, M.W., Luster, A.D. and Segal, R.A. (2003). A small-molecule antagonist of CXCR4 inhibits intracranial growth of primary brain tumors. *Proc Natl Acad Sci U S A*, **100**(23): 13513-8.
- 202. Dewan, M.Z., Ahmed, S., Iwasaki, Y., Ohba, K., Toi, M. and Yamamoto, N. (2006). Stromal cell-derived factor-1 and CXCR4 receptor interaction in tumor growth and metastasis of breast cancer. *Biomed Pharmacother*, **60**(6): 273-6.

- 203. Rae, J.M., Creighton, C.J., Meck, J.M., Haddad, B.R. and Johnson, M.D. (2007). MDA-MB-435 cells are derived from M14 melanoma cells--a loss for breast cancer, but a boon for melanoma research. *Breast Cancer Res Treat*, **104**(1): 13-9.
- 204. Kang, H., Mansel, R.E. and Jiang, W.G. (2005). Genetic manipulation of stromal cell-derived factor-1 attests the pivotal role of the autocrine SDF-1-CXCR4 pathway in the aggressiveness of breast cancer cells. *Int J Oncol*, **26**(5): 1429-34.
- 205. Liang, Z., Yoon, Y., Votaw, J., Goodman, M.M., Williams, L. and Shim, H. (2005). Silencing of CXCR4 Blocks Breast Cancer Metastasis. *Cancer Res*, 65(3): 967-971.
- 206. Prasad, A., Fernandis, A.Z., Rao, Y. and Ganju, R.K. (2004). Slit Protein-mediated Inhibition of CXCR4-induced Chemotactic and Chemoinvasive Signalling Pathways in Breast Cancer Cells. *J Biol Chem*, **279**(10): 9115-24.
- 207. Fernandis, A.Z., Prasad, A., Band, H., Klosel, R. and Ganju, R.K. (2004). Regulation of CXCR4-mediated chemotaxis and chemoinvasion of breast cancer cells. *Oncogene*, **23**(1): 157-67.
- 208. Lechertier, T., Berard, M., Vassy, R., Herve, M.A. and Crepin, M. (2004). Transendothelial migration of two metastatic breast carcinoma cells depend on the SDF-lalpha-CXCR4 complexes. *Anticancer Res*, **24**(6): 4011-7.
- 209. Lee, B.-C., Lee, T.-H., Avraham, S. and Avraham, H.K. (2004). Involvement of the Chemokine Receptor CXCR4 and Its Ligand Stromal Cell-Derived Factor 1 {alpha} in Breast Cancer Cell Migration Through Human Brain Microvascular Endothelial Cells. *Mol Cancer Res*, **2**(6): 327-338.
- 210. Bachelder, R.E., Wendt, M.A. and Mercurio, A.M. (2002). Vascular endothelial growth factor promotes breast carcinoma invasion in an autocrine manner by regulating the chemokine receptor CXCR4. *Cancer Res*, **62**(24): 7203-6.
- 211. Sheridan, C., Sadaria, M., Bhat-Nakshatri, P., Goulet, R., Jr., Edenberg, H.J., McCarthy, B.P., Chang, C.H., Srour, E.F. and Nakshatri, H. (2006). Negative regulation of MHC class II gene expression by CXCR4. *Exp Hematol*, **34**(8): 1085-92.
- 212. Lee, C.H., Kakinuma, T., Wang, J., Zhang, H., Palmer, D.C., Restifo, N.P. and Hwang, S.T. (2006). Sensitization of B16 tumor cells with a CXCR4 antagonist increases the efficacy of immunotherapy for established lung metastases. *Mol Cancer Ther*, **5**(10): 2592-9.
- 213. Mellado, M., de Ana, A.M., Moreno, M.C., Martinez, C. and Rodriguez-Frade, J.M. (2001). A potential immune escape mechanism by melanoma cells through the activation of chemokine-induced T cell death. *Curr Biol*, **11**(9): 691-6.
- 214. Zou, W., Machelon, V., Coulomb-L'Hermin, A., Borvak, J., Nome, F., Isaeva, T., Wei, S., Krzysiek, R., Durand-Gasselin, I., Gordon, A., Pustilnik, T., Curiel, D.T., Galanaud, P., Capron, F., Emilie, D. and Curiel, T.J. (2001). Stromal-derived factor-1 in human tumors recruits and alters the function of plasmacytoid precursor dendritic cells. *Nat Med*, 7(12): 1339-46.
- 215. Nomura, T., Hasegawa, H., Kohno, M., Sasaki, M. and Fujita, S. (2001). Enhancement of anti-tumor immunity by tumor cells transfected with the secondary lymphoid tissue chemokine EBI-1-ligand chemokine and stromal cell-derived factor-1alpha chemokine genes. *Int J Cancer*, **91**(5): 597-606.
- 216. Dunussi-Joannopoulos, K., Zuberek, K., Runyon, K., Hawley, R.G., Wong, A., Erickson, J., Herrmann, S. and Leonard, J.P. (2002). Efficacious immunomodulatory activity of the chemokine stromal cell-derived factor 1 (SDF-1): local secretion of SDF-1 at the tumor site serves as T-cell chemoattractant and mediates T-cell-dependent antitumor responses. *Blood*, **100**(5): 1551-8.

- 217. Shi, M., Hao, S., Su, L., Zhang, X., Yuan, J., Guo, X., Zheng, C. and Xiang, J. (2005). Vaccine of engineered tumor cells secreting stromal cell-derived factor-1 induces T-cell dependent antitumor responses. *Cancer Biother Radiopharm*, **20**(4): 401-9.
- 218. Fushimi, T., O'Connor, T.P. and Crystal, R.G. (2006). Adenoviral Gene Transfer of Stromal Cell-Derived Factor-1 to Murine Tumors Induces the Accumulation of Dendritic Cells and Suppresses Tumor Growth. *Cancer Res*, **66**(7): 3513-3522.
- 219. Vianello, F., Papeta, N., Chen, T., Kraft, P., White, N., Hart, W.K., Kircher, M.F., Swart, E., Rhee, S., Palu, G., Irimia, D., Toner, M., Weissleder, R. and Poznansky, M.C. (2006). Murine B16 melanomas expressing high levels of the chemokine stromal-derived factor-1/CXCL12 induce tumor-specific T cell chemorepulsion and escape from immune control. *J Immunol*, **176**(5): 2902-14.
- 220. Wilbanks, A., Zondlo, S.C., Murphy, K., Mak, S., Soler, D., Langdon, P., Andrew, D.P., Wu, L. and Briskin, M. (2001). Expression cloning of the STRL33/BONZO/TYMSTRligand reveals elements of CC, CXC, and CX3C chemokines. *J Immunol*, **166**(8): 5145-54.
- 221. Matloubian, M., David, A., Engel, S., Ryan, J.E. and Cyster, J.G. (2000). A transmembrane CXC chemokine is a ligand for HIV-coreceptor Bonzo. *Nat Immunol*, 1(4): 298-304.
- 222. Abel, S., Hundhausen, C., Mentlein, R., Schulte, A., Berkhout, T.A., Broadway, N., Hartmann, D., Sedlacek, R., Dietrich, S., Muetze, B., Schuster, B., Kallen, K.J., Saftig, P., Rose-John, S. and Ludwig, A. (2004). The transmembrane CXCchemokine ligand 16 is induced by IFN-gamma and TNF-alpha and shed by the activity of the disintegrin-like metalloproteinase ADAM10. *J Immunol*, **172**(10): 6362-72.
- 223. Gough, P.J., Garton, K.J., Wille, P.T., Rychlewski, M., Dempsey, P.J. and Raines, E.W. (2004). A disintegrin and metalloproteinase 10-mediated cleavage and shedding regulates the cell surface expression of CXC chemokine ligand 16. *J Immunol*, **172**(6): 3678-85.
- 224. Nakayama, T., Hieshima, K., Izawa, D., Tatsumi, Y., Kanamaru, A. and Yoshie, O. (2003). Cutting edge: profile of chemokine receptor expression on human plasma cells accounts for their efficient recruitment to target tissues. *J Immunol*, **170**(3): 1136-40.
- 225. Chandrasekar, B., Bysani, S. and Mummidi, S. (2004). CXCL16 signals via Gi, phosphatidylinositol 3-kinase, Akt, I kappa B kinase, and nuclear factor-kappa B and induces cell-cell adhesion and aortic smooth muscle cell proliferation. *J Biol Chem*, **279**(5): 3188-96.
- 226. Hofnagel, O., Luechtenborg, B., Plenz, G. and Robenek, H. (2002). Expression of the novel scavenger receptor SR-PSOX in cultured aortic smooth muscle cells and umbilical endothelial cells. *Arterioscler Thromb Vasc Biol*, **22**(4): 710-1.
- 227. Shashkin, P., Simpson, D., Mishin, V., Chesnutt, B. and Ley, K. (2003). Expression of CXCL16 in human T cells. *Arterioscler Thromb Vasc Biol*, **23**(1): 148-9.
- 228. Liao, F., Alkhatib, G., Peden, K.W., Sharma, G., Berger, E.A. and Farber, J.M. (1997). STRL33, A novel chemokine receptor-like protein, functions as a fusion cofactor for both macrophage-tropic and T cell line-tropic HIV-1. *J Exp Med*, **185**(11): 2015-23.
- 229. Sharron, M., Pohlmann, S., Price, K., Lolis, E., Tsang, M., Kirchhoff, F., Doms, R.W. and Lee, B. (2000). Expression and coreceptor activity of STRL33/Bonzo on primary peripheral blood lymphocytes. *Blood*, **96**(1): 41-9.

- 230. Unutmaz, D., Xiang, W., Sunshine, M.J., Campbell, J., Butcher, E. and Littman, D.R. (2000). The primate lentiviral receptor Bonzo/STRL33 is coordinately regulated with CCR5 and its expression pattern is conserved between human and mouse. *J Immunol*, **165**(6): 3284-92.
- 231. Kim, C.H., Kunkel, E.J., Boisvert, J., Johnston, B., Campbell, J.J., Genovese, M.C., Greenberg, H.B. and Butcher, E.C. (2001). Bonzo/CXCR6 expression defines type 1-polarized T-cell subsets with extralymphoid tissue homing potential. *J Clin Invest*, **107**(5): 595-601.
- 232. Ignatius, R., Wei, Y., Beaulieu, S., Gettie, A., Steinman, R.M., Pope, M. and Mojsov, S. (2000). The immunodeficiency virus coreceptor, Bonzo/STRL33/TYMSTR, is expressed by macaque and human skin- and bloodderived dendritic cells. *AIDS Res Hum Retroviruses*, 16(11): 1055-9.
- 233. Wagsater, D., Olofsson, P.S., Norgren, L., Stenberg, B. and Sirsjo, A. (2004). The chemokine and scavenger receptor CXCL16/SR-PSOX is expressed in human vascular smooth muscle cells and is induced by interferon gamma. *Biochem Biophys Res Commun*, **325**(4): 1187-93.
- 234. Wuttge, D.M., Zhou, X., Sheikine, Y., Wagsater, D., Stemme, V., Hedin, U., Stemme, S., Hansson, G.K. and Sirsjo, A. (2004). CXCL16/SR-PSOX Is an Interferon-{gamma}-Regulated Chemokine and Scavenger Receptor Expressed in Atherosclerotic Lesions. *Arterioscler Thromb Vasc Biol*.
- 235. Yamauchi, R., Tanaka, M., Kume, N., Minami, M., Kawamoto, T., Togi, K., Shimaoka, T., Takahashi, S., Yamaguchi, J., Nishina, T., Kitaichi, M., Komeda, M., Manabe, T., Yonehara, S. and Kita, T. (2004). Upregulation of SR-PSOX/CXCL16 and recruitment of CD8+ T cells in cardiac valves during inflammatory valvular heart disease. *Arterioscler Thromb Vasc Biol*, 24(2): 282-7.
- 236. Shimaoka, T., Nakayama, T., Fukumoto, N., Kume, N., Takahashi, S., Yamaguchi, J., Minami, M., Hayashida, K., Kita, T., Ohsumi, J., Yoshie, O. and Yonehara, S. (2004). Cell surface-anchored SR-PSOX/CXC chemokine ligand 16 mediates firm adhesion of CXC chemokine receptor 6-expressing cells. *J Leukoc Biol*, **75**(2): 267-74.
- 237. Shimaoka, T., Nakayama, T., Kume, N., Takahashi, S., Yamaguchi, J., Minami, M., Hayashida, K., Kita, T., Ohsumi, J., Yoshie, O. and Yonehara, S. (2003). Cutting edge: SR-PSOX/CXC chemokine ligand 16 mediates bacterial phagocytosis by APCs through its chemokine domain. *J Immunol*, **171**(4): 1647-51.
- 238. Fahy, O.L., Townley, S.L. and McColl, S.R. (2006). CXCL16 regulates cellmediated immunity to Salmonella enterica serovar Enteritidis via promotion of gamma interferon production. *Infect Immun*, **74**(12): 6885-94.
- 239. Xu, H., Xu, W., Chu, Y., Gong, Y., Jiang, Z. and Xiong, S. (2005). Involvement of up-regulated CXC chemokine ligand 16/scavenger receptor that binds phosphatidylserine and oxidized lipoprotein in endotoxin-induced lethal liver injury via regulation of T-cell recruitment and adhesion. *Infect Immun*, **73**(7): 4007-16.
- 240. Sato, T., Thorlacius, H., Johnston, B., Staton, T.L., Xiang, W., Littman, D.R. and Butcher, E.C. (2005). Role for CXCR6 in recruitment of activated CD8+ lymphocytes to inflamed liver. *J Immunol*, **174**(1): 277-83.
- 241. Wang, J., Holmes, T.H., Cheung, R., Greenberg, H.B. and He, X.S. (2004). Expression of chemokine receptors on intrahepatic and peripheral lymphocytes in chronic hepatitis C infection: its relationship to liver inflammation. *J Infect Dis*, 190(5): 989-97.
- 242. Heydtmann, M., Lalor, P.F., Eksteen, J.A., Hubscher, S.G., Briskin, M. and Adams, D.H. (2005). CXC chemokine ligand 16 promotes integrin-mediated adhesion of

liver-infiltrating lymphocytes to cholangiocytes and hepatocytes within the inflamed human liver. *J Immunol*, **174**(2): 1055-62.

- 243. Boisvert, J., Kunkel, E.J., Campbell, J.J., Keeffe, E.B., Butcher, E.C. and Greenberg, H.B. (2003). Liver-infiltrating lymphocytes in end-stage hepatitis C virus: subsets, activation status, and chemokine receptor phenotypes. *J Hepatol*, 38(1): 67-75.
- 244. Geissmann, F., Cameron, T.O., Sidobre, S., Manlongat, N., Kronenberg, M., Briskin, M.J., Dustin, M.L. and Littman, D.R. (2005). Intravascular immune surveillance by CXCR6+ NKT cells patrolling liver sinusoids. *PLoS Biol*, **3**(4): e113.
- 245. Ruth, J.H., Haas, C.S., Park, C.C., Amin, M.A., Martinez, R.J., Haines, G.K., 3rd, Shahrara, S., Campbell, P.L. and Koch, A.E. (2006). CXCL16-mediated cell recruitment to rheumatoid arthritis synovial tissue and murine lymph nodes is dependent upon the MAPK pathway. *Arthritis Rheum*, **54**(3): 765-778.
- 246. Nanki, T., Shimaoka, T., Hayashida, K., Taniguchi, K., Yonehara, S. and Miyasaka, N. (2005). Pathogenic role of the CXCL16-CXCR6 pathway in rheumatoid arthritis. *Arthritis Rheum*, **52**(10): 3004-3014.
- 247. van der Voort, R., van Lieshout, A.W., Toonen, L.W., Sloetjes, A.W., van den Berg, W.B., Figdor, C.G., Radstake, T.R. and Adema, G.J. (2005). Elevated CXCL16 expression by synovial macrophages recruits memory T cells into rheumatoid joints. *Arthritis Rheum*, **52**(5): 1381-91.
- 248. Jiang, X., Shimaoka, T., Kojo, S., Harada, M., Watarai, H., Wakao, H., Ohkohchi, N., Yonehara, S., Taniguchi, M. and Seino, K.-i. (2005). Cutting Edge: Critical Role of CXCL16/CXCR6 in NKT Cell Trafficking in Allograft Tolerance. *J Immunol*, **175**(4): 2051-2055.
- 249. Zhuge, X., Murayama, T., Arai, H., Yamauchi, R., Tanaka, M., Shimaoka, T., Yonehara, S., Kume, N., Yokode, M. and Kita, T. (2005). CXCL16 is a novel angiogenic factor for human umbilical vein endothelial cells. *Biochem Biophys Res Commun*, **331**(4): 1295-300.
- 250. Minami, M., Kume, N., Shimaoka, T., Kataoka, H., Hayashida, K., Akiyama, Y., Nagata, I., Ando, K., Nobuyoshi, M., Hanyuu, M., Komeda, M., Yonehara, S. and Kita, T. (2001). Expression of SR-PSOX, a novel cell-surface scavenger receptor for phosphatidylserine and oxidized LDL in human atherosclerotic lesions. *Arterioscler Thromb Vasc Biol*, **21**(11): 1796-800.
- 251. Sheikine, Y., Bang, C.S., Nilsson, L., Samnegard, A., Hamsten, A., Jonasson, L., Eriksson, P. and Sirsjo, A. (2006). Decreased plasma CXCL16/SR-PSOX concentration is associated with coronary artery disease. *Atherosclerosis*, 188(2): 462-6.
- 252. Aslanian, A.M. and Charo, I.F. (2006). Targeted disruption of the scavenger receptor and chemokine CXCL16 accelerates atherosclerosis. *Circulation*, **114**(6): 583-90.
- 253. Hanamoto, H., Nakayama, T., Miyazato, H., Takegawa, S., Hieshima, K., Tatsumi, Y., Kanamaru, A. and Yoshie, O. (2004). Expression of CCL28 by Reed-Sternberg cells defines a major subtype of classical Hodgkin's disease with frequent infiltration of eosinophils and/or plasma cells. *Am J Pathol*, **164**(3): 997-1006.
- 254. Wagsater, D., Hugander, A. and Dimberg, J. (2004). Expression of CXCL16 in human rectal cancer. *Int J Mol Med*, **14**(1): 65-9.
- 255. Wagsater, D. and Dimberg, J. (2004). Expression of chemokine receptor CXCR6 in human colorectal adenocarcinomas. *Anticancer Res*, **24**(6): 3711-4.
- 256. Hojo, S., Koizumi, K., Tsuneyama, K., Arita, Y., Cui, Z., Shinohara, K., Minami, T., Hashimoto, I., Nakayama, T., Sakurai, H., Takano, Y., Yoshie, O., Tsukada, K. and Saiki, I. (2007). High-Level Expression of Chemokine CXCL16 by Tumor Cells Correlates with a Good Prognosis and Increased Tumor-Infiltrating Lymphocytes in Colorectal Cancer. *Cancer Res*, **67**(10): 4725-4731.
- 257. Sasai, K., Romer, J.T., Kimura, H., Eberhart, D.E., Rice, D.S. and Curran, T. (2007). Medulloblastomas Derived from Cxcr6 Mutant Mice Respond to Treatment with a Smoothened Inhibitor. *Cancer Res*, **67**(8): 3871-3877.
- 258. Biswas, S.K., Gangi, L., Paul, S., Schioppa, T., Saccani, A., Sironi, M., Bottazzi, B., Doni, A., Bronte, V., Pasqualini, F., Vago, L., Nebuloni, M., Mantovani, A. and Sica, A. (2005). A distinct and unique transcriptional programme expressed by tumor-associated macrophages: defective NF-{kappa}B and enhanced IRF-3/STAT1 activation. *Blood*.
- 259. Ludwig, A., Schulte, A., Schnack, C., Hundhausen, C., Reiss, K., Brodway, N., Held-Feindt, J. and Mentlein, R. (2005). Enhanced expression and shedding of the transmembrane chemokine CXCL16 by reactive astrocytes and glioma cells. J Neurochem, 93(5): 1293-303.
- Xu, Y. and Ganem, D. (2007). Induction of chemokine production by latent Kaposi's sarcoma-associated herpesvirus infection of endothelial cells. *J Gen Virol*, 88(Pt 1): 46-50.
- 261. Ou, D.L., Chen, C.L., Lin, S.B., Hsu, C.H. and Lin, L.I. (2006). Chemokine receptor expression profiles in nasopharyngeal carcinoma and their association with metastasis and radiotherapy. *J Pathol.*
- 262. Ju, X., Katiyar, S., Wang, C., Liu, M., Jiao, X., Li, S., Zhou, J., Turner, J., Lisanti, M.P., Russell, R.G., Mueller, S.C., Ojeifo, J., Chen, W.S., Hay, N. and Pestell, R.G. (2007). Akt1 governs breast cancer progression in vivo. *Proc Natl Acad Sci U S A*.
- Seidl, H., Richtig, E., Tilz, H., Stefan, M., Schmidbauer, U., Asslaber, M., Zatloukal, K., Herlyn, M. and Schaider, H. (2007). Profiles of chemokine receptors in melanocytic lesions: de novo expression of CXCR6 in melanoma. *Hum Pathol*, 38(5): 768-80.
- 264. Zlotnik, A. and Yoshie, O. (2000). Chemokines: a new classification system and their role in immunity. *Immunity*, **12**(2): 121-7.
- 265. Nanki, T. and Lipsky, P.E. (2001). Stimulation of T-Cell Activation by CXCL12/Stromal Cell Derived Factor-1 Involves a G-Protein Mediated Signalling Pathway. *Cellular Immunology*, **214**(2): 145-154.
- 266. Salvucci, O., Basik, M., Yao, L., Bianchi, R. and Tosato, G. (2004). Evidence for the involvement of SDF-1 and CXCR4 in the disruption of endothelial cellbranching morphogenesis and angiogenesis by TNF-alpha and IFN-gamma. *J Leukoc Biol*, **76**(1): 217-26.
- 267. De Falco, E., Porcelli, D., Torella, A.R., Straino, S., Iachininoto, M.G., Orlandi, A., Truffa, S., Biglioli, P., Napolitano, M., Capogrossi, M.C. and Pesce, M. (2004). SDF-1 involvement in endothelial phenotype and ischemia-induced recruitment of bone marrow progenitor cells. *Blood*, **104**(12): 3472-82.
- Li, Q., Shirabe, K., Thisse, C., Thisse, B., Okamoto, H., Masai, I. and Kuwada, J.Y. (2005). Chemokine Signalling Guides Axons within the Retina in Zebrafish. J. Neurosci., 25(7): 1711-1717.
- 269. Patel, D.N., Bailey, S.R., Gresham, J.K., Schuchman, D.B., Shelhamer, J.H., Goldstein, B.J., Foxwell, B.M., Stemerman, M.B., Maranchie, J.K., Valente, A.J., Mummidi, S. and Chandrasekar, B. (2006). TLR4-NOX4-AP-1 signalling mediates

lipopolysaccharide-induced CXCR6 expression in human aortic smooth muscle cells. *Biochem Biophys Res Commun*, **347**(4): 1113-20.

- 270. Alkhatib, G., Liao, F., Berger, E.A., Farber, J.M. and Peden, K.W. (1997). A new SIV co-receptor, STRL33. *Nature*, **388**(6639): 238.
- 271. Mundy, G.R. (2002). Metastasis to bone: causes, consequences and therapeutic opportunities. *Nat Rev Cancer*, **2**(8): 584-93.
- 272. Hobbs, S., Jitrapakdee, S. and Wallace, J.C. (1998). Development of a bicistronic vector driven by the human polypeptide chain elongation factor 1alpha promoter for creation of stable mammalian cell lines that express very high levels of recombinant proteins. *Biochem Biophys Res Commun*, **252**(2): 368-72.
- 273. Bendtsen, J.D., Nielsen, H., von Heijne, G. and Brunak, S. (2004). Improved prediction of signal peptides: SignalP 3.0. *J Mol Biol*, **340**(4): 783-95.
- 274. Nielsen, H., Engelbrecht, J., Brunak, S. and von Heijne, G. (1997). Identification of prokaryotic and eukaryotic signal peptides and prediction of their cleavage sites. *Protein Eng*, **10**(1): 1-6.
- 275. Clark-Lewis, I., Dewald, B., Loetscher, M., Moser, B. and Baggiolini, M. (1994). Structural requirements for interleukin-8 function identified by design of analogs and CXC chemokine hybrids. *J Biol Chem*, **269**(23): 16075-81.
- 276. Lelekakis, M., Moseley, J.M., Martin, T.J., Hards, D., Williams, E., Ho, P., Lowen, D., Javni, J., Miller, F.R., Slavin, J. and Anderson, R.L. (1999). A novel orthotopic model of breast cancer metastasis to bone. *Clin Exp Metastasis*, 17(2): 163-70.
- 277. Aslakson, C.J. and Miller, F.R. (1992). Selective events in the metastatic process defined by analysis of the sequential dissemination of subpopulations of a mouse mammary tumor. *Cancer Res*, **52**(6): 1399-405.
- 278. Eckhardt, B.L., Parker, B.S., van Laar, R.K., Restall, C.M., Natoli, A.L., Tavaria, M.D., Stanley, K.L., Sloan, E.K., Moseley, J.M. and Anderson, R.L. (2005). Genomic analysis of a spontaneous model of breast cancer metastasis to bone reveals a role for the extracellular matrix. *Mol Cancer Res*, 3(1): 1-13.
- 279. Anzai, R., Seki, S., Ogasawara, K., Hashimoto, W., Sugiura, K., Sato, M., Kumagai, K. and Takeda, K. (1996). Interleukin-12 induces cytotoxic NK1+ alpha beta T cells in the lungs of euthymic and athymic mice. *Immunology*, **88**(1): 82-9.
- 280. Takeda, K., Seki, S., Ogasawara, K., Anzai, R., Hashimoto, W., Sugiura, K., Takahashi, M., Satoh, M. and Kumagai, K. (1996). Liver NK1.1+ CD4+ alpha beta T cells activated by IL-12 as a major effector in inhibition of experimental tumor metastasis. *J Immunol*, **156**(9): 3366-73.
- 281. Takehara, T., Uemura, A., Tatsumi, T., Suzuki, T., Kimura, R., Shiotani, A., Ohkawa, K., Kanto, T., Hiramatsu, N. and Hayashi, N. (2007). Natural killer cellmediated ablation of metastatic liver tumors by hydrodynamic injection of IFNalpha gene to mice. *Int J Cancer*, **120**(6): 1252-60.
- 282. Godfrey, D.I., MacDonald, H.R., Kronenberg, M., Smyth, M.J. and Van Kaer, L. (2004). NKT cells: what's in a name? *Nat Rev Immunol*, **4**(3): 231-7.
- 283. Danna, E.A., Sinha, P., Gilbert, M., Clements, V.K., Pulaski, B.A. and Ostrand-Rosenberg, S. (2004). Surgical removal of primary tumor reverses tumor-induced immunosuppression despite the presence of metastatic disease. *Cancer Res*, **64**(6): 2205-11.
- 284. Russell, J.H. and Ley, T.J. (2002). Lymphocyte-mediated cytotoxicity. *Annual Review of Immunology*, **20**(1): 323-370.
- 285. Walczak, H. and Krammer, P.H. (2000). The CD95 (APO-1/Fas) and the TRAIL (APO-2L) Apoptosis Systems. *Experimental Cell Research*, **256**(1): 58-66.

- 286. Yoneda, Y. and Yoshida, R. (1998). The role of T cells in allografted tumor rejection: IFN-gamma released from T cells is essential for induction of effector macrophages in the rejection site. *J Immunol*, **160**(12): 6012-7.
- 287. Langowski, J.L., Zhang, X., Wu, L., Mattson, J.D., Chen, T., Smith, K., Basham, B., McClanahan, T., Kastelein, R.A. and Oft, M. (2006). IL-23 promotes tumour incidence and growth. *Nature*, 442(7101): 461-465.
- 288. Weaver, C.T., Hatton, R.D., Mangan, P.R. and Harrington, L.E. (2007). IL-17 Family Cytokines and the Expanding Diversity of Effector T Cell Lineages. *Annual Review of Immunology*, **25**(1).
- 289. Lin, W.-W. and Karin, M. (2007). A cytokine-mediated link between innate immunity, inflammation, and cancer. J. Clin. Invest., **117**(5): 1175-1183.
- 290. Burger, M., Glodek, A., Hartmann, T., Schmitt-Graff, A., Silberstein, L.E., Fujii, N., Kipps, T.J. and Burger, J.A. (2003). Functional expression of CXCR4 (CD184) on small-cell lung cancer cells mediates migration, integrin activation, and adhesion to stromal cells. *Oncogene*, **22**(50): 8093-101.
- 291. Criscuoli, M.L., Nguyen, M. and Eliceiri, B.P. (2005). Tumor metastasis but not tumor growth is dependent on Src-mediated vascular permeability. *Blood*, **105**(4): 1508-1514.
- 292. Dickson, P.V., Hamner, J.B., Streck, C.J., Ng, C.Y., McCarville, M.B., Calabrese, C., Gilbertson, R.J., Stewart, C.F., Wilson, C.M., Gaber, M.W., Pfeffer, L.M., Skapek, S.X., Nathwani, A.C. and Davidoff, A.M. (2007). Continuous delivery of IFN-beta promotes sustained maturation of intratumoral vasculature. *Mol Cancer Res*, 5(6): 531-42.
- 293. Blunt, T., Finnie, N.J., Taccioli, G.E., Smith, G.C., Demengeot, J., Gottlieb, T.M., Mizuta, R., Varghese, A.J., Alt, F.W., Jeggo, P.A. and et al. (1995). Defective DNA-dependent protein kinase activity is linked to V(D)J recombination and DNA repair defects associated with the murine scid mutation. *Cell*, **80**(5): 813-23.
- 294. Nehls, M., Pfeifer, D., Schorpp, M., Hedrich, H. and Boehm, T. (1994). New member of the winged-helix protein family disrupted in mouse and rat nude mutations. *Nature*, **372**(6501): 103-7.
- 295. Nehls, M., Kyewski, B., Messerle, M., Waldschutz, R., Schuddekopf, K., Smith, A.J. and Boehm, T. (1996). Two genetically separable steps in the differentiation of thymic epithelium. *Science*, **272**(5263): 886-9.
- 296. Huleatt, J. and Lefrancois, L. (1995). Antigen-driven induction of CD11c on intestinal intraepithelial lymphocytes and CD8+ T cells in vivo. *J Immunol*, **154**(11): 5684-5693.
- 297. Vicari, A.P., Treilleux, I. and Lebecque, S. (2004). Regulation of the trafficking of tumour-infiltrating dendritic cells by chemokines. *Semin Cancer Biol*, **14**(3): 161-9.
- 298. Cyster, J.G. (1999). Chemokines and cell migration in secondary lymphoid organs. *Science*, **286**(5447): 2098-102.
- 299. Kabashima, K., Sugita, K., Shiraishi, N., Tamamura, H., Fujii, N. and Tokura, Y. (2007). CXCR4 engagement promotes dendritic cell survival and maturation. *Biochem Biophys Res Commun.*
- Sallusto, F., Schaerli, P., Loetscher, P., Schaniel, C., Lenig, D., Mackay, C.R., Qin, S. and Lanzavecchia, A. (1998). Rapid and coordinated switch in chemokine receptor expression during dendritic cell maturation. *Eur J Immunol*, 28(9): 2760-9.
- 301. Ngo, V.N., Tang, H.L. and Cyster, J.G. (1998). Epstein-Barr virus-induced molecule 1 ligand chemokine is expressed by dendritic cells in lymphoid tissues and strongly attracts naive T cells and activated B cells. *J Exp Med*, **188**(1): 181-91.

- 302. Willimann, K., Legler, D.F., Loetscher, M., Roos, R.S., Delgado, M.B., Clark-Lewis, I., Baggiolini, M. and Moser, B. (1998). The chemokine SLC is expressed in T cell areas of lymph nodes and mucosal lymphoid tissues and attracts activated T cells via CCR7. *Eur J Immunol*, **28**(6): 2025-34.
- 303. Itano, A.A. and Jenkins, M.K. (2003). Antigen presentation to naive CD4 T cells in the lymph node. **4**(8): 733-739.
- 304. Yu, P., Spiotto, M.T., Lee, Y., Schreiber, H. and Fu, Y.-X. (2003). Complementary Role of CD4+ T Cells and Secondary Lymphoid Tissues for Cross-presentation of Tumor Antigen to CD8+ T Cells. *J. Exp. Med.*, **197**(8): 985-995.
- 305. Pajusto, M., Ihalainen, N., Pelkonen, J., Tarkkanen, J. and Mattila, P.S. (2004). Human in vivo-activated CD45R0(+) CD4(+) T cells are susceptible to spontaneous apoptosis that can be inhibited by the chemokine CXCL12 and IL-2, -6, -7, and -15. *Eur J Immunol*, **34**(10): 2771-80.
- 306. Li, Z., Pradera, F., Kammertoens, T., Li, B., Liu, S. and Qin, Z. (2007). Cross-Talk between T Cells and Innate Immune Cells Is Crucial for IFN-{gamma}-Dependent Tumor Rejection. *J Immunol*, **179**(3): 1568-1576.
- 307. Sad, S., Marcotte, R. and Mosmann, T.R. (1995). Cytokine-induced differentiation of precursor mouse CD8+ T cells into cytotoxic CD8+ T cells secreting Th1 or Th2 cytokines. *Immunity*, **2**(3): 271-9.
- 308. Kagi, D., Ledermann, B., Burki, K., Seiler, P., Odermatt, B., Olsen, K.J., Podack, E.R., Zinkernagel, R.M. and Hengartner, H. (1994). Cytotoxicity mediated by T cells and natural killer cells is greatly impaired in perforin-deficient mice. *Nature*, 369(6475): 31-7.
- 309. LeBlanc, H.N. and Ashkenazi, A. (2003). Apo2L/TRAIL and its death and decoy receptors. *Cell Death Differ*, **10**(1): 66-75.
- 310. Cretney, E., Takeda, K., Yagita, H., Glaccum, M., Peschon, J.J. and Smyth, M.J. (2002). Increased Susceptibility to Tumor Initiation and Metastasis in TNF-Related Apoptosis-Inducing Ligand-Deficient Mice. *J Immunol*, **168**(3): 1356-1361.
- 311. Taieb, J., Chaput, N., Menard, C., Apetoh, L., Ullrich, E., Bonmort, M., Pequignot, M., Casares, N., Terme, M., Flament, C., Opolon, P., Lecluse, Y., Metivier, D., Tomasello, E., Vivier, E., Ghiringhelli, F., Martin, F., Klatzmann, D., Poynard, T., Tursz, T., Raposo, G., Yagita, H., Ryffel, B., Kroemer, G. and Zitvogel, L. (2006). A novel dendritic cell subset involved in tumor immunosurveillance. 12(2): 214-219.
- 312. Liu, S., Yu, Y., Zhang, M., Wang, W. and Cao, X. (2001). The involvement of TNF-alpha-related apoptosis-inducing ligand in the enhanced cytotoxicity of IFN-beta-stimulated human dendritic cells to tumor cells. *J Immunol*, **166**(9): 5407-15.
- 313. Chan, C.W., Crafton, E., Fan, H.-N., Flook, J., Yoshimura, K., Skarica, M., Brockstedt, D., Dubensky, T.W., Stins, M.F., Lanier, L.L., Pardoll, D.M. and Housseau, F. (2006). Interferon-producing killer dendritic cells provide a link between innate and adaptive immunity. *Nature Med*, **12**(2): 207-213.
- 314. Harty, J.T., Tvinnereim, A.R. and White, D.W. (2000). CD8+ T cell effector mechanisms in resistance to infection. *Annu Rev Immunol*, **18**: 275-308.
- 315. Shurin, M.R., Shurin, G.V., Lokshin, A., Yurkovetsky, Z.R., Gutkin, D.W., Chatta, G., Zhong, H., Han, B. and Ferris, R.L. (2006). Intratumoral cytokines/chemokines/growth factors and tumor infiltrating dendritic cells: friends or enemies? *Cancer Metastasis Rev*, **25**(3): 333-56.
- 316. Sinha, P., Clements, V.K., Miller, S. and Ostrand-Rosenberg, S. (2005). Tumor immunity: a balancing act between T cell activation, macrophage activation and tumor-induced immune suppression. *Cancer Immunol Immunother*.

- 317. Pak, A.S., Ip, G., Wright, M.A. and Young, M.R. (1995). Treating tumor-bearing mice with low-dose gamma-interferon plus tumor necrosis factor alpha to diminish immune suppressive granulocyte-macrophage progenitor cells increases responsiveness to interleukin 2 immunotherapy. *Cancer Res*, **55**(4): 885-90.
- 318. Scotton, C.J., Wilson, J.L., Scott, K., Stamp, G., Wilbanks, G.D., Fricker, S., Bridger, G. and Balkwill, F.R. (2002). Multiple actions of the chemokine CXCL12 on epithelial tumor cells in human ovarian cancer. *Cancer Res*, **62**(20): 5930-8.
- 319. Bezzi, P., Domercq, M., Brambilla, L., Galli, R., Schols, D., De Clercq, E., Vescovi, A., Bagetta, G., Kollias, G., Meldolesi, J. and Volterra, A. (2001). CXCR4-activated astrocyte glutamate release via TNFalpha: amplification by microglia triggers neurotoxicity. *Nat Neurosci*, 4(7): 702-10.
- 320. Han, Y., He, T., Huang, D.R., Pardo, C.A. and Ransohoff, R.M. (2001). TNF-alpha mediates SDF-1 alpha-induced NF-kappa B activation and cytotoxic effects in primary astrocytes. *J Clin Invest*, **108**(3): 425-35.
- 321. Sinha, P., Clements, V.K. and Ostrand-Rosenberg, S. (2005). Reduction of myeloid-derived suppressor cells and induction of M1 macrophages facilitate the rejection of established metastatic disease. *J Immunol*, **174**(2): 636-45.
- 322. Lee, C., Liu, Q.H., Tomkowicz, B., Yi, Y., Freedman, B.D. and Collman, R.G. (2003). Macrophage activation through CCR5- and CXCR4-mediated gp120-elicited signalling pathways. *J Leukoc Biol*, **74**(5): 676-82.
- 323. Zou, L., Barnett, B., Safah, H., Larussa, V.F., Evdemon-Hogan, M., Mottram, P., Wei, S., David, O., Curiel, T.J. and Zou, W. (2004). Bone marrow is a reservoir for CD4+CD25+ regulatory T cells that traffic through CXCL12/CXCR4 signals. *Cancer Res*, **64**(22): 8451-5.
- 324. Wei, S., Kryczek, I., Edwards, R.P., Zou, L., Szeliga, W., Banerjee, M., Cost, M., Cheng, P., Chang, A., Redman, B., Herberman, R.B. and Zou, W. (2007). Interleukin-2 administration alters the CD4+FOXP3+ T-cell pool and tumor trafficking in patients with ovarian carcinoma. *Cancer Res*, **67**(15): 7487-94.
- 325. Smyth, M.J., Crowe, N.Y., Hayakawa, Y., Takeda, K., Yagita, H. and Godfrey, D.I. (2002). NKT cells conductors of tumor immunity? *Curr Opin Immunol*, **14**(2): 165-71.
- 326. Cui, J., Shin, T., Kawano, T., Sato, H., Kondo, E., Toura, I., Kaneko, Y., Koseki, H., Kanno, M. and Taniguchi, M. (1997). Requirement for V{alpha}14 NKT Cells in IL-12-Mediated Rejection of Tumors. *Science*, **278**(5343): 1623-1626.
- 327. Terabe, M., Swann, J., Ambrosino, E., Sinha, P., Takaku, S., Hayakawa, Y., Godfrey, D.I., Ostrand-Rosenberg, S., Smyth, M.J. and Berzofsky, J.A. (2005). A nonclassical non-V{alpha}14J{alpha}18 CD1d-restricted (type II) NKT cell is sufficient for down-regulation of tumor immunosurveillance. J. Exp. Med., 202(12): 1627-1633.
- 328. Stockmeyer, B., Beyer, T., Neuhuber, W., Repp, R., Kalden, J.R., Valerius, T. and Herrmann, M. (2003). Polymorphonuclear granulocytes induce antibody-dependent apoptosis in human breast cancer cells. *J Immunol*, **171**(10): 5124-9.
- 329. Snijdewint, F.G., von Mensdorff-Pouilly, S., Karuntu-Wanamarta, A.H., Verstraeten, A.A., Livingston, P.O., Hilgers, J. and Kenemans, P. (2001). Antibody-dependent cell-mediated cytotoxicity can be induced by MUC1 peptide vaccination of breast cancer patients. *Int J Cancer*, **93**(1): 97-106.
- 330. Cooley, S., Burns, L.J., Repka, T. and Miller, J.S. (1999). Natural killer cell cytotoxicity of breast cancer targets is enhanced by two distinct mechanisms of antibody-dependent cellular cytotoxicity against LFA-3 and HER2/neu. *Exp Hematol*, **27**(10): 1533-41.

- 331. Padmanabhan, R.R., Paul, R.D., Watson, G.A. and Lopez, D.M. (1988). Emergence of a B lymphocyte population with ADCC effector function in mammary tumor bearing mice. *J Leukoc Biol*, **43**(6): 509-19.
- 332. Hargreaves, D.C., Hyman, P.L., Lu, T.T., Ngo, V.N., Bidgol, A., Suzuki, G., Zou, Y.R., Littman, D.R. and Cyster, J.G. (2001). A coordinated change in chemokine responsiveness guides plasma cell movements. *J Exp Med*, **194**(1): 45-56.
- Shibuya, A. and Honda, S. (2006). Molecular and functional characteristics of the Fcalpha/muR, a novel Fc receptor for IgM and IgA. *Springer Semin Immunopathol*, 28(4): 377-82.
- 334. Mosley, R.L., Styre, D. and Klein, J.R. (1990). Differentiation and functional maturation of bone marrow-derived intestinal epithelial T cells expressing membrane T cell receptor in athymic radiation chimeras. *J Immunol*, **145**(5): 1369-75.
- 335. Sato, K., Ohtsuka, K., Hasegawa, K., Yamagiwa, S., Watanabe, H., Asakura, H. and Abo, T. (1995). Evidence for extrathymic generation of intermediate T cell receptor cells in the liver revealed in thymectomized, irradiated mice subjected to bone marrow transplantation. *J Exp Med*, **182**(3): 759-67.
- 336. Bandeira, A., Itohara, S., Bonneville, M., Burlen-Defranoux, O., Mota-Santos, T., Coutinho, A. and Tonegawa, S. (1991). Extrathymic origin of intestinal intraepithelial lymphocytes bearing T-cell antigen receptor gamma delta. *Proc Natl Acad Sci U S A*, 88(1): 43-7.
- 337. Rocha, B., Vassalli, P. and Guy-Grand, D. (1991). The V beta repertoire of mouse gut homodimeric alpha CD8+ intraepithelial T cell receptor alpha/beta + lymphocytes reveals a major extrathymic pathway of T cell differentiation. *J Exp Med*, **173**(2): 483-6.
- 338. Yamagiwa, S., Sugahara, S., Shimizu, T., Iwanaga, T., Yoshida, Y., Honda, S., Watanabe, H., Suzuki, K., Asakura, H. and Abo, T. (1998). The primary site of CD4- 8- B220+ alphabeta T cells in lpr mice: the appendix in normal mice. J Immunol, 160(6): 2665-74.
- 339. Payer, E., Strohal, R., Kutil, R., Elbe, A. and Stingl, G. (1992). Demonstration of a CD3+ lymphocyte subset in the epidermis of athymic nude mice. Evidence for T cell receptor diversity. *J Immunol*, **149**(2): 413-20.
- 340. Girardi, M., Oppenheim, D.E., Steele, C.R., Lewis, J.M., Glusac, E., Filler, R., Hobby, P., Sutton, B., Tigelaar, R.E. and Hayday, A.C. (2001). Regulation of cutaneous malignancy by gammadelta T cells. *Science*, **294**(5542): 605-9.
- 341. Naito, M., Nagai, H., Kawano, S., Umezu, H., Zhu, H., Moriyama, H., Yamamoto, T., Takatsuka, H. and Takei, Y. (1996). Liposome-encapsulated dichloromethylene diphosphonate induces macrophage apoptosis in vivo and in vitro. *J Leukoc Biol*, 60(3): 337-44.
- 342. van Rooijen, N., Sanders, A. and van den Berg, T.K. (1996). Apoptosis of macrophages induced by liposome-mediated intracellular delivery of clodronate and propamidine. *J Immunol Methods*, **193**(1): 93-9.
- 343. Van Rooijen, N. and Sanders, A. (1994). Liposome mediated depletion of macrophages: mechanism of action, preparation of liposomes and applications. J Immunol Methods, 174(1-2): 83-93.
- 344. Liang, Z., Wu, T., Lou, H., Yu, X., Taichman, R.S., Lau, S.K., Nie, S., Umbreit, J. and Shim, H. (2004). Inhibition of Breast Cancer Metastasis by Selective Synthetic Polypeptide against CXCR4. *Cancer Res*, **64**(12): 4302-4308.
- 345. Tamamura, H., Hori, A., Kanzaki, N., Hiramatsu, K., Mizumoto, M., Nakashima, H., Yamamoto, N., Otaka, A. and Fujii, N. (2003). T140 analogs as CXCR4

antagonists identified as anti-metastatic agents in the treatment of breast cancer. *FEBS Lett*, **550**(1-3): 79-83.

- 346. Smith, M.C., Luker, K.E., Garbow, J.R., Prior, J.L., Jackson, E., Piwnica-Worms, D. and Luker, G.D. (2004). CXCR4 regulates growth of both primary and metastatic breast cancer. *Cancer Res*, **64**(23): 8604-12.
- 347. Padera, T.P., Kadambi, A., di Tomaso, E., Carreira, C.M., Brown, E.B., Boucher, Y., Choi, N.C., Mathisen, D., Wain, J., Mark, E.J., Munn, L.L. and Jain, R.K. (2002). Lymphatic Metastasis in the Absence of Functional Intratumor Lymphatics. *Science*, **296**(5574): 1883-1886.
- 348. Mehlen, P. and Puisieux, A. (2006). Metastasis: a question of life or death. **6**(6): 449-458.
- 349. Pelus, L.M., Bian, H., Fukuda, S., Wong, D., Merzouk, A. and Salari, H. (2005). The CXCR4 agonist peptide, CTCE-0021, rapidly mobilizes polymorphonuclear neutrophils and hematopoietic progenitor cells into peripheral blood and synergizes with granulocyte colony-stimulating factor. *Experimental Hematology*, **33**(3): 295-307.
- 350. DiVietro, J.A., Brown, D.C., Sklar, L.A., Larson, R.S. and Lawrence, M.B. (2007). Immobilized stromal cell-derived factor-1alpha triggers rapid VLA-4 affinity increases to stabilize lymphocyte tethers on VCAM-1 and subsequently initiate firm adhesion. *J Immunol*, **178**(6): 3903-11.
- 351. Chan, J.R., Hyduk, S.J. and Cybulsky, M.I. (2001). Chemoattractants induce a rapid and transient upregulation of monocyte alpha4 integrin affinity for vascular cell adhesion molecule 1 which mediates arrest: an early step in the process of emigration. *J Exp Med*, **193**(10): 1149-58.
- 352. Hidalgo, A., Sanz-Rodriguez, F., Rodriguez-Fernandez, J.L., Albella, B., Blaya, C., Wright, N., Cabanas, C., Prosper, F., Gutierrez-Ramos, J.C. and Teixido, J. (2001). Chemokine stromal cell-derived factor-1alpha modulates VLA-4 integrindependent adhesion to fibronectin and VCAM-1 on bone marrow hematopoietic progenitor cells. *Exp Hematol*, 29(3): 345-55.
- 353. Sanz-Rodriguez, F., Hidalgo, A. and Teixido, J. (2001). Chemokine stromal cellderived factor-1alpha modulates VLA-4 integrin-mediated multiple myeloma cell adhesion to CS-1/fibronectin and VCAM-1. *Blood*, **97**(2): 346-51.
- 354. Minn, A.J., Gupta, G.P., Siegel, P.M., Bos, P.D., Shu, W., Giri, D.D., Viale, A., Olshen, A.B., Gerald, W.L. and Massague, J. (2005). Genes that mediate breast cancer metastasis to lung. *Nature*, **436**(7050): 518-24.
- 355. Singh, S., Singh, U.P., Grizzle, W.E. and Lillard, J.W., Jr. (2004). CXCL12-CXCR4 interactions modulate prostate cancer cell migration, metalloproteinase expression and invasion. *Lab Invest*, **84**(12): 1666-76.
- 356. Janowska-Wieczorek, A., Marquez, L.A., Dobrowsky, A., Ratajczak, M.Z. and Cabuhat, M.L. (2000). Differential MMP and TIMP production by human marrow and peripheral blood CD34(+) cells in response to chemokines. *Exp Hematol*, **28**(11): 1274-85.
- 357. Will, H., Atkinson, S.J., Butler, G.S., Smith, B. and Murphy, G. (1996). The soluble catalytic domain of membrane type 1 matrix metalloproteinase cleaves the propeptide of progelatinase A and initiates autoproteolytic activation. Regulation by TIMP-2 and TIMP-3. *J Biol Chem*, **271**(29): 17119-23.
- 358. Zeelenberg, I.S., Ruuls-Van Stalle, L. and Roos, E. (2003). The chemokine receptor CXCR4 is required for outgrowth of colon carcinoma micrometastases. *Cancer Res*, **63**(13): 3833-9.

- 359. Rotheneder, M. and Kostner, G.M. (1989). Effects of low- and high-density lipoproteins on the proliferation of human breast cancer cells in vitro: differences between hormone-dependent and hormone-independent cell lines. *Int J Cancer*, **43**(5): 875-9.
- 360. Goldstein, J.L., Helgeson, J.A. and Brown, M.S. (1979). Inhibition of cholesterol synthesis with compactin renders growth of cultured cells dependent on the low density lipoprotein receptor. *J Biol Chem*, **254**(12): 5403-9.
- 361. Aslakson, C.J., McEachern, D., Conaway, D.H. and Miller, F.R. (1991). Inhibition of lung colonization at two different steps in the metastatic sequence. *Clin Exp Metastasis*, **9**(2): 139-50.
- 362. Fulton, A., Miller, F., Weise, A. and Wei, W.Z. (2006). Prospects of controlling breast cancer metastasis by immune intervention. *Breast Dis*, **26**: 115-27.